

Title	Natural insurance, conservation of forests and poverty reduction in the semi-arid region of Tigray, northern Ethiopia
Authors	Abadi, Nigussie
Publication date	2017
Original Citation	Abadi, N. 2017. Natural insurance, conservation of forests and poverty reduction in the semi-arid region of Tigray, northern Ethiopia. PhD Thesis, University College Cork.
Type of publication	Doctoral thesis
Rights	© 2017, Nigussie Abadi http://creativecommons.org/licenses/ by-nc-nd/3.0/
Download date	2024-10-12 00:22:35
Item downloaded from	https://hdl.handle.net/10468/5451



University College Cork, Ireland Coláiste na hOllscoile Corcaigh Ollscoil na hÉireann, Corcaigh

National University of Ireland, Cork



NATURAL INSURANCE, CONSERVATION OF FORESTS AND POVERTY REDUCTION IN THE SEMI-ARID REGION OF TIGRAY, NORTHERN ETHIOPIA

Thesis presented by

Nigussie Abadi

for the degree of

Doctor of Philosophy

University College Cork

School/Department: Business School/International Development and Food Policy

Head of School/Department: Professor Thia Hennessy

Supervisor(s): Nick Chisholm

2017

ABSTRACT

This study deals with the interaction of vulnerability to shocks, conservation intervention and poverty reduction using a survey of households living along the margins of forests in the semi-arid region of Tigray, Ethiopia. Five major conclusions that have wider policy implications are derived from this study. First, forest environmental resources provide a natural insurance for households vulnerable to idiosyncratic and covariate risks in rural Tigray, enforcing the widely held view that forests are important as economic buffer in hard times. Second, even if forests provide the food security and income they may perpetuate poverty. Third, households in rural Tigray are not driven into forest extraction for coping shocks only but also by diversification strategy, suggesting that the problem for local communities has both the characteristics of portfolio analysis and economics of insurance. Fourth, forest incomes decrease the incidence of poverty while it increases income inequality in rural Tigray. Finally, we find that the Productive Safety Net Program (PSNP) in Tigray contributes in pursuing the twin goals of poverty reduction and conservation of forests that have often been portrayed as opposing goals.

One important policy implication of this study is that government, policy-makers and natural resource managers need to acknowledge the livelihood safety- net role that forests play in rural livelihoods and recognize that environmental protection policies limiting or banning access and use of forest resources can deepen rural poverty. However, while more intensive forest management to increase accessibility of forest resources is itself a possible strategy, we must be cautious as to who would gain and who would lose from such activities. Finally, the study underlines the importance of promotion of public safety nets such as the Productive Safety Net Program (PSNP), and the promotion of government programmes and policies that increase the

productivity of agricultural production and support diversification into off-farm livelihood and income sources to provide positive incentives for forest conservation and sustainable use.

Key Words: forests, shocks, safety nets, poverty trap, heterogeneity, counterfactual income, propensity score matching, endogenous switching, Tigray, Ethiopia

ACKNOWLEDGEMENT

In the long years of my education many individuals and institutions have contributed a lot and all deserve many thanks. My deepest thanks go to my mother Twres Abrha for parenting me and investing generously on my education out of her meagre income. I have no words to express my gratitude for all her generosity and selfless support without which pursuing my dreams would have been impossible. Thank you my mom for your unique maternal affection, discipline and strength amidst adversity you showed me a path of fortitude. I owe you a lot!. I also thank my sisters and brothers back home for all their support, encouragement and the sincere love over the years. They have always been my reason to persevere and excel in life. This dissertation is the fruit of their love, hard work and sacrifice.

Coming to my PhD, I would like to express my heartfelt gratitude to my supervisor Dr. Nick Chisholm for his constant support, stimulating suggestions and encouragement.

As I was trained in mainstream economics till my master's degree level, it was a big challenge for me when i was starting to write a PhD thesis on areas related to Natural Resources and Environmental Economics. It is through Dr.Fitsum Hagos 's perpetual academic guidance , literature supply, and encouragement that I build confidence and courage to write my thesis. From the critical comments and suggestions he provided at different levels of my thesis work, I have gained ample knowledge and practical skills for conducting both academic and policy research particularly in resource economics. He also challenged me on methodological issues and the presentation styles of the findings in two of my chapters.

Dr.Girmay Tesfay offered me an all –sided support starting from the onset of my study and he was so instrumental, both academically and administratively, for finalization of this work.

I thank Professor Mituku Haile for his constant support and encouragement to start looking for PhD opportunities and finally for his assistance in shaping my research idea, books, constant follow up and fatherly critics.

During these years, I have had the great opportunity of meeting and sharing unforgettable moments with truly exceptional people. I want to express gratitude to Nigus G/her, Amaha G/Tsadik, Daniel Tesfay ,MesfinHagos , HailemichaelTesfay, AgaziAyele, Deniel G/Medhin, AshenfaiAnbesu, SolomomWoldeegzi, Nigus G/Hiwot , Michael T/egzi , Dr.MehariTadele, Dr.Ataklti Beyene Mehari Teamrat, LiyaTesfay, Hiwot G/her , Hedea't and their family for their love and constant support. I will always be indebted for letting me part of their family. I would also like to acknowledge all those who made my social life in Mekelle University an unforgettable adventure. They made the load easier to carry, the tough moments enriching and all other days exciting. Unfortunately, it is out of the scope of this dissertation to name them all.

I gratefully acknowledge the financial support of the National University of Ireland – UCC, the "Robert McNamera Fellowship Program of the World Bank and the Irish Government Development Cooperation – IRISH AID, Ethiopia. Mekelle University for granting study leave, a home base for my work and administrative support for which i am thankful. Many thanks to officials of the Administration and the Rural Development officers of Enderta Atsbi Wondberta, Raya Azebo, Enda-Mohoni and Meswaeti and the farmers and field assistants who participated in the survey i conducted for this work, for their excellent cooperation.

5

LIST OF ABBREVIATIONS

ADB	African Development Bank
ADLI	Agricultural Development Led Industrialization
ATT	Average Treatment of the Treated
BOFED	Bureau of Finance and Economic Development
CBADLI	Conservation Based Agricultural Development Led Industrialization
CBAD	Conservation Based Agricultural Development Strategy
CIA	Central Intelligence Agency
CPR	Common Pool Resources
FAO	Food and Agricultural Organization
FDRE	Federal Democratic Republic of Ethiopia
FGT	Foster, Greer and Thorebecke
FHAPCO	Federal HIV/ AIDS Prevention and Control Office
FIML	Full Information Maximum Likelihood
FMOH	Federal Ministry of Health
FSP	Food Security Program
GTP	Growth and Transformation Plan
HRFA	High Return Forest Activities
LRFR	Low Return Forest Resources
M.a.s.l	Meter above Sea level
MDG	Millennium Development Goals
NFPA	National Forest Priority Areas
NTFP	Non – Timber Forest Products

OLS	Ordinary Least Square Estimation
PASDEP	Plan for Accelerated and Sustainable Development to End Poverty
PSM	Propensity Score Matching
PSNP RFPA	Productive Safety Net Program Regional Forest Priority Areas
SOPRP	Sustainable Development and Poverty Reduction Program
TRBPF	Tigray Region Bureau of Plan and Finance
WHO	World Health Organization of the United Nations

able of Contents
BSTRACT2
CKNOWLEDGEMENT
IST OF ABBREVIATIONS
ist of TablesError! Bookmark not defined.
HAPTER ONE
INTRODUCTION16
1.1. Background of the Thesis16
1.2. Objectives of the Study19
1.3. Research Issues20
1.4. Organization of the thesis22
HAPTER TWO 24
THE SOCIO-ECONOMIC SETTING OF THE STUDY AREA AND RESEARCH APPROACH24
2.1. Location, Demography and Natural Conditions24
2.2. Socio- Economic Aspects25
2.3. Land and Forest Policy Issues
2.4. Research Methodology35
HAPTER THREE 40
OO FORESTS HAVE A ROLE TO PLAY FOR COPING WITH IDIOSYNCRATIC HEATH
HOCKS IN RURAL ETHIOPIA? 40
3.1. Introduction40
3.2. Overview of the Health and Environment Nexus in Ethiopia

3.3. Theoretical model for Shocks and Labour allocation to Forests	46
3.4. Econometric Methodology	50
3.5. Results and Discussion	56
3.6. Conclusions and Policy Recommendations	64
CHAPTER FOUR	74
FOREST – FOOD SECURITY LINKAGES IN NORTH HIGHLANDS OF ETHIOPIA: SAFETY NETS	S OR
POVERTY TRAPS?	74
4.1. Introduction	74
4.2. Review of Related Literature on the safety net role of Forests	77
4.3. Food Security and Common Pool Resources use in Ethiopia	79
4.4. Data Description	80
4.5. Theoretical Framework and Empirical Procedure	83
4.6. Empirical Results and Discussions	90
4.6.1. Propensity Score Results	90
4.6.2. Average Treatment Effects	93
4.7. Does Diversification or Coping Strategy explain households' choice of Participation?	96
4.8. Conclusions and Policy Implications	98
CHAPTER FIVE	111
WELFARE IMPACTS OF FOREST ENVIRONMENTAL RESOURCE COMMERCIALIZATION: A	
MICRO- SIMULATION APPROACH FOR TIGRAY, ETHIOPIA	111
5.1. Introduction	111
5.2. Overview of forest Conservation and poverty Alleviation in Rural Tigray	115
5.3. Data Description and Study Sites	116

5.4. Econometric Model and Estimation Procedure	118
5.4.1. Expected outcome and Treatment Effects	120
5.5. Results and Discussion	123
5.6. Conclusions and Policy Implications	127
CHAPTER SIX	137
ENVIRONMENTAL RESOURCES, POVERTY AND INEQUALITY IN THE SEMI-ARID REC	GION OF
TIGRAY, ETHIOPIA	137
6.1. Introduction	137
6.2. Overview of Forest Management, Poverty and Distributional Aspects in Tigray	143
6.3. Data Description and Study Sites	145
6. 4. Econometric Model and Estimation Procedure	146
6.4.1. Expected outcome and Treatment Effects	148
6.4.2. Measuring Poverty	150
6.4.3 Measuring inequality	151
6.5. Results and Discussion	153
6.5.1 Descriptive Statistics	153
6.5.2 Poverty and inequality Analysis	153
6.6 Conclusions and Policy Implications	158
CHAPTER SEVEN	166
IMPACT OF THE PRODUCTIVE SAFETY NET PROGRAM (PSNP) IN ETHIOPIA IN CON	SERVATION
OF FORESTS	166
7.1. Introduction	166
7.2. Overview of the Productive Safety Nets Program (PSNP) in Ethiopia	169

7.3. Conceptual Framework for Labour Allocation	
7.4. Methodological Framework	
7.4.1. Definitions of Average Treatment Effects	
7.5. Empirical Results	
7.5.1. Descriptive Statistics	
7.5.2. Propensity Score Estimation Results	
7.5.3. Average Treatment Effects	
7.6. Conclusions and Policy implications	
CHAPTER EIGHT	201
DISCUSSIONS AND CONCLUSIONS	201
8.1. Forest management, Shocks and Poverty reduction	201
8.1.1. Idiosyncratic Health Shocks and Forest Environmental Resources	
8.1.2. Forest-Food Security Linkages	
8.1.3. Forest commercialization and Household income	
8.1.4. Forests income poverty and inequality	
8.1.5. PSNP and forest pressure	
8.2. Policy Implications and Further Research	211
8.2.1. Policy Implications	
8.2.2. Further Research	
BIBLIOGRAPHY	215
ANNEX A. HOUSEHOLD QUESTIONNAIRE	

List of Tables

Table 3. 1 Variables and summary statistics of the sample households
Table 3. 2 Mean Separation Test of households affected by health shock and households that did not 68
Table 3.3 IV Probit estimates of health shocks and forest sales 69
Table 3. 4 IV2sls estimates of health shocks and forest dependency
Table 3. 5 IV 2slsestimates of health shocks and forest trips 71
Table 3. 6 Probit estimation of reporting illness72
Table 3. 7 Differences in ATT for treated and control groups73
Table 4. 1 Variables and summary statistics of the sample households 100
Table 4. 2 Mean separation tests of participants and non-participants in forest environmental resource
extraction
Table 4. 3 Propensity score of participation in forest environmental resource extraction
Table 4. 4 Propensity score and covariate balances 104
Table 4. 5 Other covariate balance indicators before and after matching 105
Table 4. 6 Differences in food security outcomes for participants and non-participants for the whole
sample
Table 4. 7 Differences in food security outcomes of poor households 106
Table 4. 8 Differences in food security outcomes of non-poor households 107
Table 4. 9 Determinants of the predicted probability of zero and or negative farming profits

Table 4. 10	Differences in farming profit between participants and non-participants	110
Table 5. 1	Descriptive statistics of sample households	130
Table 5. 2	Mean separation test of participants and non-participants	131
Table 5. 3	Definition of Expected income and treatment effects	132
Table 5. 4	OLS and Endogenous Switching Regression estimates	133
Table 5. 5	Expected income and treatment effects	135
Table 6. 1	Mean separation test of participants and non-participants	161
Table 6. 2	Poverty decomposition: Scenario 1	162
Table 6. 3	Poverty decomposition: Scenario 2	162
Table 6. 4	Differences in poverty index under counterfactual and conventional scenarios	162
Table 6. 5	Gini Index: Scenario 1	163
Table 6. 6	Gini Index: Scenario 2	163
Table 6. 7	Differences in Gini index under counterfactual and conventional scenarios: Scenario 3	163
Table 7. 1	Descriptive statistics of sample households	190
Table 7. 2	Mean separation test of participants and non-participants in PSNP	192
Table 7. 3	Estimation of propensity score: probit model	194
Table 7. 4	Propensity score and covariate balances	195
Table 7. 5	Other covariate balance indicators before and after matching	197

Table 7. 6 Treatment effect and sensitivity	/ analysis19	97
---------------------------------------------	--------------	----

List of Figures

Figure 1. 1 Map of the study area	Error! Bookmark not defined.
Figure 1. 2 GDP Growth Rate Comparison	Error! Bookmark not defined.
Figure 1. 3 Sectoral Growth Rate	Error! Bookmark not defined.
Figure 3. 1 The Household Production Model under a weather Shock Sce	nario Error! Bookmark not
defined.	
Figure 3. 2 The Household Production Model under a health shock scena	ario Error! Bookmark not
defined.	
Figure 6. 1 FGT Curves for Incidence of Poverty	
Figure 6. 2 FGT Curves for Depth of Poverty	
Figure 6.3 FGT Curves for Severity of Poverty	
Figure 6. 4 Lorenz Curve of Forest Income Distribution, Different Scenario	os165
Figure 7. 1 Distribution of propensity scores	

DECLARATION

I hereby declare that this thesis submitted is my own work and has not been submitted for another degree, either at University College Cork or elsewhere.

Signed,

Nigussie Abadi

May 2017

CHAPTER ONE

INTRODUCTION

1.1. Background of the Thesis

Poor people often depend on biodiversity both for their livelihoods and as a safety net against deeper poverty (IIED, 2010). Moreover, of the 1.2 billion people estimated to live on less than US\$ 1 a day (i.e. those that are the target of MDG1), 70% live in rural areas with a high dependence on for livelihoods natural resources all of their or part (http://www.tanzaniagateway.org/docs/MDGs and natural resources management sustaina ble livelihoods resource.pdf). This direct dependence on services from ecosystems is highest amongst people living in arid and semi-arid lands such as Tigray where alternative livelihood options areoften limited and environments are particularlyfragile and risky (MA 2005b cited in Shackleton et al.2008).

Forest environmental resources play a significant role in the livelihoods of rural communities in Ethiopia (Mamoet al.2007, Babulo et al., 2009). Forest environmental income also accounts for 27% of the total household income on average, the second largest household income share next to crop income having a share of 43% in rural Tigray (Babulo et al. 2009). However, forests are also sensitive ecosystems that are threatened by human interventions. As in many other parts of the developing world, forests in Ethiopia, particularly in Tigray¹, suffer from extreme land degradation

¹Between 1990 and 2000, 141,000 ha of forest were lost every year, which equals an average annual deforestation rate of 0.93%. Then between 2000 and 2005, the rate of deforestation increased by 10.4% to 1.03% per year, which totals to 14% (or around 2,114,000 hectares) of forest cover loss in the 15 years between 1990 and 2005(FAO, 2007)

and deforestation. High population growth combined with conversion of forests to agricultural land (Nayssen et al., 2004; Mokria et al., 2009), high dependency on biomass energy (Hagoset al.1999; Gebreegziher 2007) and free grazing (Feoli et al.2002) are the major causes of forest degradation in the region.

Since 1994, the current Ethiopian government has embarked on an economic development strategy known as Agricultural Development – Led industrialization (ADLI) (MOFED, 2002). In line with this national framework, the Tigray region is following a Conservation –Based Agricultural Development –Led Industrialization (CBADLI) strategy to achieve food security and reduce poverty (BOPED, 1995), while giving greater emphasis on community participation and mobilization of local resources in environmental rehabilitation (Tesfay , 2006)².

One of the major strategies of environmental rehabilitation in Tigray includes area enclosures and the establishment of community woodlots for ecological regeneration (Nyssen et al., 2009, Mokriaet al.,2003; Gebremedhin et al., 2000). In addition to enclosures, the regional government in Tigray created restrictions on the use of certain tree species that are enforced by the development agents who are assigned to each village for this purpose (Howard and Smith, 2006;Tesfay 2006; Gebreegzabher, 2007). However, despite many researchers such as those(Nyssen et al., 2009; and Mokria et al., 2009)and the international community alike³putting forward the positive environmental outcomes such a strategy, it is viewed by many as a constraint for achieving poverty reduction (Berhanu 2004; Howard and Smith 2006 and Geberemedhin et al. 2000), equity within

²A review of the recent history of devolution in forestry indicates undeniably that the area of land — although not necessarily forests — managed under devolved and community-based forest management (CBFM) systems has dramatically increased in many countries (White and Martin 2002). However, while devolution of forestry benefits the poor in some countries, it is also remained to be prone to be captured by local elites as any other valuable local resources (see for example,Malla et al., 2003 and Mahanty et al., 2006).

³Abriha Atsbiha a grass root environmental management program in the region has won the prestigious 2012 UNDP equator Prize

and between communities (Chisholm, 2000; Howard and Smith, 2006 ;and Nedassa et al., 2005) which calls into question whether results achieved are sustainable from an economic or environmental stand point (Nedassa et al., 2005 cited in Howard and Smith, 2006).

Over all, two overarching problems have been identified concerning the issues of common pool resources (CPR) management in Tigray (Howard and Smith, 2006): the first is the possibility to halt degradation and sustainably regenerate vegetative resources, and the second is the need to generate livelihood resources that can help to alleviate poverty, food insecurity and continued degradation in the area(Howard and Smith, 2006). The general conclusion is that the conservation of common pool resources (CPR) and the poverty and food security aspects are linked. The CPR has been depleted due to drought and overuse, and has been overused because of changes in institutional and individual access rights⁴ (Howard and Smith, 2006). However, this over use of the common pool resources such as forests may constitute a poverty trap as a result of a tragedy of the commons (Hardin 1968). Too many households in rural Tigray are in need of the resources and the resources cannot provide enough to properly insure all the population. They face thus the classic poverty- environment nexus, where poor people depend on the environment, and over use it (Delacote, 2007). The resultant increased pressure often has a negative feedback on the capacity of the ecosystems to deliver these services and create a downward spiral of increasing poverty and ecosystem degradation (Shackleton et al.2008) and this dependence of the poor on the fragile open access may confirm the suggestion in some recent literature (e.g. Angelsen and Wunder, 2003; Lavange, 2005, Delacote, 2007) of a resource based "poverty trap". The key issue is therefore, to

⁴With the exception of restrictions on certain woody species, most common land across the highlands is de-facto open access since rights regimes around them are lacking or ill-defined. On the other hand, the restrictions in place in enclosures, while increasing the availability of certain resources, generate greater pressure on these open areas (Howard and Smith, 2006)

present the conservation intervention and the dependence on forests as safety nets in locations where they are more than dead –end poverty traps.

Central to these issues is a greater perception and nuanced appreciation of (a) what is meant by poverty alleviation in relation to natural resources , distinguishing between poverty prevention or mitigation ; (b) the links between natural resources dependence and the potential of the natural resource to provide the pathways out of poverty; (c) the safety net of the forest resources and when and how might translate into poverty trap, (d) the extent to which opportunities associated with forest management can be made more pro-poor and thus contribute to the efforts to combat poverty and vulnerability, and (e) the role of social protection measures to provide win –win outcomes for conservation and poverty reduction etc. This research is therefore to address these issues in the context of Tigray Regional state of Ethiopia.

1.2. Objectives of the Study

The general objectives of the research are to contribute to the existing stock of knowledge on the contributions of forest environmental resources to the livelihood of rural households, on managing the trade-offs between conservation of forests and poverty reduction and on the methodological challenges in reaching conclusions on the links between forest biodiversity and poverty using survey of households living along the margins of forests in the region of Tigray, northern Ethiopia. Specifically, the objectives of the research are:

- To investigate the role of forest environmental resources as natural insurance for coping with shocks
- To investigate the contribution of forest environmental resources for food security and household incomes

- To analyse contribution of forest environmental resources for poverty reduction and distribution of income
- To investigate the role of the ProductiveSafetyNet Program (PSNP) in Ethiopia to reconcile poverty reduction and conservation of forest ecosystems.

1.3. Research Issues

The first topic deals with *therole of forests for coping with health related shocks*. The question is how idiosyncratic health related shocks affect the allocation of labour into forests and the forest dependency of households living along the margins of forests in the semi-arid region of Tigray, Ethiopia.

The second issue of this research addresses the *link between forests and food security of smallholder farmers along the margins of forests*. An attempt is made to explicitly focus on food security indicators, instead of restricting to annualized income in which the temporary contribution of forests do not always feature as significant. In addition, this research identifies the implication of forest - food security linkages on the long term livelihood outcome of the smallholder farm households. The main research questions in this respect are: Do forests contribute to food security of farm households? If yes, what are the characteristics of farm households who are dependent on forests for food security? Does diversification or coping strategy motivate farm households to participate in forest environmental resource extraction? What is the implication of forest dependency on the long term livelihood outcome of the smallholders?

The third issue is related to the second issue but deals with the *contribution of forest environmental resources commercialization to the annualized income of households and with the methodological challenge in measuring the contribution of forests*. The conventional wisdom on the dependence

on forest environmental resources claims that the poor tend to depend on forests disproportionately. However, where commercial production and sales are involved, this relationship may be reversed (Vira and Kontoleon, 2010). Moreover, the household decision to participate in forest environmental resource commercialization is one of the potential sources of endogeneity in household income. The purpose of this paper is to investigate the effects of participation in forest environmental resource commercialization on welfare of households. We examine three empirical questions that are central to forests and the household welfare debate. First, does participating in forest environmental resource commercialization contribute to household income? If it does, then failing to account for differences in participation in income modelling will lead to biased parameter estimates and misleading information for policy analysis. Second, does household income actually differ between households that participated and did not participate in forest environmental resource commercialization? Third, would household income change if all households participated? While the answer to the second question is directly observable, the last question requires an analysis that accounts for the effects of participation in forest environmental resource commercialization and other unobservable factors that may influence household income. These unobservable factors may play an important role in anticipating the potential outcomes from policies designed to influence household income.

The fourth issue that this study addresses relates to the contribution of *forest environmental resources to poverty and distribution of income under the current conservation intervention in rural Tigray*. It aims to consider the interdependency of participation in forest environmental resources extraction and other household characteristics before decomposing the poverty and inequality indicators that previous studies on the area of forest income, poverty and inequality have neglected.

Finally, the fifth issue that this study deals withis the impact of social protection programs such as the *Productive safety Net Program (PSNP) in Ethiopia to reduce the cost of ex-ante risk management such as depleting forests from commons in the semi-arid region of Tigray.* It aims to investigate whether the PSNP could provide a win-win for the goals of poverty reduction and conservation of forest environmental resources.

These five issues addressed in this study provide an in-depth insight into the poverty –environment nexus in the semi- arid region of Tigray and in finding a win- win for poverty reduction and conservation of the environment.

1.4. Organization of the thesis

The motivations for, and major issues addressed in this PhD thesis are explained briefly in the preceding sections of this chapter. The subsequent chapters of this study are organized as follows:

Chapter 2 briefly provides the setting of the research area focusing on the economic situation, environment and the development and natural resource management policies and strategies in Ethiopia, and the Tigray region in particular. It also provides a description of the research methodology and analytical approaches of the research.

Chapter 3 investigates household allocation of labour for coping with health related shock and forest dependency in rural Tigray. Because illness is not exogenous, the analysis uses an Instrumental Variables approach and the Propensity Score Matching (PSM) method that capture several dimensions of illness.

Chapter 4 focuses on the forest-food security linkage in rural Tigray and its implication in the long term livelihood outcome of smallholder farm households. We use propensity score matching

methods to account for a selection bias that normally occurs when unobservable factors influence both the participation in forest environmental resources extraction and food security outcomes.

Chapter 5 investigates the driving forces behind rural households' decisions to participate in forest environmental resource commercialization and the impact of the participation in commercialization of forests on the income of the households in rural Tigray. We estimate a simultaneous equations model with endogenous switching (following Di-falco and Veronesi, 2010) to account for the heterogeneity in the decision to participate in forest environmental resource commercialization or not, and for unobservable characteristics of households.

Chapter 6 focuses on the contribution of income from forest environmental resources on poverty and distribution of income in semi-arid region of Tigray under the existing forest management regime. Counterfactual incomes are estimated based on the endogenous switching regression model applied in chapter five, taking into account what the participant group of households would have earned had they not participated. The results are compared with alternative income estimates in which forest income is treated simplistically as exogenous income.

Chapter 7 investigates the potential of Productive Safety Net Program (PSNP) in Ethiopia to provide a win-win for poverty reduction and conservation of forests.

Chapter 8 discusses the major findings and outlines the main conclusions from this study. It draws out major policy implications and discusses issues for further research.

CHAPTER TWO

THE SOCIO-ECONOMIC SETTING OF THE STUDY AREA AND RESEARCH APPROACH

2.1. Location, Demography and Natural Conditions

Tigray is the northern most region of Ethiopia located at latitude of 12 to 15 degree North and Longitude of 36 c 30" to 41 30" East and covers an area of 53,000 square kilometres (Hagos et al., 1999; and Tesfay, 2006). Tigray's population is around 4.3 million and growing at 2.5 percent, where 80.5 percent resides in rural areas (Howard and Smith, 2006). Administratively, the region has 35 *woredas*, 12 town woredas , and 665 *tabias*. Each woreda is subdivided into tabias and each tabia is divided into *kushets*(Babulo, 2007)

The landform of the region is complex, composed of highlands (2300-3200 meter a.s.l) ,Lowland plains (500-1500 M), mountain peaks (as high of 3935 m), and high to moderate relief hills(1600-2200 m). By virtue of the complexities in topography, the region has diversified agro-ecological zones and niches, each with distinct soil, geology, vegetation cover, and natural resources (Taffere, 2003). Tigray is relatively dry and is subject to frequent drought (Howard and Smith 2006). Average annual rainfall ranges between 500 and 900 mm yr⁻¹, with a unimodal pattern except in the southern and eastern highlands where a second, smaller rainy season allows local growing of two successive crops within one year (Nyssen et al., 2005). Taking into account rainfall, atmospheric temperature, and evaporation, more than 90 % of the region is categorized as semi-arid (Taffere, 2003)



Figure 1. 1 Map of the study area

2.2. Socio- Economic Aspects

2.2.1. The Economy

Ethiopia has experienced strong economic growth in recent years. With real GDP growth at or near double digit levels since 2003/04, the country has consistently outperformed most other countries in Africa and expanded much faster than the continent –wide average (Figure 2). Real GDP growth averaged 11.2% per annum during the 2003/04 and 2008/09 period, placing Ethiopia among the top performing economies in Sub-Saharan Africa (ADB, 2010).



Source: ADB 2010

Figure 1. 2 GDP Growth Rate Comparison

Although the growth performance has been encouraging, the country exhibits the highest rate of inflation⁵ in its history (CIA, 2009) and low international reserves (ADB, 2010) in the same period. High inflation is not only constraining growth but also represents a heavy tax burden on the poor and is eroding the gains made under the donor-funded social protection and social safety net programs (World Bank 2011). In addition despite the performance of the economy in recent years a number of issues warrant attention.

Ethiopia's economy is highly dependent on primary commodities and rain fed agriculture and thus is highly vulnerable to the vagaries of weather. Ethiopia has experienced no less than five major national droughts since 1980 and several local droughts (Yosuf et al., 2008). Agriculture accounts

⁵ The driving forces behind the high general inflation rate in Ethiopia have been extensively discussed by the IMF (2008).

for about 40 % of national GDP, 90 % of exports, and 85% of employment (Di-Falco et al.,2011). There is also a strong correlation between weather conditions and Ethiopian's growth performance. A change of 1 percent in average annual rainfall is associated with a change of 0.3 percent in real GDP in the following year (ADB, 2010). Only about 10 % of cereal crop lands are irrigated, and yield variability at the regional level is one of the highest in the developing world (WFP, 2009). As illustrated in Figure 3, the pace of agricultural sector growth during the 2003/04-2008/09 period declined, while the industrial and service sectors grew more rapidly (ADB, 2010). Agricultural GDP and per capita cereal production has been falling over the last forty years with cereal yield stagnant at about 1.2 tons per hectare (Di-falco et al., 2010). However, the overall expansion of agricultural production has been driven by increases in the area of land cultivated, rather than major improvements in productivity (ADB, 2010).





Figure 1. 3 Sectoral Growth Rate

In Tigray, agriculture contributes 38.7 % of the regional GDP (TRBFB, 2010). Crop productivity is very low due to factors such as soil and water degradation and erratic rainfall. Average estimates of productivity loss due to soil degradation are estimated to be 2-3% annually, which explains most of the failure to realize the potential yields expected from agricultural intensification (World Bank, 2008). Cereal yields average less than one ton per hectare in the region (Pender and Geberemedhin, 2007). This is low compared to the national average of 1.2 tons per hectare (WFP, 2009). The average land holdings in the region are around 1.25 hectare (MORAD, 2009). Major crops in the region include teff, barley, sorghum, wheat and maize (Tesfay, 2006). Rural households in Tigray do not produce enough food to feed themselves, and they get most of the balance from the market (MORAD, 2009). Given the heavy dependence of most households on buying basic food, even in years of relatively good rainfall and production, the question of how people get cash is important to food security and livelihood analysis in relation to forest environmental resources.

Tigray has much livestock including 3.04 million cattle, 2.4 million sheep and goats, 2.3 million poultry, 187,000 beehives and good potential of lakes and river fishery (WFP, 2009). However, livestock production is secondary (Tesfay, 2006). In fact Babulo et al. (2009) challenged the traditional belief in Tigray that livestock is the major contributor of the household income. According to the authors forest environmental income contributes the second source of income share to rural households in Tigray.

2.2.2. Poverty Situation

Conditions of poverty, food insecurity, and malnutrition in Ethiopia are well documented and have been the focus of development efforts for more than half a century (see for example, Dercon and Krishnan, 1998 and Dercon et al., 2005). By most indicators, poverty and ill-being in Ethiopia are appalling. Yet several surveys suggest that Ethiopians are better off than they have been. The general consensus is that Ethiopia remains one of the world's poorest countries, but that economic growth trends are positive and living standards are finally starting to catch up – at least with the rest of Africa (Devereux and Sharp, 2003). Out of 80 million (2008) people, 35 million people are living in abject poverty; most of them live in rural areas with agriculture as their main occupation, while food insecurity levels in the rural areas of the country rose from about 2 million people in 1995 to about 14 million in 2008, of which 7.5 million were covered by the safety net program of the government (WFP, 2009). Increased poverty, water scarcity, and food insecurity are just some of the negative impacts set to hit small-scale farmers and pastoralists in Ethiopia as a result of climate change in the region. While Ethiopia is no stranger to climatic variability, having suffered droughts that have contributed to hunger and even famine in the past, climate change is set to make the lives of the poorest even harder. The persistent lack of rainfall is a major factor in rural poverty. Recurring droughts leave poor farming families without food crops, causing periodic famines.

In Tigray, poverty is extremely high. More than 58% of the total population were living in absolute poverty (earning less than a dollar a day), which makes the region's situation more serious compared to the national average 44.4% (WFP, 2009).The average wealth breakdown for the 16 livelihood zones taken together is 20% very poor, 30 % poor , 34% middle and 16% better off (MORAD 2009).

A Growth and Transformation Plan (GTP) has been prepared at national level since 2011. The GTP aims to achieve the Millennium Development Goals in Ethiopia by 2015 and middle-income status for Ethiopia by 2020–2. Regional governments have also prepared similar documents based on the national GTP framework. To achieve the targets of GTP, a sustained average annual growth

rate of 11.2 percent is required, the same level as the reported growth rate under the previous 5year plan (World Bank, 2011). In addition, climate resilient green economy is seen as a real opportunity to catalyse transformation, take advantage of international interest in funding, finance a new model of development (e.g., through the fast funding), and build a resource-based competitive advantage of the country, while protecting the country against the adverse effects of climate change (ECA, 2011). This will have a strong implication for the natural resource base of the country and the livelihood of the people who are mostly dependent on it.

2.2.3. Ethiopia's Food Security Program

Given Ethiopia's history of chronic food insecurity and recurrent catastrophic famines, it is hardly surprising that food security has always featured strongly as a priority in successive government development plans and strategies. Ethiopia's current Growth and Transformation Plan for the period 2011- 2015 takes forward a number of the same measures in its predecessor the Plan for Accelerated and Sustainable Development to End Poverty(PASDEP) which prioritized food security, rural development, human development and conservation of the environment. Meanwhile, the GTP also focused on a 'Green Growth' which gives more emphasis to climate resilient development strategy. In addition to its focus on agricultural commercialization, the GTP also renews the government to the food security program (FSP), which was initiated by the 'New Coalition for Food Security' after the food crisis of 2002.

The FSP aims to address food insecurity through a package of interventions that are intended to boost agricultural productivity for the estimated 8.3 million chronically (or 'predictably') food insecure, and to provide protection against agricultural vulnerability for estimated 6.7 million transitory ('unpredictable') food insecure Ethiopians (Devereux and Guenther, 2009).

The food security program has three main components, which together are designed to attain household food security: (1) the 'Productive Safety Net Program (PSNP): with two subcomponents – public works and direct support – which bridges the food gap with cash or food transfers while building community assets; (2) Household Extension Packages, which support a range of non-farm household activities through a variety of assets and input packages for agriculture and non- agricultural activities; (3) Voluntary Resettlement Program, which relocates people from the most vulnerable highland communities to more productive land.

2.3. Land and Forest Policy Issues

2.3.1. Rural land administration

Historically, institutions/property rights to land in Ethiopia were vested in the risti system, the gulti system/ private land holding, or the church (Gebreegzabher et al., 2011). The overthrow of Haile Selassie following the 1974 famine signalled the abrupt ending of an essentially feudal system in Ethiopian agriculture. The Derg used its unfettered power to force a radical agrarian transformation on rural Ethiopia. Between 1976 and 1991, all farmland in the highlands was confiscated by the state and redistributed equally per capita within rural communities. This radical land redistribution was motivated by both egalitarian and efficiency concerns. The intention was not only to break the power of landlords over peasants, but to give all rural households the means to achieve sustainable increases in agricultural productivity and rural incomes (Devereux et al., 2005). Importantly, land was conceptualized as a safety net for rural households by the Derg, a view which is shared by the current government of Ethiopia.

The recent change in the land tenure system of Ethiopia came in the early 1990s following the change in government in 1991(Tesfay 2006). The December 1994 Constitution of the Federal

Democratic Republic of Ethiopia proclaimed that 'Land is a common property of the nations, nationalities and peoples of Ethiopia and shall not be subject to sale or to other means of transfer'. In 2005, the Government of Ethiopia issued a new *Rural Land Administration and Land Use Proclamation* (456/2005) (amended in 2007). The aim was to increase tenure security, improve productivity and avoid expectations of land re-distribution. Article 6(3) of the proclamation states that land holders will be issued with certificates that indicate the size and fertility of their holding as well as its borders (FDRE, 2005). Land certification is currently taking place in the Amhara, Oromia, SNNP and Tigray Regional States. However, the land registration activities have commenced without due regard for the sustainable use of forest resources because forests are not yet mapped and registered and rural households are able to clear land and stake a claim to it before the registration process begins(Abebe et al. 2009). Conditions of tenure security and its impact in various aspects in Ethiopia, particularly in Tigray region is well documented and has been the focus of research for decades (Pender et al., 2002; Pender and Gebremedhin, 2003; Hagos and Holden 2006; Deininger et al., 2009; Tenaw et al., 2009 and Ghebru and Holden, 2013).

2.3.2. Forest Policy

The new federal *Forest Development, Conservation and Utilization Proclamation* (542/2007)provides the framework for making fundamental changes to forest resource management across the country. Both the proclamation and the country's first policy and strategy on forest development that accompanies it demonstrate a much greater acceptance of community management, and the conceptual links being made between livelihoods and environmental resources (FDRE, 2007a; 2007b).

The overall objective of the new policy is '... to meet public demand in forest products and foster the contribution of forests in enhancing the economy of the country through appropriately *conserving and developing forest resources.*' Significantly, this is to be achieved through an overall reduction in the role of the state, the promotion of private investment and the devolution of authority to regional administrations (FDRE, 2007b). To this end, the new proclamation allows for forests to be designated as either private or state owned. State forests constitute natural high forests preserved for biodiversity conservation and other purposes in which the property rights are vested in the state. It could be either national forestry area (NFPAs) or regional forestry priority areas (RFPAs). Cross boundary forest areas also fall within the category of state forests. The first priority within the context of these high forests is protection and conservation. State forests may be given as concessions for privately managed plantations or retained for conservation management by government organizations in participation with local communities (FDRE, 2007a).

The government of Ethiopia had classified 58 Forest Priority Areas (FPAs) covering 2.8 million hectares in the 1994 Ethiopian Forestry Action Programme (UNDP-GEF, 2005). Out of these 58 FPAs, 2 of them are found in the Tigray Regional State (Gebreegziabher 2007). However, the state forests appear to function more like unmanaged state forests with multiple resource uses (grazing, settlements, agriculture, fuel wood gathering etc), with low priority given to conservation and preservation of wild life and habitats(USAID, 2008). No official maps exist, no management plans have been prepared, and none have been gazetted. Regional and federal resources to delineate state forests and prepare management plans are extremely limited. Currently, the land registration process does not register individually held land located within high forests on the assumption that these areas are state forests. The policies do not refer to the location of these forests or explain how they can be identified implying that there are no legal grounds to resolve any conflict regarding the boundaries and locations of these state forests (USAID, 2008).

2.3.4. National Priority Areas (NPAs) in Tigray

With about 120,000 hectares, Desse'a is probably the largest natural forest in Tigray National Regional state. The forest stretches in a narrow corridor along the eastern escarpment from close to Senquata town near Adigrat to the east of Quiha town⁶. The topography in Dessa'a is varied and includes some flatter areas and gentle slopes as well as steep scarps. The forest extends in an easterly direction along and down the escarpment, and forms a climatic buffer zone between the cool highlands of Tigray and the hot lowlands in Afar Region. Most of the trees in the forest are found at the higher altitudes, and relatively undisturbed forest exists only on the steep eastern slopes. It appears that there is currently no natural forest regeneration. Grazing by domestic animals is given as a main cause for this, but the area is now very dry and climatic changes should not be ruled out.

On the other hand, Hugumburda and Grat-Kahsu are two contiguous forests situated between the towns of Mai Chew and Alamata, Southern Zone. Local people make extensive use of any easily accessible areas of forest to provide fuel wood and construction materials. Until the early 1990s, this part of Tigray was better known for its huge camps of famine victims than its natural resources, so it is not surprising that the forests have not been properly managed. Afforestation activities on the more accessible slopes began in the early 1970s, and the regional government is now helping to expand these further. Three tree nurseries in Korem, Ashenge and Addis-Fana are producing exotic species for the afforestation program.

⁶<u>http://www.cbnrm.net/pdf/sussex_001_briefing_et18.pdf</u> (Accessed on December 2011)

2.4. Research Methodology

2.4.1. Data sources and Description

Chapters 3 -7 of the thesis are based on household survey data. Primary data were collected via survey questionnaire. The primary sampling units were tabias within approximately 5 km radius of the National Priority Areas (NPAs). This distance is purposely chosen for convenience and since an empirical evidence from by (Guthiga and Mburu, 2006) also indicated that there were progressively fewer people that extract non-timber forest products beyond 5km stretch from forest in rural Kenya. Accordingly, a total of ten tabias, namely Arato, Derga -ajen, Hugumrda, Meswaeti, Kara_adishawo, Worebayu, Kal_amin, Kelisha_emni,andFelege-woini were selected for the survey (see figure 2.1).

Villages were randomly selected from each tabia. In each village 26 households were selected randomly, yielding a total sample of 260 households. However, due to budgetary and logistics problems the final survey ended up with 254 households. A multi-purpose questionnaire was used to gather information on household income and expenditure, farm and off farm activities, household assets, vulnerability to idiosyncratic and covariate shocks, participation in local institutions and a host of other information related to production and sales, climate change adaptation, participation in the Productive Safety Net Program (PSNP) and social networks. The survey was conducted from June-August 2010. The survey paid special attention to labour allocation to forests, participation in forest environmental resource extraction from private sources, enclosures, woodlots, and other common pool resources. Eight experienced enumerators who had served in similar surveys were hired for the data collection. The enumerators were given two days intensive training and one day pilot testing was made. During the survey field work, close and regular supervision was made. Filled out questionnaires were checked on the spot and those with
significant inconsistencies were made to be filled again. Survey data was processed using STATA 10 software. To determine the extent to which households in rural Tigray use forest environmental resources for their livelihood and their dependency in forests, we follow (Narrain et al., 2008) to calculate the income that each household obtained from major sources as follows.

2.4.2. Income Accounting

To determine the extent to which households in rural Tigray use forest environmental resources for their livelihood, we calculate the income that each household obtained from 8 major sources namely, (1) agriculture, (2) livestock, (3) forest environmental resources, (4) household business, (5) wage employment, (6) Productive safety net program, (7) financial transactions and (8) transfers.

Income from each source is calculated as the difference between total revenue obtained and total input costs incurred, where these totals include both market transactions and imputed values from non- market transactions.

Income definitions

Net crop Income: is defined as the difference between the revenue obtained from all crops and crop-residues harvested by the household using January-March 2010 local market price and the input costs incurred for crop production. Input costs, in turn are defined as the sum of wages paid to hired agricultural labour, cost of fertilizers, manure, pesticides; rent paid on land rented in; and rent paid for farm capital rented in. We also include income obtained by the household from trees on its private lands, equal to the revenues from fuel and construction wood, flowers, fruits extracted. Finally, we added any income from rental out of household's farm capital.

Net livestock income: is defined as the difference between revenue obtained from sales, products and services of the main types livestock found in Tigray, namely, cows, goats, sheep, donkeys, horse, camel and chickens and costs that include cost of veterinary services, hired labour to graze them and the cost of fodder to feed them that includes the imputed value of fodder grown as a crop and not sold, residue from other crops used as fodder, fodder collected from village commons and not sold, and fodder bought from the market.

Forest Environmental Income: To identify the forest environmental resources that rural households in Tigray collect, we prepared an open ended question during the pre-testing of the household survey. Then we listed all the main forest resources that the households use in the questionnaire. Our survey also asked households to list other forest resources obtained if any. We follow an approach by Babulo (2007) as a basis of valuation. We relied on household's own reported values for environmental goods from homestead, enclosures and other community owned forest resources and grazing fields. Respondents reported the weekly, monthly or yearly amount of each product harvested/ gathered by the households (gift), the amount consumed/used, and the amount exchanged, the amount given to other households (gift), the amount sold, and the cash amount received from sales of these products. In case the forest products were bartered rather than sold, the retail sale price of the exchanged commodity was recorded as the cash income. But only in rare instances did the barter occur in our survey.

Income from Household Business: is defined as income from any non- agricultural business such as trade operated by the household.

Wage income: is defined as the sum of income in cash and in-kind wages received from invillage casual employment off the household's own farm, off-village casual employment and regular employment in the private and public sector. **Net income from financial transactions**: Households in our sample own a variety of financial assets, including deposits at Dedebit micro finance, saving and credit associations, cooperatives, and loans given to relative or friends. They also owe debt to a number of sources - Dedebit micro finance, saving and credit associations/cooperatives, moneylenders, friends or relatives. During the survey year, households earned interest income on their deposits and paid out interest on their debts. Net interest (interest income earned less income paid out) constitute household's income from financial transactions.

Income from Transfer: is defined as the sum of PSNP and in-kind payments received by a household from its family, friends, the state, and any non–governmental organizations operating in the area.

2.4.3. Analytical Approach

The econometric analysis for this study mainly uses cross-sectional data collected in the year 2010. The hypotheses in each of the topics studied are derived from a review of theoretical and empirical literature on the respective subjects. For the econometric analysis the choice of feasible techniques is dictated by the nature of the dependent variable and research question. In each chapter an explanation is given for the choice of the econometric model. For example, in chapter 3, we used a Heckman selection model. In chapter 4, we also use PSM approach to control for the selection problem and endogeneity of participation in forest environmental resource extraction. We use the Full Information Maximum Likelihood Estimation in chapter 5. In chapter 6 we use Foster, Greer and Thorbecke (1984) poverty indices (commonly known as FGT indices) and Gini decomposition (Stark et al., 1986). Applying these analyses for "with forest income", "without forest income" and "counterfactual income" we were able to measure the contribution of forest income to rural poverty and inequality, using analytical methods which some previous studies have

not used, as a result of which their estimates of forest income –poverty and inequality may have been overestimated or underestimated. Finally, we use the Propensity Score Matching (PSM) in chapter 7 which is the most recent research tool in the program evaluation literature.

CHAPTER THREE

DO FORESTS HAVE A ROLE TO PLAY FOR COPING WITH IDIOSYNCRATIC HEATH SHOCKS IN RURAL ETHIOPIA?

3.1. Introduction

Low and volatile incomes coupled with the absence of or poor development of financial or risk sharing institutions make consumption smoothing an important issue in low - income countries like Ethiopia (Barrett et al. 2001). A typical household in rural areas of developing countries faces substantial idiosyncratic and covariate risk, resulting in high-income variability (Dercon et al., 2005). Covariate risks include uncertainties associated with nature, markets (both input and output), social unrest, policy and institutional failures etc. On the other hand, idiosyncratic risks include shocks related with income failure, illness, and shortage of agricultural inputs, (Weinberger and Jutting 2000). Usually, due to the absence of the first best solutions (formal risk sharing institutions), rural households in developing countries have developed their own risk reduction, mitigation, and coping strategies (Weinberger and Jutting 2000). Coping, broadly defined as a short-term strategy of households to prevent a negative effect of crises (Sauerborn et al., 1996) may take several forms. Some of the most common coping mechanisms in rural areas include: livestock husbandry (Reardon et al., 1992), reduction in consumption (Porter, 2008), sale of assets (Hoddinott, 2006), off-farm activities (Barrett et al. 2001) and social ties (Dercon et al., 2005).

In the livelihoods literature, forests are also identified as a prominent safety-net source, accessed principally by reallocating more labour to forest extraction (Wunder et al., 2014). This may be

because forest products are often available at times when other income sources are not, for example, when crops fail (Byron & Arnold, 1999; Pattanayak & Sills, 2001; Angelsen et al., 2007), and have better insurance properties than insurance markets in the presence of information and enforcement problems (Baland & Francois, 2005) which is a typical situation in developing countries like Ethiopia. The importance of resource extraction is amplified in the presence of risk, and such risk is expected to intensify as future climate change precipitates more extreme weather events, especially in marginal agricultural areas (Sivakumar et al., 2005 cited in Angelsen et al., 2007) such as northern Ethiopia.

Yet the empirical literature on biodiversity as a means of risk coping is ethnographic (McSweeney, 2004), considerably smaller than that on biodiversity as a source of livelihood (CBD, 2010, Vira and Kontoleon, 2010) and typically relates only extraction volume, income, or income share with total income or asset holdings (e.g., Cavendish, 2000; Coomes et al., 2001; Godoy et al., 1995). As a result, there is very little systematic analysis to help guide conservation and development promoters in their efforts to understand how the poor (and others) deal with negative shocks (McSweeney, 2004).

A few existing studies such as those by Pattanayak & Sills (2001), Takasaki et al.,(2004), McSweeney (2004), Debela et al., (2012) find positive correlation between shocks and forest extraction. For instance, Pattanayak & Sills (2001) argued that forest environmental resource extraction is correlated positively with agricultural shortfalls and expected agricultural risk in their study of the Brazilian Amazon. In the same way, Debela et al. (2012) argued that shocks tend to lead rural households in Uganda to depend on forests to meet subsistence and cash needs. Similarly, Takasaki et al. (2004) conclude that forests serve households in Peru to cope with flood shocks. On the other hand, Wunder et al, (2014) using global comparative data sets conclude that

forest resources may be less important as a buffer between agricultural harvests and in times of unforeseen hardship than has been found in many case studies. Drawing overall conclusions from these literatures is difficult due to wide differences across studies in fieldwork method, variable definitions (e.g., the type of shock experienced), and empirical model (Angelsen et al, 2007). For example, nearly all studies devoted to the link between shocks and forests have centred on Latin America (Pattanayak & Sills, 2001; Takasaki et al., 2004 and McSweeny, 2005) and Asia (Shively, 1997; Takasaki, 2009 and Volker & Waibel, 2010). Evidence from sub-Saharan Africa is much more limited. Important exceptions include Monica and Fisher, (2005); Debela et al. (2013). Thus, it is important to examine the role of forests as a risk coping strategy in an African setting. Our contribution here is that the present study was undertaken in another tropical ecosystem, dry deciduous woodlands, the dominant type in Southern and Eastern Africa (Campbell et.al, 2006 and Fisher and Shively, 2005).

There are also additional features that make this research work especially interesting. First, much of the research done on the insurance value of forests deals with covariate shocks (see for example Pattanayak & Sills, 2001; Takasaki et al., 2004; and Takasaki, 2009). Meanwhile, a growing body of empirical evidence suggests that idiosyncratic risk may be as important, or indeed may dominate covariate risk in rural Africa and Asia (Udry, 1990; Townsend, 1995; Deaton, 1997; Lybbert *et al.*, 2004; Morduch, 2004; Dercon, 2005; Kazianga and Udry, 2006). This relative importance of idiosyncratic risk and the relative dearth of attention given to it by researchers and policymakers alike raise the possibility of significant untapped potential for improved local risk management in developing countries, and significant impact for high quality research that tackles this topic in the African setting. Many critical questions remain: for example, do rural households cope with idiosyncratic health shocks of household members by increasing the rate of forest extraction? And,

if they do, what are the characteristics of households most reliant on forests for coping with idiosyncratic health shocks? Does reliance on forests for coping with idiosyncratic health shocks depend on the endowment of physical and human capital as well as access to institutional services? Answering these and other questions can enable us to develop recommendations that maximize the potential for "win – win" for both human and environmental health. Policy options on this front could be of two types, i.e. those policies to mitigate interactions that both aggravate the impacts of health shocks on households and concomitantly affect forests negatively. Thus, the primary objective of this paper is to examine the potential role of forests for coping with idiosyncratic health shocks of farm households living at the margins of protected forests, found in the north highlands of Ethiopia.

Dealing with health shocks is an important issue through which to view the poverty-environment nexus in many ways. First, many of the environmental hot spots in the world are located in impoverished regions where individuals live under the constant threat of serious illness (Rasmus and Lund, 2009). Second, the environmental impacts of health shocks are theoretically ambiguous (Volker and Waibel, 2010).Resource extraction and activities that use these resources for production tend to be labour intensive, suggesting greater environmental conservation in the face of significant morbidity. On the other hand, health may also affect discount rates households use when making trade-offs over time. Altered planning horizons due to shortened life expectations can undermine incentives to conserve, while increased medical costs and caloric needs can force households to liquidate capital, both physical and natural. As morbidity and mortality also decrease income, families may increasingly turn to less sustainable activities such as hunting, logging, and charcoal making for subsistence needs, precipitating environmental degradation (Joshua et al., 2010). Finally, poor households are more exposed to health shocks such as illness and death of a household member than wealthier households (Tongruksawattana et al., 2008; Rasmus and Lund, 2009). In some circumstances, these households are even more fragile to health shocks than to crop income shocks (Kochar, 1995). As such, this poverty-health-environment link seems to limit the prospects for win-win outcomes of conservation-development initiatives proposed by policy makers and NGOs, especially if negative shocks of various sorts are likely to propel the poor to hit nature hard (Barrett and Arcese, 1998). Knowledge about the mechanisms is important for policy purposes. For instance, if differential access to health service were the main reason for the observed gradient, then improved access to health care for people around the forest margin would be one policy option one could use in order to weaken the gradient.

This chapter is structured as follows: section 3.2 provides an overview of the health and environment nexus in Ethiopia. Section 3.3 outlines the econometric procedure. Section 3.4 discusses the empirical results, while the final section draws concluding remarks.

3.2. Overview of the Health and Environment Nexus in Ethiopia

Millions of poor people in Ethiopia still depend on ecosystems and natural resources for their income and livelihoods. However, economic and political processes, often beyond their control, continue to degrade these resources in much of the region (Howard and Smith, 2006). There has been progressive degradation of forest resources in Ethiopia. Between 1990 and 2000, 141,000 ha of forest were lost every year, which equals an average annual deforestation rate of 0.93% (Babulo, 2007). Between 2000 and 2005, the rate of deforestation increased by 10.4% to 1.03% per year, which totals to 14% (or around 2,114,000 hectares) of forest cover loss in the 15 years between 1990 and 2005 (FAO 2007).

The challenge of poverty alleviation, promoting the conditions of improved health, securing the basis of rural households, and conserving the environment, is particularly acute in Ethiopia. The

country is home to about 94 million people many of whom still live in poverty despite recent economic growth. The health system of Ethiopia is still in a transitory stage and hence fragile. The Ethiopian population faces high rates of morbidity and mortality mainly resulting from a high prevalence of communicable infections (WHO 2009). About 75% of the population suffers from some type of communicable disease and malnutrition, which are potentially preventable. The leading causes of disease and death are malaria, bronchopneumonia and tuberculosis. Widespread poverty, along with generally low income levels, low educational levels (especially among women), inadequate access to clean water and sanitation facilities, and poor access to health services, have contributed to the high burden of ill-health in the country. Health expenditure in Ethiopia is still high and out of pocket payment for health care in relation to incomes and the total health expenditure is catastrophic⁷.

Moreover, health care is predominantly a forest – based service in Sub-Saharan Africa (Barany et al. 2001) including Ethiopia. Traditional healers are the dominant providers of medical care in forested areas, often providing between 70 and 95 % of primary health care (Colfer et al., 2006). Because of remoteness of many rural, and particularly forest dependent, communities, a lack of access to health care could be the primary cause for increased reliance on forest medicines and traditional healers. Whether the change in forest resource availability is sudden or gradual, forest degradation has potentially negative implications for households affected by illness. Such impacts could include: loss of income generating options; increase in labour required to collect forests; reduction in use of firewood (possibly leading to inadequate energy to meet household needs including cooking and sanitation); and reduced access to medicines (Barany et al.2005). For

⁷ The economic burden of illness in terms of reduced per capita expenditure and farming productivity is well documented in Dercon et al.,(2005) and John (2009) respectively.

example, if important medicinal plant species become scarce, the results for an ill individual could be a loss of stamina and a compromised immune system. The repercussions of this cycle then could be a continued decrease in stamina, increased vulnerability to further infections and other diseases, and fewer livelihoods and food security as well as a reduced natural resource base. Thus, in rural areas such as Ethiopia, developmental and human health gains depend on ecosystem services (Costanza et al., 1997; Daily, 1997) and environmental sustainability is a criterion for population health (McMichael, 2002).

3.3. Theoretical model for Shocks and Labour allocation to Forests

The theoretical framework underlying the empirical analysis of forest extraction as a coping mechanism is based on "new home economics" theory (Becker, 1965) adapted after Volker and Waibel (2010) and Ellis (1993). The emphasis of the "new home economics" theory is on the household's time allocation assuming that labour is the major factor of production. Assuming there is no access to off-farm wage employment opportunities, figures 3.1 and 3.2 give a graphical presentation of the basic home economics household model showing household output in relation to available time. Time, depicted on the horizontal axis of the graphs, is divided into labour for agriculture (0TA), forest extraction (TATF) and leisure (TFT). The time constraint is determined by the total number of person days available for agricultural production, forest extraction and leisure. The graphs depict two scenarios of household labour allocation comparing a situation with and without shocks. Figure 3.1 represents a weather shock scenario with effects on agricultural output while Figure 3.2 represents a scenario where a family member is falling sick with corresponding effects on labour capacity.



Figure 3. 1 The Household Production Model under a weather Shock Scenario

To explore in more details the effects of a covariate weather shock on the time allocation of the household Figure. 3.1 is used as a framework. Weather shocks such as heavy rain or flooding lower agricultural output. A key assumption of the model is that weather risk associated with agricultural production is uncorrelated with forest extraction risk (as also suggested by McSweeney, 2004) and that even for weather risks like storms the effect on forests is comparatively small (unless it is an extremely strong typhoon) due to the diversity of products which can be extracted. Therefore the effect of a weather shock is modelled solely by its effect on the household's agricultural production function while the household's average returns to forest extraction (ARF) remain unaffected. Weather shocks reduce the household's agricultural production output in the form of crop yields and livestock products, thereby decreasing labour productivity. The shape of the agricultural

production function changes (APF to APF') and reflects a relatively poor input-output response as compared to the baseline scenario. Using the same amount of time for farm work, the household produces less output because the marginal productivity of time allocated to agricultural production has decreased. This results in a new equilibrium of the household in agricultural production (point A' instead of point A), which shows that the household allocates less time to agriculture (0TA' instead of 0TA) and accordingly more time to forest extraction (TA'TF instead of TATF). The time allocated to leisure remains the same as neither the indifference curve nor the average returns to forest extraction are assumed to be altered by the weather shock. Consequences of ex-ante weather risk can be modelled accordingly by assuming that households make time allocation decisions based on expected states of nature, meaning that pessimistic households face a relatively poor subjective ex-ante input– output response curve regarding agricultural production as compared to more optimistic households.



Adapted after Ellis (1993) and Volker and Waibel (2010)

Figure 3. 2 The Household Production Model under a health shock scenario

Figure 3.2 analyses the effects of a health shock. Health shocks relate to active household members who become ill. Illnesses of household members are assumed to have two effects. On the one hand, they reduce the total household time capacity if the affected household member is not able to carry out the normal amount of work. This is reflected by a shortening of the horizontal axis to the left. On the other hand, the household faces additional needs for health care which increases the family's preferences for consumption goods instead of leisure. This is reflected by the shape and position of the consumption– leisure indifference curve which has a shallower slope than in the baseline

scenario (I' instead of I). Provided the household is able to reallocate leisure time to forest extraction, a new equilibrium in the consumption of goods emerges (point B'), which shows that the household allocates less time to leisure (TF'T' instead of TFT) and accordingly more time to forest extraction (TATF ' instead of TATF). The optimal production level of consumption goods (point A) remains the same as both the agricultural production function and the average returns to forest extraction are assumed to remain unaffected by the health shock. Again, consequences of respective ex-ante risk can be modelled accordingly. Households who have pessimistic expectations about future health shocks might face a shallower income–leisure indifference curve as they may want to accumulate savings for future times of hardship. The theoretical model suggests the hypothesis that a covariate weather shock and an idiosyncratic health shock, as well as decision makers' future expectations of such risks will make the household allocate more labour to forest extraction.

3.4. Econometric Methodology

3.4.1. Instrumental Variables (IV) Approach

Our main objective in this chapter is to measure the role of forests in coping with idiosyncratic shocks. The main econometric issue that arises in attempting to identify this role is the potential endogeneity of those shocks. For instance the standard probit estimate of the effect of weather shocks on forest labour supply could suffer from endogeneity bias as forest extraction in previous periods may determine both the frequency of weather shocks (because forest extraction deteriorates the erosion protection function of forests) and the probability of engaging in forest extraction (Volker and Waibel,2010). In addition, the perception of being sick or being healthy can lead to a significant measurement error problem. If measurement error is random, then we do

not need to worry about this. However, it is possible that the likelihood of reporting illness is closely related to the socio-economic status of the household (for example the income of the household or the education level of the most educated member of the family). In fact, Dercon et al. (2005) argued that illness shocks appear more important for richer households as measured by relative landholding in Ethiopia.

Additionally, our sample consists of households who have been exposed to the treatment (a health shock) and those who have not been. If households in the treatment group have a better knowledge about how to prevent sickness, or have better coping strategies because of training provided by the health service provider, then we could expect that either households in the treatment group are systematically less exposed to shocks or even when they experience such a shock we would not observe significant difference in coping strategies because of the specific awareness given to them. The presence of this unobserved heterogeneity could bias our results, so to deal with it we need to search for a variable that is correlated with the shock but does not directly affect participation in forest environmental resource extraction, and is not correlated with the idiosyncratic and covariate error term. Accordingly, we follow an instrumental variables (IV) approach to overcome this problem. The instrumental variables that we use are (1) Kebele (Tabia) rainfall, and (2) distance to health center. Rainfall is a good instrument because it is an exogenous variable to the households and cannot directly affect the outcome variable. Distance to clinic is also selected as an instrument, as it is assumed to be a significant determinant of households' reporting of illness shocks. However, it is unlikely that distance to clinic has a significant direct effect on forest activity.

For the participation in forest sales, we used an IV Probit since the dependent variable is binary: whether the household reported participation in forest commercialization (=1) or not (=0).

However, forest dependency and number of trips for forest collection can take a value of zero with positive probability, as many households do not participate in forest environmental resource extraction (in our sample almost 15% of the households do not participate in forest extraction). In order to deal with a censored sample we conducted estimation using instrumental variables two-stage least squares.

3.4.2. Using Alternative Estimation Techniques

To supplement the results from IV, we also chose to adopt the propensity score matching (PSM) strategy of Rosenbaum and Rubin (1983) that is now widely used in the program evaluation literature⁸. Typically, we would expect that the likelihood of reporting illness is closely related to individual/household characteristics. We therefore match households based on demographic, asset and institutional variables. This is essentially the route taken by Smith (2004) in that the onset of health shock is assumed exogenous (conditional on a set of observed covariates) in the labour outcomes equation.

Our empirical strategy relies on the possibility of conditioning on sufficient observable information to obtain a credible counterfactual against which we may measure the impact of the health shock. Let T =1, 0 indicate treatment (health shock) and control (no health shock) respectively and let Y_1 and Y_0 denote the outcome of interest (number of trips to forest) for households with treatment and without treatment respectively. Since we observe households to be

⁸ In our case, PSM compares households who reported illness to those that did not, with the same (or similar) values of those variables thought to influence both illness and coping strategy. We can think of households reporting illness in our sample as the treatment group and the households that did not as the control group, following the program evaluation literature. Under the matching assumption, the only remaining difference between the two groups is reported sickness. Any difference in outcome between these two groups can be entirely attributed to the sickness effect provided we have made sufficient arguments to guarantee that there are no further systematic differences between these two groups.

either with treatment or without treatment, we cannot observe the causal effect of interest: $Y_1 - Y_0$. Some features of this distribution are estimable, nevertheless. In particular, we may consider the Average Treatment Effect of the Treated (ATT):

$$ATT = E(Y_1 - Y_0 | T = 1)$$
(3.1)

This magnitude measures how much the outcome of interest changes on average for those households who undergo the treatment (who suffer the health shock to be defined below). Clearly, simply computing the difference in the average outcomes of those under treatment and those not under treatment is open to bias, as there are observed and unobserved characteristics that determine whether the household undergoes the treatment. That is,

$$E(Y_{1} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} | T = 1) - E(Y_{0} | T = 1) + E(Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) + E(Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

$$E(Y_{1} - Y_{0} | T = 1) - E(Y_{0} | T = 0) =$$

Only if we can guarantee that outcomes of the control group are equal on average to what the outcomes of the treatment group would have been in the absence of treatment does this consistently estimate the ATT. With non-random sorting into treatment and control, this condition is rarely met.

Now suppose that by conditioning on an appropriate set of observables, X, the non-participation outcome Y_0 is independent of the participation status T. This is the weak version of the unconfoundness assumption, also called ignorable treatment assignment (Rosenbaum and Rubin,

1983) or conditional assumption (Lechner, 2000) or selection on the observables, which suffices when the parameter of interest is the ATT, as only assumptions about the potential outcomes of comparable individuals are needed to estimate counterfactuals.

$$Y_0 \perp T \mid X \tag{3.3}$$

This implies that

$$E(Y_0 | T = 1, X) - E(Y_0 | T = 0, X) = 0$$
(3.4)

In order to identify the ATT, the overlap or common–support condition is also assumed. It ensures that, for each treated household, there are control households with the same X.

$$Pr(T=1 \mid X) < 1 \tag{3.5}$$

Therefore, under the assumptions stated in equation (3.3) and (3.5) above, we could estimate the ATT from the differences in outcomes between treated and controls within each cell defined by the conditioning variables (Blundell and Costa Dias, 2002). Using the law of iterated expectations and the conditional independence assumption, the ATT can be retrieved from observed data in the following way:

ATT =
$$E(Y_1 | T = 1) - E(Y_0 | T = 1)$$

= $E_X \left[\left(E(Y_1 | X, T = 1) - E(Y_0 | X, T = 1) \right) | T = 1 \right]$
= $E_X \left[\left(E(Y_1 | X, T = 1) - E(Y_0 | X, T = 0) \right) | T = 1 \right]$
(3.6)

The estimate of ATT as shown in equation (3.6) turns out to be prohibitive in terms of data when the set of conditioning variables X is large. An alternative is to use the results of Rosenbaum and Rubin (1983, 1984) and condition on the probability of treatment as a function of X, the propensity score P(X), since the conditional independence assumption also implies that

$$E(Y_0 | T = 1, P(X)) - E(Y_0 | T = 0, P(X)) = 0$$
(3.7)

Therefore, we could estimate ATT from the differences in outcomes between treated and controls within each cell defined by values of P(X).

ATT =
$$E(Y_1 | T = 1) - E(Y_0 | T = 1)$$

= $E_{P(X)} \left[\left(E(Y_1 | P(X), T = 1) - E(Y_0 | P(X), T = 1) \right) | T = 1 \right]$
= $E_{P(X)} \left[\left(E(Y_1 | P(X), T = 1) - E(Y_0 | P(X), T = 0) \right) | T = 1 \right]$ (3.8)

Provided that the conditional participation probability can be estimated using a parametric method such as a probit model, matching on the univariate propensity score reduces the dimensionality problem.

To estimate the propensity score we estimate a probit model with binary dependent variable whether a member of household was reported to be sick (=1) or not (=0) using our sample data. So, we control for unobservables that may influence households' reporting of sickness. We then discard observations that do not have any common support and observations with households having very low or very high probability of sickness. We consider nearest neighbourhood and kernel matching. Thus we combine matching with the IV approach (due to measurement error) to estimate the effects of health shocks, and the role of forests in mitigating the consequences of shocks or serving as a coping mechanism for health shocks.

3.5. Results and Discussion

3.5.1. Descriptive Statistics

Table 3.1 presents the definitions and sample statistics of the variables used in the analysis. Panel A (Table 3.7) shows the treatment variable. We use self- reported health shocks: respondents in our survey were asked about new or ongoing and past illness of all household members in the household. This measure, while being simple to understand and compute is not particularly informative because of its binary nature. The problem is that an individual's self-reported health status is subjectively affected by an individual's social and cultural background, given the individual's subjective health. Schultz and Tansel (1997) argue that this is because of "cultural conditioning": the threshold of what is considered good health varies systematically across society, controlling for their subjective health status. For example, individuals who are more educated are wealthier and from socially advantaged groups, are typically more aware of the limitations imposed on them by their health status and are more likely to report themselves (and their family) as being of poor health. To control these unobservable characteristics of reporting health related shocks to the researcher, we used a rich data set of control variables based on the literature. 19 % of sample households report that some member of their household was affected by health related shocks in the survey year. The average frequency of shocks due to illness was 2.12.

<< INSERT TABLE 3.1 ABOUT HERE >>

Table 3.1 also presents descriptive statistics on other socio-economic and demographic factors. On average the age of the household head is 46.7 years and she/he has less than 2 years of formal education. Almost 85% of the households in the sample are male headed. The average size of the household is 5.74, while the number of adult male and female household member is 1.37 and 1.49

respectively. In terms of asset holdings, the average size of land holding in Tsimdi (= 0.25 hectare) is 4.36 while the average number of livestock owned in TLU is 3.11.

<< INSERT TABLE 3.2 ABOUT HERE >>

Also presented in table 3.2 is the difference in means of the variables for households affected by health related shocks and those that were not, alongside their significance levels. The significance levels suggest that there are some differences between the two groups with respect to sociodemographic characteristics. With regard to the outcome variables, there appear to be statistically significant differences between both groups. For example, the average forest income for households that have been affected by health related shocks was 3427 Ethiopian Birr, while the corresponding figure for households that did not report the same shock was 2534 Ethiopian Birr. In terms of forest dependency, households affected by the shock earn 31 % of their income from forests, while the same figure for the latter was only 24%. Finally, there is also a significant difference in the average trips for collection of forest environmental resources. On average households affected by health related shocks made 92.9 trips to collect forest environmental resources, while the corresponding figure for the control groups was 70.6.

<<INSERT TABLE 3.2>>

The findings from the previous section that simply compare mean differences in the outcome variables and other socio-demographic variables between the two groups suggest that households that are affected by health related shocks are generally more dependent on forest environmental resources than their counterparts. Given that comparisons of mean differences do not account for the effect of other characteristics of farm households, they may confound the effect of health shocks in the outcome variables with the influence of other characteristics. Multivariate

approaches, that account for the selection bias arising from the fact that households reporting illness shocks and those that did not may be systematically different, are essential in providing sound estimates of the health shocks and participation in forest environmental resource extraction.

3.5.2. Instrumental Variable Results

The results of the IV Probit model⁹ for participation in forest commercialization are shown in table 3.3. Our parameter estimate indicates that age of the household head is negatively correlated with forest environmental resource commercialization. This is expected since the physical demands of forest activities should make forest commercialization less accessible to elderly households. The number of adult male households and dependency ratio are positively correlated with participation in forest related activities. With more adult members, households may have access to labour that could be involved in forest activity. Households with more dependents have higher food demand above what they produce, and depend on forest resources commercialization. This is also expected since jewellery is a relatively liquid asset that can be sold in response to price signals to smooth consumption, or to provide financial capital to start business. Households who are members of the household extension package, and have their own business, are less likely to participate in forest resource commercialization. These highly and statistically significant results of the two variables

⁹The instrumental variable regression is a two-stage regression where the first –stage regresses health shock on the instruments, in addition to other covariates. The second stage IV regression uses predicted values of health shocks from the first stage regression to estimate participation in forest activities.

support the idea that the household extension package that provides credit in cash or in kind helps poor households to diversify income and free up other sources of financing that can be used to directly smooth consumption. Access to irrigation was found to have a negative and significant effect on forest resource commercialization. This may be because with irrigation households can harvest more and engage themselves in high yielding agricultural activities instead of forest resources extraction which has been termed " an activity of the last resort" (Angelson and Wunder 2003).

<< INSERT TABLE 3.3 ABOUT HERE >>

Results from table 3.4 also show that forest dependency (share of forest income to the total household income) is negatively correlated with age and education of the household head, access to irrigation and liquid assets, ownership of business, access to transfers, extension package and awareness of climate change, while it is positively correlated with the number of dependents at the household level. A similar trend was also observed in table **3.5**.

The final step of the empirical analysis is to examine the effect that health-related shocks have on the probability of participating in forest resources commercialization. To do so we include in the estimations a dummy variable (ill_hhs) that takes the value of one, if according to the information obtained from the household questionnaire, the household were affected by health related shocks. Nevertheless, it could be argued that the causality could go the other way as participation in forest resource commercialization might affect the probability that a given household member is affected by health related shocks. For example, one could think that forest resource commercialization in a previous period may determine both the frequency of health shocks (because forest extraction deteriorates the quality of forests¹⁰ and causes deforestation¹¹) and the probability of engaging in forest resource commercialization. Simply put, there is a potential endogeneity¹² problem when the variable (ill_hhs) is included in the econometric analysis. In fact, the Wald test for the variable (ill_hhs) shows that the variable is endogenous.

Turning to our variable of interest, health shocks, we find that health shocks related to a household member has a highly significant positive impact on participation in forest environmental resources sales. This echoes previous studies of the "Natural Insurance" literature (Pattanayak and Sills, 2001; McSweeney 2004; Takasaki et al.2004; Fisher et al. 2010) and may reflect that good health signals to prospective employers an individual's potential productivity, increasing the likelihood of being hired in relatively remunerative labour markets rather than engaging in forest related activities.

<< INSERT TABLE 3.4 and 3.5 ABOUT HERE >>

3.5.3. Propensity Score Matching Results

3.5.3.1. Propensity Scores

¹⁰Villages with a greater percentage of their forests under primary forest cover (as opposed to regenerating, or secondary forests, or agricultural lands) have lower incidence of child malaria when compared with villages with lower percentages of primary forests.

¹¹Deforestation can also affect the local climate and thereby affect the spread of diseases by reducing moisture held by the vegetation, and raising ground temperatures.

¹²A wald test examines evidence of the correlation between unobserved explanatory variables from the equation to test the null hypothesis that there is no endogeneity bias (ill_hhs is not influenced by unobserved heterogeneity) i.e. $\rho = 0$. If the test shows that ρ is insignificant, then ill_hhs is not endogenous in the regression and the model can be estimated separately. However, if ρ is significant (i.e. $\rho \neq 0$) then endogeneity of the variable (ill_hhs) is confirmed.

Regarding the relevance of the instruments, the weak identification F-statistic is way greater than the usual accepted ($F \ge 10$) for all specifications. This is an indication of strong instruments and the over-identification test result of 0.009 Sergan score indicates that the instruments are valid (see Bound et al., 1995).

Measuring forest dependency as the difference in mean outcomes between all households reporting to have been affected by health related shocks and those not affected by the same shock may give biased estimates of the health related shocks. This bias arises if there are unobserved characteristics that affect the probability of reporting health related shocks in the household. One important source of this bias includes reporting of health related shocks based on the characteristics unobservable to the researcher. The Propensity Score Matching (PSM) estimator used in this analysis helps to control for these sources of selection bias and provides reliable, low – bias estimators of the impact of illness provided sufficient control variables relevant to modelling health related shocks are used (Heckman et al. 1997). Our data set contains a rich data set of conditioning variables to control for households reporting illness or health related shocks.

It has been widely documented that people of higher socio-economic position enjoy better health. This socio-economic gradient in health seems to hold up irrespective of what health measures are used (see e.g. Marmot 1999 and Smith 1998). People with higher education and income report worse health for a given condition (Etile and Milcent 2006).On the other hand, Pattanayak et.al.(2011) reported that individuals from lower socio-economic groups not only suffer from frequent health shocks but also suffer disproportionately for a given health shock. To capture this effect we include control variables which include whether the household head is educated or not, total asset value owned by the household, financial value owned, access to transfer, financial saving, and total expenditure per adult equivalent.

Access to credit can improve households' ability to cope with risks and reduce the risks they face. Accordingly, we include membership in the household extension package to control the effect of access to credit on reporting health shocks. We also include awareness to climate change¹³ and membership in local institutions to indicate the breadth and depth of household social connectedness to identify the role of these connections and access to information in reporting health shocks.

Finally, we include demographic factors (such as age of household head, number of adult male and number of adult female household members) associated with vulnerability to health shocks and our outcome variables. Adult household members, for example, may be correlated with exposure to participate in forest extraction because as a household member reaches adulthood, he/she participates in a greater number of chores, such as fuel wood and water collection, that involve going into forests. In addition, many cultures are preferential to boys so it is possible that girls will suffer from more diseases than boys will (Akin et al. 1985). Different behaviours between boys and girls could also lead to differences in disease rates (Kondrashin and Orlov 1989) and thus in reporting of vulnerability to health related shocks. With this rich set of control variables one can capture many of the determinants of reporting health related stocks that are typically unobservable to the researcher, which helps to reduce potentially significant sources of bias in the propensity score matching estimators.

Table 3.6 presents the empirical results of reporting illness used to create propensity scores for the matching algorithm. Results of the econometric analysis confirm that education level of the household, access to credit and membership in household extension packages and access to transfer (remittances) were associated with a higher probability of reporting illness. This echoes the findings of Etile and Milcent (2006); d'Uva et al.(2008) that people with higher education and

¹³ We believe that households who are aware of climate change may take adaptation measures to avoid health related risks more than those who are not aware that climate is changing in their locality.

income report worse health for a given condition. Similarly, number of adult female household members had a significant and positive effect on reporting illness. As argued before, many cultures are preferential to boys so it is possible that girls suffer more from diseases than boys do (Akin et al., 1985). Interestingly, the number of months that the household had enough food stock is associated with a lower probability of reporting illness. Arguably, households with enough food stock stock have access to better nutrition and are less likely to be prone to disease.

Remoteness from basic infrastructure, measured by the distance to the nearest health post as well as distance to seasonal road, increases the probability of being sick. This is in line with the findings by Klemick et al, (2007) that in rural Tanzania, improvements in roads have a greater impact on health care access than improvements in health facilities because travel costs are one of the major impediments to health care access. Similarly, villages in Indonesia, the Philippines, and SriLanka participating in rural roads projects reported better access to health services based on several indicators compared with non-project villages (Hettige 2006).

<< INSERT TABLE 3. 6 ABOUT HERE>>

3.5.3.2. Average Treatment Effect

Table 3.7 below shows results of the Average Treatment Effects (ATT) estimated by the nearest neighbour and kernel matching techniques for the outcome variables, along with average differences and T-values for treated and control groups, where treatment is defined as a binary variable which equals one if the household has been affected by health related shocks in 2010, 0 otherwise. The results from both the matching algorithms produced consistent estimates of the treatment effects of health shocks. Focusing first on number of forest collection trips, the ATT reported in table 3.7 indicates that vulnerability to health related shocks exerts a positive and

significant effect on the number of trips for forest collection. The difference in ATT for treated and control groups from nearest neighbourhood and the kernel matching results is found to be 18.3 and 22.4 days per year respectively, These results are significant at 5% and 1% level respectively.

<< INSERT TABLE 3.7 ABOUT HERE >>

As shown in panel B of table 3.7 the second outcome estimated for treated and control groups was forest dependency (forest income as a proportion of total household income). This was also found to be positive and statistically significant, again implying that households affected by health shocks are more dependent on forest environmental resources than their counterpart households who were not affected by the same shock. The difference in ATT for treated and control groups is 1.5% and 5.9% for nearest neighbour and kernel matching respectively and both results are significant at the 10 % level of significance. However, despite a positive difference in the ATT for both groups, we found the difference in forest commercialization to be non-significant for non-poor households.

3.6. Conclusions and Policy Recommendations

Low and volatile incomes coupled with the absence of or poor development of financial or risk sharing institutions make consumption smoothing an important issue in low – income countries like Ethiopia (Barrett et al. 2001). One of the most common coping mechanisms in rural areas is forest extraction. Yet the empirical literature on forests as a means of risk coping is relatively limited, in some cases is qualitative/ethnographic (McSweeney, 2004), considerably smaller than that on biodiversity as a source of livelihood (CBD, 2010, Vira and Kontoleon, 2010) and typically relates only extraction volume, income, or income share with total income or asset holdings (e.g., Cavendish, 2000; Coomes et al., 2001; Godoy et al., 1995). As a result, there is little systematic analysis to help guide conservation and development promoters in their efforts to understand how

the poor (and others) deal with negative shocks (McSweeney, 2004). In this study, we analyse the role of forests for coping with idiosyncratic health shocks using sample data from Tigray Region in the northern highlands of Ethiopia. Previous studies used either forest sales or forest collection trips to show the role of forests in buffering misfortune. In this paper, we use a broader approach of outcome variables to analyse the issue. In addition, most of the previous economic studies on the impact of health effects and coping mechanisms treat them as exogenously given. In our study, we employ the Instrumental Variable Approach (IV) techniques and Propensity Score Matching (PSM) and thereby account for endogenous factors at the household level that may be associated with underlying health and labour allocation to forests.

In terms of conclusions, first, on a broader and quite positive note, forests appear to play a significant role in insuring households against idiosyncratic health shocks. This is in line with evidence from a number of different developing countries around the world regarding the role of forests in providing "natural insurance" (Pattanayak and Sills, 2001; McSweeney 2004; Takasaki et al.2004, Fisher et al .2010).

Second, access to institutional support and social capital are significantly and negatively related to forest dependency, forest sales and forest collection trips. In this regard access to credit, transfer, irrigation and ownership of jewellery were found to play a significant role. In addition, households who are aware of the change in their local climate tend to have lower trips to forest collection and lower forest sales. Finally, our results are robust to the econometric method used, i.e. results from the IV and PSM models are consistent.

These findings suggest that in order to achieve a win – win for human and environmental health, policies are needed which ensure that local values for forest resources (such as medicinal plants) are considered by forest management plans (Cofler et al.2006; Holding Anyonge et al.2006). In

addition, policy makers need to examine rural health delivery to determine the environmental consequences of access, or lack thereof, to health services. Also, policy makers need to promote a range of development measures such as improved access to credit through household extension packages and construction of health infrastructures so as to minimize the pressure on natural forests.

Variables	Descriptions and measurements	Mean	S.D
Ill_hhs	1 if the household is affected by health shock, 0 otherwise	0.19	0.39
Hhh_sex	Sex of the household head (1 if male,0 otherwise)	0.85	0.36
Age_hh_1	Age of the household head in years	46.7	12.5
Edu_hhh	Education of the household head in years	1.17	2.18
Family_size	Number of household members per adult equivalent	5.74	2.11
Aware_cc	Awareness of Climate change in the household(1= yes, 0 otherwise)	0.35	0.47
Male_adults	Number of male adult labour in the household	1.37	0.91
Female_adults	Number of female adult labour in the household	1.49	0.87
P_size _tsimdi	Plot size of land owned by the household in Tsimdi	4.36	2.93
Tlu	Number of livestock owned by the household in TLU	3.11	2.59
t_assets	Total value of assets owned by the household	1315	2099
t_hh_income	Total household income in Ethiopian Birr	11859	9054
Fooda_months	Number of months in a year that the household had enough food stock	5.7	3.42
Jewellery	Ownership of Jewellery in the household ($1 = yes, 0$ otherwise)	0.33	0.47
D_health_post	Distance in minutes to the nearest health center	40.1	42.9
D_seasonal_road	Distance in minutes to the nearest seasonal road	9.1	14.6
M_hh_package	Household is a member of household package $(1 = yes, 0 \text{ otherwise})$	0.18	0.37
Access_extension	Access to extension service (=1 yes, 0 otherwise)	0.84	0.36
N_ext_visit	Number of extension visit per year	7.79	9.92
D_transfer	Dummy if the household access to transfer (=1 yes, 0 otherwise)	0.31	0.46
Access_irrigation	1 if the household has access to irrigation, 0 otherwise	0.14	0.34
Log_dis_forests	Log transformed distance to forests in minutes	1.71	0.83
Log_dis_market	Log transformed distance to markets in minutes	1.24	1.33
Southern Zone	1 if the household lives in southern zone, 0 otherwise	0.41	0.49
South_Eastern	1 if the household lives in southern eastern zone, 0 otherwise	0.20	0.40
Eastern Zone	1 if the household lives in Eastern zone, 0 otherwise	0.39	0.48
Average Rainfall in	1 if the household lives in Eastern zone, 0 otherwise	589.4	100.5
mm			

Table 3. 1 Variables and summary statistics of the sample households

Variable definition	Households	Households did	T-value
	reported health	not report health	
	shock $(n=47)$	shock (n= 204)	
Age of Household head in years	48 (1.660)	46(0.894)	0.446
Sex of Household Heads (=1 if male, 0, otherwise)	0.89(0.045)	0.83(0.03)	0.342
Family size of the household	6.1 (0.14)	5.7(0.30)	0.252
Number of Female Adult household Members	1.83(0.153)	1.42(0.056)	0.003***
Number of male Adult household numbers	1.43(0.113)	1.36(0.066)	0.646
Education of the household head in years	1.45(0.32)	1.10(0.15)	0.331
Ownership of land in Tsimdi (equals 0.25 hectare)	3.80(0.40)	4.48(0.21)	0.146
Log total household expenditure per adult equivalent	7.48(0.006)	7.52(0.034)	0.638
Access to irrigation (=1 if yes, 0 otherwise)	0.26(0.064)	0.11(0.022)	0.007***
Number of months that the household had enough food in a year	4.79(0.480)	5.91(0.240)	0.042**
Ownership of Land in TLU	3.24(0.36)	3.08(0.18)	0.694
Forest income in Ethiopian Birr	3427(397.1)	2534(170.2)	0.027**
Participation in Forest Resource Extraction (=1 if yes, 0, otherwise)	0.91(0.04)	0.83(0.02)	0.160
Participation in forest Resource Commercialization (=1 if yes, 0 otherwise)	0.40(0.07)	0.31(0.03)	0.236
Forest Dependency (Share of forest income to overall household income)	0.31(0.035)	0.24(0.014)	0.036**
Average number of trips to forest per annum	92.9(7.50)	70.6(3.52)	0.006***
Access to any transfer (=1 if yes, 0, otherwise)	0.43(0.07)	0.28(0.03)	0.051*
Access to household extension package loans (=1 if yes, 0, otherwise)	0.32(0.07)	0.13(0.02)	0.002***
Participation in the Productive safety net program (=1 if yes, 0, otherwise)	0.55(0.07)	0.70(0.03)	0.061*
Access to extension visit (=1 if yes, 0, otherwise)	0.81(0.06)	0.86(0.02)	0.397
Membership in any organization ((=1 if yes, 0, otherwise)	0.57(0.07)	0.49(0.04)	0.272
Mean Annual Rainfall in MM	558(12.13)	596(7.21)	0.020**
Awareness of climate change (=1 if yes, 0 otherwise)	0.47(0.074)	0.33(0.329)	0.071*
Distance to the nearest health centre in minutes	63.0(7.8)	34.8(2.7)	0.000***
Log distance to local Market	3.98(0.177)	3.39(0.100)	0.010**
Distance to fuel wood collection	7.42(0.66)	10.39(0.47)	0.004***
Distance to Seasonal Road	17.8(3.39)	7.09(0.762)	0.000***
Distance to all weather road	46.4(6.96)	24.0(2.266)	0.001***

 Table 3. 2 Mean Separation Test of households affected by health shock and households that did not

Significant level: *** = 1%; ** = 5% and * = 10%.

Table 3.3 IV Probit estimates of health shocks and	forest sales
----------------------------------------------------	--------------

Variable definition	Coefficient	S.E	T-value
Age of the household head	-0.020	0.009	0.023**
Education of the household head in years	-0.032	0.045	0.478
Gender of the household head (= 1 if Male, 0 otherwise)	0.009	0.350	0.979
Average grade of education in the household	-0.057	0.051	0.260
Number of Male Adult household member	0.511	0.240	0.033**
Number of Female Adult household member	0.248	0.256	0.332
Number of dependent household members	2.050	0.950	0.031**
Land owned by the household in Tsimdi	-0.037	0.051	0.467
Number of livestock owned in TLU	0.004	0.042	0.934
Family size	-0.194	0.119	0.102
Ownership of Jewellery	-0.512	0.208	0.014**
Dummy if the household is receiving any transfer (=1 if yes, 0 otherwise)	-0.284	0.190	0.135
Access to Irrigation (= 1 if yes, 0 otherwise)	-0.583	0.252	0.020**
Membership in the Household Extension package(=1 if yes, 0 otherwise)	-0.682	0.272	0.012**
Awareness of climate change in the household(=1 if yes, 0 otherwise)	-0.062	0.184	0.738
Access to Extension service (=1 if yes, 0 otherwise)	0.379	0.268	0.157
Dummy if the household has own business $(=1 \text{ if yes}, 0 \text{ otherwise})$	-0.670	0.240	0.005***
Shock related to illness of a household member (=1 if yes, 0 otherwise)	2.191	0.522	0.000***
Constant	-0.270	0.658	0.681

* significant at 10%; ** significant at 5%, *** significant at 1 %

Wald test of exogeneity (/athrho = 0): chi2(1) = 5.85 Prob > chi2 = 0.0156

Table 3. 4 IV2sls estimates of health shocks and forest dependency

Variable definition	Coefficient	S.E	T-value
Age of the household head	-0.004	0.001	0.003***
Education of the household head in years	-0.019	0.008	0.017**
Gender of the household head (= 1 if Male, 0 otherwise)	-0.006	0.574	0.918
Average grade of education in the household	0.003	0.007	0.660
Number of Male Adult household member	-0.303	0.343	0.376
Number of Female Adult household member	0.009	0.388	0.822
Number of dependent household members	0.266	0.164	0.017*
Land owned by the household in Tsimdi	0.004	0.007	0.580
Number of livestock owned in TLU	-0.006	0.007	0.410
Family size	-0.023	0.019	0.230
Ownership of Jewellery	-0.065	0.034	0.056*
Dummy if the household is receiving any transfer (=1 if yes, 0 otherwise)	-0.078	0.039	0.049**
Access to Irrigation (= 1 if yes, 0 otherwise)	-0.154	0.048	0.002***
Membership in the Household Extension package(=1 if yes, 0 otherwise)	-0.156	0.063	0.015**
Awareness of climate change in the household(=1 if yes, 0 otherwise)	-0.111	0.032	0.001***
Access to Extension service (=1 if yes, 0 otherwise)	0.139	0.043	0.749
Dummy if the household has own business $(=1 \text{ if yes}, 0 \text{ otherwise})$	-0.158	0.034	0.000***
Shock related to illness of a household member (=1 if yes, 0 otherwise)	0.735	0.250	0.003***
Constant	0.470	0.106	0.000***

* Significant at 10%; ** significant at 5%, *** significant at 1 %

•

Table 3.5	IV 2slsestimates	of health shoc	ks and forest t	rips
-----------	-------------------------	----------------	-----------------	------

Variable definition	Coefficient	S.E	T-value
Age of the household head	-0.509	0.272	0.061*
Education of the household head in years	-6.196	1.637	0.000***
Gender of the household head (= 1 if Male, 0 otherwise)	-1.581	1.671	0.111
Average grade of education in the household	0.435	1.491	0.771
Number of Male Adult household member	-6.800	6.970	0.329
Number of Female Adult household member	1.399	7.883	0.859
Number of dependent household members	1.475	3.233	0.599
Land owned by the household in Tsimdi	0.269	1.485	0.856
Number of livestock owned in TLU	-0.451	1.491	0.762
Family size	2.872	3.907	0.462
Ownership of Jewellery	-1.663	1.908	0.034**
Dummy if the household is receiving any transfer (=1 if yes, 0 otherwise)	- 2.309	8.016	0.773
Access to Irrigation ($= 1$ if yes, 0 otherwise)	0.409	9.953	0.967
Membership in the Household Extension package(=1 if yes, 0 otherwise)	1.904	1.793	0.186
Awareness of climate change in the household(=1 if yes, 0 otherwise)	-1.325	1.601	0.267
Access to Extension service (=1 if yes, 0 otherwise)	1.853	0.836	0.265
Dummy if the household has own business $(=1 \text{ if yes}, 0 \text{ otherwise})$	3.910	2.998	0.576
Shock related to illness of a household member (=1 if yes, 0 otherwise)	1.149	0.744	0.063*
Constant	3.126	1.634	0.001***

* Significant at 10%; ** significant at 5%, *** significant at 1 %
Table 3. 6 Probit estimation of reporting illness

Variable definition	Coefficient	S.E	T-value
Age of the household head	0.01	0.01	0.263
Education of the household head in years	0.11	0.05	0.049**
Number of Male Adult household member	0.01	0.13	0.932
Number of Female Adult household member	0.25	0.13	0.041**
Total value of Durable assets owned by the household	-0.00	0.00	0.233
Financial Value of assets owned by the household	3.87	0.00	0.999
Log total expenditure per adult equivalent	0.27	0.28	0.334
Membership in social organization ((=1 if yes, 0 otherwise)	0.23	0.22	0.306
Access to Extension service (=1 if yes, 0 otherwise)	0.15	0.32	0.640
Membership in the Household Extension package(=1 if yes, 0 otherwise)	1.01	0.31	0.001***
Awareness of climate change in the household(=1 if yes, 0 otherwise)	0.30	0.24	0.219
Dummy if the household is receiving any transfer (=1 if yes, 0 otherwise)	0.52	0.24	0.029**
Number of months that the household had enough food in 2010	-0.08	0.04	0.064*
Log distance to Forests in minutes	0.21	0.15	0.165
Distance to the nearest health service in minutes	0.01	0.00	0.077*
Distance to all weather road in minutes	0.00	0.00	0.558
Distance to seasonal road in minutes	0.02	0.00	0.077*
Constant	-4.25	2.17	0.117

Table 3. 7 Differences in ATT for treated and control groups

Outcome	Matching	E(Y)	E(Y)	Differences in	P-Value	
	Algorism	H=1	H=0	Average outcome		
	-			(ATT)		
PANEL A: Nu	mber of trips to	forests per y	year			
Treatment : Dur	nmy =1 if the h	ousehold wa	as affected by hea	lth related shocks, 0 ot	herwise	
Impact : Me	an Impact					
Forest_trips	N-neighbor	92.9	74.6	18.3	0.010**	
	K-matching	94.7	72.3	22.4	0.005***	
PANEL B: Forest Dependency (Ratio of forest income to overall household income)						
Treatment : Du	mmy = 1 if the 1	household w	as affected by hea	alth related shocks, 0 o	otherwise	
Impact : Me	an Impact					
Forest_dep	N-neighbor	31.0	29.5	1.5	0.086*	
	K-matching	33.8	27.9	5.9	0.059*	
PANEL C: Nor	n-Poor Househo	olds				
Treatment : Du	mmy =1 if the	household w	as affected by hea	alth related shocks, 0 o	otherwise	
Impact : Me	an Impact					
Forest_sales	N-neighbor	61.7	40.4	21.3	0.242	
	K-matching	56.8	39.0	17.8	0.299	

* Significant at 10%; ** significant at 5%, *** significant at 1 %

Note: H = 1 and H = 0 refer to households affected by health shocks and those that were not,

respectively.

FOREST – FOOD SECURITY LINKAGES IN NORTH HIGHLANDS OF ETHIOPIA: SAFETY NETS OR POVERTY TRAPS?

4.1. Introduction

The importance of NTFP (non-timber forest products) captured the imagination of conservationists around the world, when an article by Peters et al (1989) published in Nature claimed that more money could be earned from tropical forests by collecting these products than from logging (Kaimowitz, 2004). The perception that NTFPs are more accessible to rural populations, especially to the rural poor (Saxena 2003) and that their exploitation is more benign than timber harvesting (Myers 1988) favored NTFP collection becoming an ecologically acceptable economic option for development. There was also an assumption, often implicit, that making forests more valuable to local users can encourage forest conservation (Plotkin and Famolare, 1992). As a result NTFP-based development was born as a new development paradigm capable of accommodating many potentially conflicting needs: of local livelihoods and of global markets; of balancing regional developmental aspirations with that of national growth; and above all that of reconciling environmental and development goals.

These developments initiated an increased level of research investigations and publications in the last two decades on different aspects of NTFP and in the process supported the global NTFP discussion leading to policy changes in many countries. However, the NTFP statistics in the form of macro and micro-economic indicators, portrayed in these discourses with varied degrees of claims on pro-poor benefits, also translate to very poor economic returns to the poorest. In fact

refocusing of the development agenda on poverty has led to recent reassessment of the role that bio-diversity plays in livelihoods and poverty alleviation. A profusion of new commentary has emerged. This poses many fresh questions, and, to some extent, tempers previous optimism regarding the ability of this sector to make a difference by providing a more subtle and complex picture of livelihood – biodiversity linkages. For example, Neumann and Hirch(2000) present evidence from an array of studies that show NTFP extraction is an activity of the poor, and Wunder (2001) and Angesen and Wunder(2003) argue that optimism about a win – win for development and conservation is unwarranted, and that NTFP extraction is generally a low- income activity that may even result in a poverty trap. The general conclusion is that the safety net and poverty trap aspects of NTFPs are linked, in as much as features that make non-timber forest products attractive to the poor also limit their potential for generating higher income and escaping their poverty (Angelson et al, 2007). However, many of the existing arguments and conclusions made on the link between poverty and biodiversity do not support their claims with strong evidence. Full understanding of the links between biodiversity and poverty require the ability to make causal inferences about the counterfactual, and few of the previous studies do this¹⁴. As a result, they have limitations in shedding light on the nature and the extent of linkages between biodiversity and poverty and hence miss opportunities for identifying common causes and common solutions to the two issues (Roe, 2005).

In a wide ranging review on the research done on the link between poverty and biodiversity, (CBD, 2010) reported that if nature's resources help to temporarily smooth consumption and incomes, their impacts may be better captured through an explicit focus on this temporality as part of our

¹⁴Delacote (2007) analyzes the role of the commons as insurance and how it might be linked to a poverty trap. However, his work is applied theoretic, which needs empirical testing.

poverty measure, instead of annualized income or consumption (in which these temporary contributions do not always feature as significant) (CBD, 2010). This relative importance of the temporality contribution of forests and the relative dearth of attention given to it by researchers and policy makers alike raise the possibility of significant untapped potential of forests to rural livelihoods, and provides an opportunity to increase our understanding about the specific roles of forests in different contexts and a significant impact for high quality research that tackles this topic. The main objective of this research is therefore to investigate the role of forest environmental resources in household food security and its implications on the long-term livelihood of farm households living along the margins of protected forests in the northern high lands of Ethiopia.

Our study contributes to the existing literature on biodiversity and poverty in two ways: first unlike the previous works done we are explicitly focusing on the temporality contribution of forests along with annualized consumption; second, we employ the Propensity Score Matching (PSM) method to consider the causal relationship between participation in forest environmental resource extraction and food security indicators, to address the counterfactual questions that may be significant in informing policy makers on the impact of forest environmental resources on the welfare of rural households in developing countries such as Ethiopia.

The chapter is structured as follows: Section 4.2 provides a review of literature on the role of forests as safety nets. Section 4.3 provides an overview of food security and forest environmental resources in Ethiopia. Section 4.4 provides a description of data. Section 4.5 discusses thetheoretical framework and empirical procedure of the study. Section 4.6 presents the empirical results, while section 4.7 explains households' choice of participation in forest environmental resources. The final section provides conclusions and policy implications.

4.2. Review of Related Literature on the safety net role of Forests

Several studies have examined the extent to which poor people depend upon biodiversity as means of subsistence or income to the poor and as insurance to the poor from risks and shocks(CBD, 2010). A substantial amount of the existing literature examines the dependency of the poor on biodiversity reflected in the proportion of income that is derived from forests (Vira and Kontoleon 2010), on income equity effects (e.g. Babulo et al., 2009) and on correlations rather than causal effects of income from forest environmental resources on household welfare (Pattanayak and Sills 2001). A few studies assess the role of tropical forests (and hence biodiversity rich ecosystems) as an insurance against food security (and income) variability (CBD, 2010; Vira and Kontoleon 2010).

Forest resources, such as non-timber forest products (NTFPs), have long served as safety nets or "natural insurance" to help cope with environmental and economic shocks (Angelsen & Wunder, 2003; Shively, 1997, Fisher et al., 2010), and are an important part of livelihood strategies in sub-Saharan Africa (Cavendish, 2000; Paumgarten, 2005; Shackleton &Shackleton, 2004; Mamo et al., 2007 Babulo et al., 2009). Rural households, which have limited credit and insurance options, choose a diversification of their activities in order to reduce aggregate risk (Delacote, 2007). For example, Fisher et al. (2010) in their study of southern Malawi found that forests are providing a source of cash for coping with weather related crop failure. They also found that households most reliant on forests have low income per person, are located close to forests, and are headed by individuals who are older, more risk averse, and less educated than their counterparts who are less forest-dependent. Similarly, Volker and Waibel (2010) found that households in Vietnam who are affected by idiosyncratic health shocks, experienced by economically active household members, and by severe weather shocks were more likely to extract forest products. Debela et al. (2012)

argued that shocks tend to lead rural households in Uganda to depend on forests to meet subsistence and cash needs. In Kenya and Tanzania, six out of 16 strategies for dealing with drought involved the use of indigenous plant species found in forests (Eriksen et al., 2005). Forest gathering was also an important strategy for coping with covariate flood shocks in Peru, particularly in households with few physical assets and more adult members (Takasaki et al., 2004). Similar findings are reported in Pattanayk and Sills (2001); McSweeney (2003); Fisher and Shively (2005).Hence, at least for the case of poor communities living close to tropical forests, there appears to be support for conjecture that forests act as a safety net against food insecurity (and income) variability (CBD, 2010; Vira and Kontoleon, 2010).

However, even if the above papers studied the insurance role of common property resources, many other researchers argued that resource reliance may represent a strategy that prevents the poor from participating fruitfully in other activities and escaping their poverty (Angelsen et al., 2007). Indeed existing overviews (Byron and Arnold 1999; Neumann and Hirsch 2000) suggest that most NTFPs produce low returns (particularly in terms of return per ha and often, also per labour unit), are primarily used for subsistence and often fill income gaps. They are accessible to the poor people precisely because no one else wants them. In economic terms, many are inferior goods which are substituted by superior products when incomes rise (Arnold and Perez 2001), and/or become intensively cultivated, in which case poorer people may be less well-placed to benefit (Homma 1996, Ruiz-Perez et al. 2003).

Delacote (2007) analyses the use of the commons as insurance, exploring whether the commons are a safety net or poverty trap. However, his work is applied theoretic, and was not tested empirically. It is therefore interesting to analyse the forest-food security linkage and test whether

it is a safety net or a poverty trap in the context of rural households who are dependent on forest environmental resources as part of their livelihood strategy.

4.3. Food Security and Common Pool Resources use in Ethiopia

Chronic food insecurity has been a defining feature of the poverty that has affected millions of Ethiopians for decades. The vast majority of these very poor households live in rural areas that are heavily reliant on rain fed agriculture and thus, in years of poor rainfall, the threat of widespread starvation is high (Hoddinott et al., 2009). Moreover, many smallholder farmers in the region need to supplement the income derived from rain fed agriculture. Income from forest products contributes as high as 27% and 39% for rural households living close to natural forests in Northern and Central Ethiopia, respectively (Babulo et al., 2009; Mamo et al., 2007). Studies have also shown that forest dependent people are poorer, live close to the forest, are younger, and have lower livestock and crop income (Illukpitiya and Yanagida, 2008; McElwee, 2008; Mamo et al., 2007). These factors often interact with one another resulting in a reinforcing cycle of the "poverty, food insecurity and natural resources degradation trap". This problem is most acute in the Ethiopian highlands (where over 85 percent of the country's population lives), which is affected by recurrent drought and famine affecting millions of people (FAO 2003).

To mitigate this poverty and resource degradation, closing degraded land areas from human and livestock intervention to promote natural regeneration of plants, commonly termed as enclosures, is among the major efforts to constrain resource use practiced in the highland areas of Tigray, northern Ethiopia (Tewelde-Berhan, 2002; Babulo, 2007). However, the adoption of restrictive use rules and the limitation in allowable harvests have led many local residents to view enclosures as a less favoured land use option (Babulo, 2007). There is also consensus in the relevant literature that benefit streams from enclosures to date are largely environmental and economic benefits for

most people are so far minimal (Howard and Smith, 2006). These arguments are partly because of a lack of clear understanding about the contribution of forest environmental resources to the food security of rural households in the region.

4.4. Data Description

A household survey was conducted in 2010/11 on 254 households in Tigray region of northern Ethiopia. A two stage sampling design was made in the study. The primary sampling units (PSUs) were tabias. Sample tabias were selected on the basis of secondary information collected from all the Woredas. In this category, a total of ten tabias namely Arato, Derga _ajen, Hugumrda, Meswaeti, Kara_adishawo, WorebayuKal_amin, Kelisha_emni,andFelege_woini were selected for the survey (see Figure 2.1). The tabias selected are representative of the three different agro-ecological zones of the region identified on the basis of altitude. Areas with altitude ranging from 1500- 2300 m.a.s.l. are locally termed as *woinadouga* i.e. midland areas, areas above 2300 m.a.s.l. are locally known as *douga* i.e. Highland areas and areas with altitude less than 1500 m.a.s.l. are termed as *kola* i.e., lowland. A multi-purpose questionnaire was used to gather information on household income, expenditure, household assets, vulnerability to shocks, labor allocation and local institutions alongside a host of other information related to production and sales.

The purpose of this part of the analysis is to examine the relationship between forest dependency and food security. The dependent variable used in the study is a dummy variable that takes the value of one, if the household participated in forest environmental resource extraction, and a value zero, if no participation was recorded. The outcome variables used in this study are log transformed total per capita expenditure per adult equivalent (Log_tot_exp_aue), the number of months in a year that the household had enough food stock available (Fooda_months), and one binary indicator that takes one, if the household mortgaged its assets (distress_sale) for current consumption during the farming season, zero otherwise. These food security indicators were chosen to be in line with those suggested in the literature (Owusu and Abdulai, 2011).

<< INSERT TABLES 4.1 AND 4.2 ABOUT HERE >>

Table4.1 presents the definitions and sample statistics of the variables used in the analysis. Table 4.2 presents differences in means of the variables used in the matching analyses along with their significance levels. The significance levels suggest that there are some differences between participants and non-participants with respect to household and outcome variables. With regards to the outcome variables, there appear to be statistically significant differences in household total per capita expenditure per adult equivalent, distress sales and in the number of months in a year that the household had enough food stock available between participants and non-participants. There are also significant differences in sex and education of household head, household size, awareness of climate change, net financial assets, total assets, off farm income and total farm income.

There is a significant difference between participants and non-participants regarding awareness of climate change. 31% of the participants have awareness of climate change compared to 58% of non-participants; this suggests awareness of climate change might reduce participation in extractive activities. There are also significant differences between participants and non-participants in terms of education: on average, participants have a lower number of years of formal education than non-participants. With more education, households may access a broad variety of livelihood activities, have higher opportunity cost of time (Adhikari et al., 2004) and may disregard the collection of forest resources. In addition, participants have lower value of total assets, net financial assets and durable assets compared to their non-participant counterparts. As argued by Angelsen and Wunder (2003) households with more assets may easily access more profitable livelihood activities and

may give less priority to environmental resource incomes, which are often considered "employment of the last resort".

Many researchers argued that female-headed households may exhibit greater dependence on forest resources (Cavendish, 2000; Shackleton and Shackleton, 2006a, b), possibly because they often are poorer, have less access to adult labor (Vedeld et al., 2004), and may lack the means to seek employment away from their families (Shackleton and Shackleton, 2004). In contrast to this view, our survey indicated that male-headed households participate more in forest environmental resource collection; this finding is similar to Adhikari (2005). Furthermore, there is a significant difference between participants and non-participants in terms of family size:participants have higher family size than the corresponding figure of the non-participant counterparts. Larger households may collect forest resources for two reasons: they have more labor to allocate to this activity, but may also be forced to do so by greater dependence and consumer burdens (Godoy et al., 1997).

The findings from the previous section that simply compare mean differences in the outcome variables and other household variables between participants and non-participants suggest that non-participants are generally better off than participants. Given that the comparisons of mean differences do not account for the effect of other characteristics of farm households, they may confound the impact of participation in forest environmental resource extraction on food security status with the influence of other characteristics. Multivariate approaches that account for selection bias arising from the fact that participants and non-participants may be systematically different are essential in providing sound estimates of the impact of participation in forest extraction on household income and food security. The independent variables used in the probit regression models to predict the propensity scores were based on past research on the determinants of

82

participation in forest environmental resource extraction (Cavendish, 2000; Shackleton and Shackleton, 2006a,b, Vedeld et al., 2004 Adhikari et al., 2004 Adhikari, 2005, Babullo et al, 2009).

4.5. Theoretical Framework and Empirical Procedure

4.5.1. Theoretical issues

The conditions of the study area, where households live in relatively remote areas on forest margins and where household time endowments are their main factor input, are well matched to the basic assumptions of household production theory. The fundamental issue in household production theory is the non-separability of production and consumption decisions, and arises when markets are missing or incomplete.

Sills et al (2003) discuss some of the theoretical considerations in applying the household production model to households living on forest margins. The following briefly describes the main components of the theoretical model developed by Sills et al: their approach essentially provides the theoretical underpinning for the analysis conducted in this chapter. Their model assumes that a household engages in agriculture (*A*) and collection of NTFPs(*F*). It also assumes complete markets for agricultural products and for market goods(*M*), but incomplete markets for NTFPs and labour. The amount of labour and leisure available are constrained by the value of agricultural output plus any exogenous income(*I*) such as remittances. The household seeks to maximize a single utility function, which depends on consumption of agricultural goods(A_H), market goods(M_H), forest goods(F_H), and home time T_H , (including leisure, childcare, etc.). Household utility is conditioned on preferences (Φ).

Max $U(A_H, M_H, F_H, T_H; \Phi)$

s.t.

 $(1)T \geq T_H + T_A + T_F$

$$(2)A = a(T_A, F_A, M_A; \Psi)$$

$$(3)F = f(T_f; B, H, \Psi)$$

$$(4.1)$$

$$(4)F \ge F_H + F_A$$

$$(5)P_{A}(A - A_{H}) + I \ge P_{M}(M_{H} + M_{A})$$

The constraints apply to (1) household income, (2) agricultural production, (3) non-timber forest production, (4) forest output allocation, and (5) budget. The choice variables are T_H ,

 T_F , M_A , M_H , F_H , F_A , and A_H .

Agricultural production is a function (a) of household time allocated to agriculture (T_A) and other inputs collected from the forest (F_A) or purchased in the market (M_A) , conditioned on fixed household production endowments (e.g., land, livestock) and technology (Ψ). Forest production (f) is also conditioned on fixed production endowments. However, the model also assumes that it does not compete with agriculture for land, but rather takes place in public forest, conditioned by its biophysical state (B) and household knowledge of the forest (H). The only variable input in forest product collection is household time (T_F). Forest products are either consumed (F_H) or used as inputs to agriculture (F_A).

Sills et al (2003) go on to derive first order conditions for utility maximisation. These include (i) that marginal cost equals marginal benefit in forest collection; (ii) the marginal utility of increased agricultural production arising from inputs of forest products must equal the marginal utility of household consumption of forest products. Taken together, these conditions imply that the shadow value of time must also equal the marginal utility of increased agricultural production due to forest inputs obtained with more time spent collecting.

Generally Sills et al (2003) demonstrate that the shadow values in their model – of time, income, increased agricultural production, and the "price" of NTFPs governing collection and consumption decisions - which are internal to each household, depend on all exogenous variables in the system. "*This dependence of production decisions on preference and endowments is termed nonseparability in the household production literature and results whenever key markets are missing or incomplete*"(Sills et al., 2003, pg 264).

As noted by Sills et al, if markets are incomplete, household behavior can be modeled "*as a function of socioeconomic and environmental characteristics reflecting preferences, technology, and input endowments, rather than prices.*"(Sills et al, pg 267). No specific restrictions on the functional form of models are warranted under these conditions. Exogenous variables used in empirical estimation generally represent household demographics, assets, income opportunities/sources, possible substitutes, and Regional/locational characteristics.

4.5.2. Empirical estimation issues and procedures

The above theoretical considerations provide a guide for the empirical estimation procedures used in this chapter. In the analysis in this chapter, which is focused on factors influencing the decision to participate in forest environmental resource extraction, and the relationship between participation and food security status, the independent variables used in the analysis (Table 4.1) comprise demographic variables, asset variables, access to other employment, location dummies, a knowledge variable (awareness of climate change) and variables reflecting experience of shocks. The decision to participate (Pi*) can be specified as an index function as follows:

$$P_i^* = \beta Z_i' + \mu_i$$

$$P_i = 1 \text{ if } P_i^* > 0$$

$$P_i = 0 \text{ if } P_i^* \le 0$$

$$(4.2)$$

Where Z'_i denotes a vector of variables such as household and location characteristics that influence decision to participate, and μ_i is the random disturbance term.

To analyse the relationship between participation in forest environmental resource extraction and outcomes such as household income and food security status, following Osuwu et al (2011) we can specify a linear function

$$Y_i = \lambda_i + \alpha_i X'_i + \delta_i P_i + \varepsilon_i \tag{4.3}$$

Where Y_i is household income or food security status, P denotes a dummy variable representing one if the household participates in forest related activity and zero otherwise; X'_i is a vector of household characteristics, λ_i is a vector of unknown parameters, and ε_i is a random error term. Participation in forest related activity is treated as an exogenous variable on the basis that households participate in forest related activity to increase their income or improve their food security status. However, participation in forest environmental resource extraction may not be random, with the group of participants and non-participants being systematically different: if this is the case the measured return to participation in forest environmental resource collection may be biased. Therefore, it is necessary to ensure that there is a credible counterfactual against which we may measure the impact of participation in forest environmental resource extraction on food security. Following Owusu et al (2011) we can resort in these circumstances to using the propensity score matching approach to overcome potential problems of selection bias. The steps involved in this approach are further discussed below.

Let P =1, 0 indicate treatment (participation in forest environmental resource extraction) and control (non-participation) respectively and let Y_1 and Y_0 denote the outcome of interest (food security) for households with treatment and without treatment respectively. Since we observe households to be either with treatment or without treatment, we cannot observe the causal effect of interest: $Y_1 - Y_0$. Some features of this distribution are estimable, nevertheless. In particular, we may consider the Average Treatment Effect of the Treated (ATT):

$$ATT = E(Y_1 - Y_0 | T = 1)$$
(4.4)

The ATT measures how much the outcome of interest changes on average for those households who undergo the treatment (in this case, those who participated in forest environmental resource extraction). Clearly, simply computing the difference in the average outcomes of those households who are participating in forest environmental resources and those who did not is open to bias, as there are observed and unobserved characteristics that determine whether the household undergoes the treatment. That is,

$$E(Y_1 | P = 1) - E(Y_0 | P = 0) =$$

$$E(Y_1 | P = 1) - E(Y_0 | P = 1) + (Y_0 | P = 1) - E(Y_0 | P = 0) =$$

$$E(Y_1 - Y_0 | P = 1) + E(Y_0 | P = 1) - E(Y_0 | P = 0) =$$

$$\frac{E(Y_1 - Y_0 \mid P = 1)}{ATT} + \frac{(Y_0 \mid P = 1) - E(Y_0 \mid P = 0)}{BIAS}$$
(4.5)

Only if we can guarantee that outcomes of the control group are equal on average to what the outcomes of the treatment group would have been in the absence of treatment does this consistently estimate the ATT. With non-random sorting into treatment and control, this condition is rarely met.

Now suppose that by conditioning on an appropriate set of observables, X, the non-participation outcome Y_0 is independent of the participation status P. This is the weak version of the unconfoundness assumption, also called ignorable treatment assignment (Rosenbaum and Rubin, 1983) or conditional assumption (Lechner, 2000) or selection on the observables, which suffices when the parameter of interest is the ATT, as only assumptions about the potential outcomes of comparable individuals are needed to estimate counterfactuals.

$$Y_0 \perp P \mid X \tag{4.6}$$

This implies that

$$E(Y_0 | P = 1, X) - E(Y_0 | P = 0, X) = 0$$
(4.7)

In order to identify the ATT, the overlap or common –support condition is also assumed. It ensures that, for each treated household, there are control households with the same X.

$$Pr(P=1 \mid X) < 1 \tag{4.8}$$

Therefore, under the assumptions stated in equation (4.6) and (4.8) above, we could estimate the ATT from the differences in outcomes between treated and controls within each cell defined by

the conditioning variables (Blundell and Costa Dias, 2002). Using the law of iterated expectations and the conditional independence assumption, the ATT can be retrieved from observed data in the following way:

ATT =
$$E(Y_1 | P = 1) - E(Y_0 | P = 1)$$

= $E_X [(E(Y_1 | X, P = 1) - E(Y_0 | X, P = 1)) | P = 1]$
= $E_X [(E(Y_1 | X, P = 1) - E(Y_0 | X, P = 0)) | P = 1]$ (4.9)

The estimate of ATT as shown in equation (4.9) turns out to be prohibitive in terms of data when the set of conditioning variables X is large. An alternative is to use the results of Rosenbaum and Rubin (1983, 1984) and condition on the probability of treatment as a function of X, the propensity score P(X), since the conditional independence assumption also implies that

$$E(Y_0 | P = 1, P(X)) - E(Y_0 | P = 0, P(X)) = 0$$
(4.10)

Therefore, we could estimate ATT from the differences in outcomes between treated and controls within each cell defined by values of P(X).

ATT =
$$E(Y_1 | P = 1) - E(Y_0 | P = 1)$$

= $E_{P(X)} \left[\left(E(Y_1 | P(X), P = 1) - E(Y_0 | P(X), P = 1) \right) | P = 1 \right]$
= $E_{P(X)} \left[\left(E(Y_1 | P(X), P = 1) - E(Y_0 | P(X), P = 0) \right) | P = 1 \right]$ (4.11)

Provided that the conditional participation probability can be estimated using a parametric method such as a probit model, matching on the univariate propensity score reduces the dimensionality problem.

To estimate the propensity score we estimate a probit model with the binary dependent variable whether a member of household participated in forest environmental resource extraction (=1) or not (=0) - using our sample data. Using this approach, we are able to control for unobservable factors that may influence households' reported participation in forest environmental resource extraction. A number of matching algorithms are suggested in econometric literature: in this study we consider nearest neighbour and kernel matching methods.

4.6. Empirical Results and Discussions

4.6.1. Propensity Score Results

To construct the propensity score used to match participants and non-participants in forest environmental resource extraction, a probit regression model was estimated, where the dependent variable is a binary variable indicating whether the household participates in forest environmental resource collection or not, using a broad set of control variables. In propensity score matching, it is important to condition the match on variables that are highly associated with the outcome variables (Heckman and Navarro-Lozano, 2004). However, Smith and Todd (2005) argued that there is little guidance on how to select the set of conditioning variables used to construct the propensity score. As stated in the empirical strategy section, we focused on finding a set of conditioning variables that on theoretical grounds and information in the survey should be highly associated with the probability of participating in forest environmental resource extraction and with the outcomes of interest. In general the control variables were chosen based on empirical studies of rural households' reliance on forest environmental resources (Angelsen and Wunder,2003; Vedeld et al., 2004; Adhikari et al., 2004; Fisher and Shively, 2005; Cavendish, 2000 ; Babulo et al., 2009). Table 4.3 presents the empirical results of participation in forest environmental resource extraction used to create propensity scores for the matching algorithm. The control variables chosen include household demographic variables (sex, age and education of the household head and the number of students in the household, number of male and female adult labor), whether the household is aware of climate change or not, whether the household experienced weather related shocks, death or illness of a household member, loss of livestock, whether the household earns off-farm income, whether the household has access to private forest, land holding in tsimdi, number of livestock owned in TLU and number of donkeys owned, log distance to market and to forest in minutes, and location dummies.

<<INSERT TABLE 4.3 ABOUT HERE>>

Results of the econometric analysis in table 4.3 confirm that education of household head is a significant determinant of participation in forest environmental resources extraction. Higher education levels of a household head are associated with lower probability of participating in forest environmental resource extraction. With more education, households may access a broad variety of livelihood activities, have higher opportunity cost of time (Adhikari et al., 2004) and may disregard the collection of forest resources. Second, our result indicated that the probability of participating in forest environmental extraction is significantly higher for male-headed households compared to female-headed households. This result is different from many other studies that have concluded that forest extraction isparticularly an activity of female headed households (Cavendish, 2000; Shackleton and Shackleton, 2006a, b), possibly because they often are poorer, have less access to adult labour (Vedeld et al., 2004), and may lack the means to seek employment away from their families (Shackleton and Shackleton, 2004). Our finding is similar to that of Adhikari (2005). Likewise, households engaged in off farm employment are less likely to participate in

forest extraction, which usually has lower returns on labor effort and typically requires quite strenuous physical activity (Angelsen and Wunder, 2003). It is interesting to note the negative and significant impact of awareness of climate change: farm households that are aware of climate change are less likely to participate in forest extraction.

The density distribution of the propensity score for households that are participating in forest extraction and those that did not shows a good overlap (see figure 4.1) suggesting that the common support condition is satisfied. The bottom half of the graph shows the propensity score distribution for the untreated, while the upper –half refers to treated households.

<< INSERT FIGURE 4.1 ABOUT HERE>>

Table 4.4 demonstrates how matching restricts the control sample in order to increase the similarity of the subsample of control cases that are directly compared with treated cases, in order to estimate the consequences of treatment. Table 4.4 presents the balancing information for the propensity scores and for each covariate before and after matching. We used the standardized bias differences between treatment and control samples as a convenient way to quantify the bias between treatment and control samples. In many cases, we found that sample differences in the unmatched data significantly exceed those in the samples of matched cases. The process of matching thus creates a high degree of covariate balance between the treatment and control groups that are used in the estimation procedure.

The imbalances between the treatment and control samples in terms of the propensity score had been more than 100 % before matching as shown in table 4.4. This bias was significantly reduced to a level of 4.3% after matching. The same table also shows that before matching, several variables exhibit statistically significance differences, while after matching the covariates are balanced.

The low pseudo R^2 and the insignificance likelihood ratio tests also support the hypothesis that both groups have the same distribution in covariates after matching (see table 4.5). These results clearly show that the matching procedure is able to balance the characteristics in the treated and the matched comparison groups. We therefore used these results to evaluate the impact of forest extraction on food security among groups of households having similar observed characteristics. This allows us to compare outcomes for households participating in forest related activities with those of a comparison group showing common support.

4.6.2. Average Treatment Effects

Having looked at the determinants of the probability of participating in forest environment resource extraction in rural Tigray, we now examine the size of consumption expenditure per adult equivalent, distress sale and food availability during the critical periods between participants and non-participant households. We then split the sample into two sub samples: the poor (households belonging to the first and second quintiles) and the non – poor (households belonging to the two upper quintiles), and we do the same comparisons. This split enables us to see the extent to which participation in forest environmental resource is pro- poor or otherwise.

Table 4.6 presents estimates of the average impact of participation in forest environmental resource extraction. The outcomes considered include total expenditure per adult equivalent, distress sale of assets for immediate consumption and the number of months in a year that the household had enough food available. Panel A of table 4.6 shows the average level of food security outcomes defined above along with the average differences and P-values for treated and control variables, where treatment is defined by participation in forest environmental resource extraction for the whole sample, while panels B (table 4.7) and C (table 4.8) present the same outcome for poor and non-poor households respectively.

The results indicate that the difference in household expenditure per adult equivalent between treated (participants in forest environmental resource extraction) and untreated (non-participants) households for the whole sample is substantial when households are matched based on demographic, assets and other covariates, using nearest neighbor and kernel matching respectively. Participants in the matched sample have on average 0.11 and 0.18 lower per capita log expenditure per adult equivalent than non-participants in the same sample as shown by the nearest neighbor and kernel matching respectively. This suggests that participation in forest environmental resource extraction has a causal (negative) influence on total expenditure per adult equivalent when households are matched on relevant socio-demographic, assets and other covariates.

<< INSERT TABLE 4.6 ABOUT HERE>>

For the number of months that the household had enough stocks of food available (fooda_months), a similar trend is reflected. Treated households in the matched sample have fewer months of having enough food available during the year than non-participants in the same sample. The participants have enough food stock available on average for 1.74 months (24%) lower than non-participants and the difference is statistically significant at 5% levels of significance.

The third outcome estimated for participants and non-participants is distress sale of assets for immediate consumption (distress sale). Again, participants have higher levels of distress sale than non-participants in the same sample do. Participants have a distress sale level that is on average 0.05 points (i.e 13%) higher than non-participants and the difference is statistically significant at 1% level of significance.

To sum up, Panel A (table 4.6) indicated that treated households (those participating in forest resource extraction) have significantly lower food security outcomes than the untreated households in rural Tigray, and the finding is robust to whether we use different food security indicators or to the matching methods used.

In Panel B (table 4.7) and C (table 4.8), we split the sample into two sub samples: the poor (households belonging to the first and second quintiles) and the non–poor (households belonging to the two upper quintiles), and we do the same comparisons as panel A. This split enables us to see if the average impacts in panel A (table 4.6) mask significant impacts of participation in forest environmental resource extraction for some participants and to see the extent to which participation in forest environmental resource extraction is pro- poor or otherwise.

<< INSERT 4.7 ABOUT HERE>>

The results show that both poor and non-poor participants (treated) have lower total expenditure per adult equivalent than poor and non-poor untreated households (non-participants).Conversely, we find that the poor and non-poor households that are participating in forest extraction had lower distress sale of assets for immediate consumption than non-participants in the same sample,supporting the widely held view that forests are important as an economic buffer in hard times. Finally while we find that the non-poor participants had higher number of months of food stock available compared to non-participants in the same sample,there was no significant different in the number of months thatpoor households had enough food available,between participants and non-participants. These results need to be treated with some caution due to the relatively small number of non-participants in the sample, but they do suggest the importance of conducting such disaggregated analysis to identify differences in the functions of forest resource utilization for different categories of households.

<< INSERT 4.8 ABOUT HERE >>

4.7. Does Diversification or Coping Strategy explain households' choice of Participation?

It is now generally established that the safety net use of NTFP extraction may take two forms, corresponding to two kinds of risk-management strategies: first, the diversification strategy, because the households use NTFP extraction as a risk-free asset (Alderman and Paxson, 1994); second, the coping strategy which consists of extracting NTFP only when agricultural output is too low, working as a "natural" insurance mechanism. Therefore, the problem for local communities has both the characteristics of portfolio analysis and economics of insurance (Delacote 2007). In the latter case, while NTFP extraction insures against a bad outcome (in case a household relied on just one risky activity) as discussed in section 4.2, it also potentially excludes the household from engaging in high return activities that come from specialization.

We therefore further explore the idea that participation in forest environmental resource collection is motivated by mitigation or reduction of the potentially high risk in agriculture. We measure the risks that households face in agriculture through the predicted probability that the farm households have zero or negative profits from their agricultural activities¹⁵. We compare this outcome across comparable groups of participant and non –participant households. Next, we matched participants and non-participants on the propensity score. Having obtained groups of comparable households, we calculate the average differences in the predicted probabilities of having zero and/or negative profit from farming across the participants and non-participants. These predicted probabilities are

¹⁵Farming profits are calculated as net income from farming. Net income is obtained even for households which do not sell on the market by applying to their farm production the average prices faced by the households in the same area. Moreover, in doing this part of analysis we exclude households who are landless although they are very few.

obtained through a probit model where having negative or zero profit from farming is a function of sets of variables which normally influence farming profits (see table 4.9).

Our results in Panel A of table 4.10 suggest that households in rural Tigray are not driven into NTFP extraction by risks in farm output. This finding is consistent with the diversification strategy usually observed in Latin American countries (Alderman and Paxson, 1994) that involves households raising ex ante the number of their activities, choosing possible activities that have low covariance.

However, the average pair-wise differences for poor and non-poor (panel B and C) of table 4.10 revealed that participants and non-participants are significantly different in their predicted probabilities of negative profit from farming. For example, results from panel B indicate that the difference in poor households' farming profit between treated (participants) and control (nonparticipants) is substantial (-0.05) and (-0.15) for the nearest neighbour and kernel matching respectively, i.e. participants in the matched sample from poor households have on average higher negative profit from farming than non-participants. Moreover, while participants from the subsample of non-poor households in panel C have on average 0.15 negative profits from farming, the non-participant households have on average zero negative farming profits. In this regard, our analysis suggests that the poor and non –poor farm households in Tigray are also driven into forest environmental resource extraction by risks in farm output or being more prone to volatile agricultural income. This finding is also consistent with the coping strategy which is observed in African countries (Alderman and Paxson, 1994) and which suggests that agricultural risk mitigation partially explains participation in forest resource extraction (Delacote, 2007, Pattanayak and Sills, 2001). Overall, it appears that the decision to participate in forest environmental resource extraction in rural Tigray is motivated by both the diversification and coping strategies.

4.8. Conclusions and Policy Implications

In this chapter, we investigate the differences in food security outcomes (total expenditure per adult equivalent, distress sale of assets for immediate consumption and number of months that the household had enough food stock available in a year) between households that participate in forest environmental resource extraction and those who do not, using 251 sample farm households from ten Peasant associations (tabias) living in Tigray Regional State Ethiopia. We use a propensity score matching method to account for selection bias that normally occurs when unobservable factors influence both participation in forest environmental resources extraction and food security outcomes. The paper also addresses household heterogeneity by explicitly providing separate estimates of the average treatment effect (ATT) for poor and non-poor households.

Our analysis leads to several interesting results. First we find that education, age and sex of household head, access to off–farm employment, shocks related to death or illness of a household member and awareness of climate change significantly affects participation in forest environmental resources. Second, we find that participation in forest environmental resource extraction in rural Tigray has significantly lower food security outcomes than farm households that are not participating. Third, we find that the poor and non-poor households that are participating in forest extraction had lower distress sale of assets for immediate consumption than non-participants in the same sample, enforcing the widely held view that forests are important as an economic buffer in hard times. Fourth, households in rural Tigray are not driven into forest extraction by risks in farm output only but also by a diversification strategy, suggesting that the problem for local communities has both the characteristics of portfolio analysis and economics of insurance (Delacote 2007).

The implications of the above findings are that forests serve as a safety net by mitigating distress sale of assets for immediate consumption; at the same time households that are participating in forest resources are relatively poor in the sense that they have lower food security outcomes, supporting the widely held view that the poor appear to be linked with nature-based resource use, but that these may serve to perpetuate poverty and food insecurity (CBD, 2010). Thus, examining and promoting alternative livelihood strategies such as off-farm work and other social protection programs such as the productive safety net program in Ethiopia are warranted to solve the problems of food insecurity and resource degradation in the region.

Table 4. 1 Variables and summary statistics of the sample households

Variables	Descriptions and measurements	Mea	S.D
		n	
Forest_extraction	Dummy =1 if the household participates in forest extraction, 0 otherwise	0.85	0.36
Outcome Variables			
Log_tot_exp_aue	Log total expenditure of the household per adult equivalent units	7.51	0.48
Distress_sale_a	Food insecurity (Dummy =1 if the household reported distress sale of assets for immediate	0.71	0.46
	consumption, 0 otherwise		
Fooda_months	Food Security (number of months in the year that the household had enough food stock)	5.70	3.43
Hhh sex	Sex of the household head (1 if male .0 otherwise)	0.85	0.36
Age hh 1	Age of the household head in years	46.8	12.5
Edu hhh	Education of the household head in years	1.17	2.18
Student number	Number of students in the household	1.51	1.43
Aware cc	Awareness of Climate change in the household $(1 = yes, 0 \text{ otherwise})$	0.35	0.48
Male_adults	Number of male adult labour in the household	1.37	0.91
Female_adults	Number of female adult labour in the household	1.49	0.87
P_size _tsimdi	Plot size of land owned by the household in Tsimdi	4.36	2.93
Tlu	Number of livestock owned by the household in TLU	3.11	2.59
N-donkey	Number of donkey owned by the household	0.75	0.84
Private_forest	Access to private forest (1 if male,0 otherwise)	0.45	0.50
Off_farm	Access to off farm employment by the household (1 if male,0 otherwise)	0.32	0.47
Log-dis_forests	Log distance to forests in minutes	1.24	1.33
Log_dis_market	Log distance to woreda market in minutes	1.71	0.82
S_wheather	Household experienced to any weather related shock ($1 = yes$, 0 otherwise)	0.80	0.40
S-death_ill	Household experienced death or illness of a member ($1 = yes$, 0 otherwise)	0.23	0.42
S_livestock_loss	Household experienced shock due to loss of livestock (1 = yes, 0 otherwise)	0.45	0.50
PANEL G: Location v	zariables		
Southern Zone	Household lives in southern zone(1= yes, 0 otherwise)	0.41	0.49
South_Eastern	Household lives in southern eastern zone(1= yes, 0 otherwise)	0.20	0.40
Eastern Zone	Household lives in Eastern zone(1= yes, 0 otherwise)	0.39	0.49

Table 4. 2 Mean separation to	ests of participants and	non-participants in forest	t environmental resource extraction
-------------------------------	--------------------------	----------------------------	-------------------------------------

	Non-Particinant	Particinant (n–	
Variable name	(n=38)	213)	p-value*
	Mean (SE)	Mean (SE)	_ •
Log per capita expenditure per adult equivalent	7.84 (0.09)	7.45(0.03)	0.0000***
Distress Sale	0.45(0.82)	0.76(0.03)	0.0001***
Number of months with enough stock of food	6.77(0.56)	5.51(0.23)	0.0378**
Sex of household head	0.68(0.76)	0.88(0.22)	0.0020***
Ag e of household head	47.3(2.26)	46.7(0.84)	0.7584
Education of household head	2.42(0.48)	0.944(0.13)	0.0001***
Family Size	4.60(0.33)	5.95(0.14)	0.0003***
Awareness of Climate Change	0.58(0.08)	0.31(0.03)	0.0016**
Number of livestock in TLU	2.65(0.47)	3.29(0.16)	0.2341
Number of adult household member	2.53(0.20)	2.92(0.08)	0.0784*
Land holding in tsimdi	3.76(0.38)	4.46(0.21)	
Non-farm income	3352(1647)	733.92(137)	0.0007***
Food consumption expenditure per adult equivalent	7.55(0.08)	7.20(0.03)	0.0000***
Access to irrigation	0.21(0.07)	0.12(0.02)	0.1433
Off farm employment	0.55(0.08)	0.28(0.03)	0.0009***
Frequency of shocks (2005-2010)	1.82(0.21)	2.18(0.08)	0.0770*
Total Asset Value	2182(645)	1160(103)	0.0055***
Durable Asset Value	1882(617)	808.5(97)	0.0021***
Net financial asset	87.14(97.3)	-18.2(10.3)	0.0000**

Variable definition	Coefficient	S.E	P-value
Sex of the household head (=1 if male, 0, otherwise)	1.02	0.36	0.004***
Age of the household head in years	-0.03	0.01	0.006***
Education of the household head in years	-0.20	0.06	0.000***
Number of students in the household	0.12	0.98	0.230
Awareness of climate change in the household(=1 if yes, 0, otherwise)	-0.77	0.26	0.003***
Number of male adult household member	-0.04	0.14	0.808
Number of female adult household member	0.21	0.17	0.233
Land Holding in Tsimdi (= 0.25 hectare)	0.03	0.05	0.633
Number of Livestock owned in TLU	-0.02	0.07	0.752
Number of Donkey owned by the household	0.02	0.18	0.896
Access to private forest (=1 if yes, 0, otherwise)	0.12	0.07	0.613
Death or illness of a household member (=1 if yes, 0, otherwise)	0.56	0.33	0.091*
Weather related shocks (=1 if yes, 0, otherwise)	-0.11	0.40	0.790
Loss of livestock shocks (=1 if yes, 0, otherwise)	0.39	0.26	0.133
Access to off farm employment (=1 if yes, 0, otherwise)	-0.55	0.29	0.055*
Log transformed distance to forest	0.02	0.16	0.923
Log transformed distance to market	-0.04	0.10	0.671
South-Eastern (=1 if the household lives in South-Eastern zone, 0, otherwise)	-0.28	0.38	0.459
Eastern Zone (=1 if the household lives in Eastern zone, 0, otherwise)	0.04	0.31	0.890
Constant	1.83	0.78	0.020**
Number of Observations	251		
Pseudo R2	0.2685		
Log likelihood	-78.057317		

Table 4. 3 Propensity score of participation in forest environmental resource extraction

Significant level: *** = 1%; ** = 5% and * = 10%.



Figure 4. 1 Distribution of propensity scores

Table 4. 4 Propensity score and covariate balances

		Ν	Iean	% redu	uction	t- te	est
Variable	Sample	Treated	Control	% bias	bias	t	p > t
Propensity Score	Unmatched	0.88783	0.62147	135.5		9.75	0.000
	Matched	0.88783	0.87932	4.3	96.8	0.67	0.502
Hhh_sex	Unmatched	0.8774	0.6842	47.5		3.10	0.002
	Matched	0.8774	0.9067	-7.2	84.8	-0.97	0.332
Age_hh_1	Unmatched	46.604	47.342	-5.6		-0.33	0.739
	Matched	46.604	46.722	-0.9	84.0	-0.09	0.925
Edu_hhh	Unmatched	0.9481	2.4211	-59.1		-3.93	0.000
	Matched	0.9481	1.2910	-13.8	76.7	-1.61	0.109
Student_number	Unmatched	1.5708	1.1842	26.0		1.53	0.127
	Matched	1.5708	1.0229	36.9	-41.7	4.37	0.349
Dep_ratio	Unmatched	0.4882	0.4294	28.0		1.66	0.097
	Matched	0.4882	0.4739	6.8	75.7	0.78	0.436
CC_aware	Unmatched	0.3160	0.5789	-54.4		-3.17	0.002
	Matched	0.3160	0.4710	-32.0	41.1	-3.30	0.024
Male_adult	Unmatched	1.4057	1.1842	23.6		1.38	0.168
	Matched	1.4057	1.5585	-16.3	31.0	-1.69	0.091
Female_adult	Unmatched	1.5236	1.3421	22.1		1.19	0.237
	Matched	1.5236	1.4235	12.2	44.9	1.26	0.209
Land_size	Unmatched	4.4723	3.7625	26.2		1.38	0.170
	Matched	4.4723	4.8177	-12.8	51.4	-1.41	0.160
Livestock_TLU	Unmatched	3.1945	2.6468	20.1		1.20	0.231
	Matched	3.1945	3.2366	-1.5	92.3	-0.16	0.874
N_Donkey	Unmatched	0.7925	0.5526	28.4		1.63	0.104
	Matched	0.7925	0.8099	-2.1	92.7	-0.21	0.831
Private_forest	Unmatched	0.4670	0.3421	25.5		1.43	0.155
	Matched	0.4670	0.5074	-8.2	67.6	-0.83	0.406
Transfer	Unmatched	0.3113	0.2895	4.7		0.27	0.789
	Matched	0.3113	0.1684	31.3	-560.8	3.55	0.132

(Note: Figures in bold are significant variables)

Death_ill	Unmatched	0.2359	0.1316	27.0		1.43	0.154
	Matched	0.2359	0.2250	2.8	89.6	0.26	0.792
Weather_shock	Unmatched	0.8113	0.7105	23.6		1.42	0.157
	Matched	0.8113	0.7399	16.7	29.1	1.79	0.178
Livestock_loss	Unmatched	0.4856	0.2632	46.9		2.56	0.011
	Matched	0.4856	0.4621	5.0	89.3	0.49	0.625
Off_farm	Unmatched	0.2830	0.5526	-56.4		-3.33	0.001
	Matched	0.2830	0.1733	22.9	59.3	2.71	0.015
Distance_forest	Unmatched	1.7119	1.7303	-2.3		-0.13	0.899
	Matched	1.7119	1.4009	38.4	-1586.0	4.17	0.677
Distance _market	Unmatched	1.3029	0.9392	24.8		1.55	0.122
	Matched	1.3029	1.6204	-21.7	12.7	-2.59	0.110

Table 4. 5 Other covariate balance indicators before and after matching

Indicator	Sample	
Pseudo R2	Unmatched	0.27
	Matched	0.14
LR X2 (p-value)	Unmatched	58.46(0.001)***
	Matched	29.02 (0.341)

Outcome	Matching	E(Y)	E(Y)	Differences in	P-Value
	Algorism	Participants	Non-	Average outcome	
			participants	(ATT)	
PANEL A: All H	Iouseholds				
Treatment : Part	icipation in Fo	rest resource ex	traction		
Impact : Mea	an Impact				
Log_t_exp_aeu	N-neighbor	7.45	7.56	-0.11	0.000***
	K-matching	7.45	7.63	-0.18	0.000***
Distress_sale	N-neighbor	0.75	0.70	0.05	0.000***
	K-matching	0.75	0.67	0.09	0.000***
Fooda_months	N-neighbor	5.51	6.37	-0.85	0.046**
	K-matching	5.51	7.18	-1.67	0.045**
_					

Table 4. 6 Differences in food security outcomes for participants and non-participants for the whole sample

* significant at 10%; ** significant at 5%, *** significant at 1 %

Table 4. 7 Differences in food security outcomes of poor households

Outcome	Matching	E(Y)	E(Y)	Differences in	P-Value
	Algorism	Participants	Non-	Average outcome	
			participants	(ATT)	
PANEL B: Poor	Households				
Treatment : Parti	cipation in For	rest resource ex	traction		
Impact : Mea	in Impact				
Log_t_exp_aeu	N-neighbor	7.27	7.48	-0.21	0.002***
	K-matching	7.27	7.47	-0.20	0.002***
Distress_sale	N-neighbor	0.67	0.84	-0.17	0.039**
	K-matching	0.66	0.81	-0.15	0.037**
Fooda_months	N-neighbor	5.15	6.73	-1.58	0.410
	K-matching	5.16	6.90	-1.74	0.281

Table 4. 8 Differences in food security	y outcomes of non-poor households
-----------------------------------------	-----------------------------------

Outcome	Matching	E(Y)	E(Y)	Differences in	P-Value
	Algorism	Participants	Non-	Average outcome	
			participants	(ATT)	
PANEL C: Non-Poor Households					
Treatment : Participation in Forest resource extraction					
Impact : Mean Impact					
Log_t_exp_aeu	N-neighbor	7.63	7.99	-0.36	0.001***
	K-matching	7.63	7.95	-0.32	0.001***
Distress_sale	N-neighbor	0.84	0.93	-0.83	0.002***
	K-matching	0.84	0.91	-0.07	0.001***
Fooda_months	N-neighbor	5.86	5.15	0.71	0.036**
	K-matching	5.86	4.94	0.92	0.046**
--					

Variable definition	Coefficient	S.E	P-value
Dependent Variable : Dummy equals 1 if farming profit is negative, 0 otherwise			
Explanatory Variables			
Age of the household head in years	0.004	0.013	0.744
Education of the household head in years	0.072	0.073	0.325
Number of Adult Males in the household	-0.040	0.182	0.828
Participation in other business activities ((=1 if yes, 0, otherwise)	-0.102	0.354	0.772
Value of seed in Ethiopian Birr	0.001	0.000	0.007***
Land Holding in Tsimdi (= 0.25 hectare)	-0.081	0.083	0.327
Number of Livestock owned in TLU	-0.517	0.462	0.001***
Value of Fertilizer in Ethiopian Birr	0.001	0.001	0.327
Number of crops grown	-0.462	0.175	0.008***
Death or illness of a household member (=1 if yes, 0, otherwise)	-0.081	0.405	0.842
Loss of livestock shocks (=1 if yes, 0, otherwise)	0.755	0.326	0.020**
Access to credit ((=1 if yes, 0, otherwise)	-0.314	0.341	0.358
Number of Agricultural Extension Visit	-0.002	0.027	0.913
Log transformed distance to market	0.134	0.142	0.345
Distance to DA office in Munities	0.011	0.005	0.021**
South-Eastern (=1 if the household lives in South-Eastern zone, 0, otherwise)	0.798	0.397	0.044**
Eastern Zone (=1 if the household lives in Eastern zone, 0, otherwise)	0.420	0.241	0.890
Constant	-1.253	0.831	0.132
Number of Observations			
Pseudo R2	0.3230		
Log likelihood	-52.047128		

* significant at 10%; ** significant at 5%, *** significant at 1 %

Outcome	Matching	E(Y)	E(Y)	Differences	P-Value
	Algorism	Participants	non	in Average	
			participants	outcome	
				(ATT)	
PANEL A: All Househo	olds				
Treatment : Dummy =1i	f the household p	articipates in for	rest extraction,	zero otherwise	•
Impact : Mean Impac	t				
Farming profit (=1 if	N-neighbour	0.15	0.06	-0.09	0.408
zero or negative)	8				
C /	K-matching	0.14	0.08	-0.06	0.425
PANEL B: Poor Househ	olds				
Treatment : Dummy =1i	f the household p	articipates in for	rest extraction,	zero otherwise	•
Impact · Mean Impac	٠t				
Farming profit (=1 if	N-neighbour	0.14	0.09	-0.05	0.071*
zero or negative)	iv neighbour	0.11	0.09	0.05	0.071
Leto of negative)	K-matching	0.15	0.00	-0.15	0.009***
PANEL C: Non-Poor He	ouseholds				
Treatment : Dummy =1i	f the household p	articipates in for	rest extraction,	zero otherwise	•
Internet March Internet		Ĩ			
Impact : Mean Impac	N a si shh sua	0.15	0.00	0.15	0 000***
Farming profit (=1 11	IN-neighbour	0.15	0.00	-0.15	0.008
zero or negative)	V motohing	0.15	0.00	0.15	0 000***
Ψ ' 'C' / 100/ ΨΨ '	K-matching	0.15	0.00	-0.13	0.009
* significant at 10%; ** si	ginncant at 5%,	significant a	at 1 %		

Table 4. 10 Differences in farming profit between participants and non-participants

WELFARE IMPACTS OF FOREST ENVIRONMENTAL RESOURCE COMMERCIALIZATION: A MICRO- SIMULATION APPROACH FOR TIGRAY, ETHIOPIA

5.1. Introduction

Poverty in the developing world is concentrated in rural areas, and rural households, particularly in developing countries, depend on the environment for at least some of their income (Cavendish, 2000; Angelsen et al., 2001; World Bank, 2002). In this regard, commercialization of Non-Timber Forest Products (NTFPs) have been given a prominent role in many conservation and development projects based on the proposition that supporting the production and trade of NTFPs leads to livelihood improvements without compromising the environment. The attention given to the commercial extraction of NTFPs as a conservation strategy comes from three implicit assumptions: (1) commercial NTFP production can provide economically attractive options to farmers helping to increase their income and offering development opportunity (Peters et al., 1989; Clay & Clement, 1993); (2) NTFP production is a more benign way to use tropical forests than most land use alternatives, allowing for the conservation of key forest values (Myers, 1988); and (3) increased monetary value of the NTFPs will prevent people from converting the lands into other land uses (Evans, 1993). Accordingly, expectations have been raised that this sub-sector will play a significant role in reducing rural poverty, addressing international concerns for poverty reduction, conserving biodiversity and fulfilling the MDG goals (World Bank, 2002).

While the "picture" of the significance of natural product trade for livelihoods exists, the situation in reality is complex and variable, and limited empirical data from across a range of regions, vegetation types, and socio-economic contexts are available to assess the ability of these products to create lasting opportunities for local livelihood enhancement (FAO, 2003; Sunderlin et al., 2005). In fact, refocusing of the development agenda on poverty has led to recent reassessment of the role that biodiversity plays in livelihoods and poverty alleviation. This poses many fresh questions, and, to some extent, tempers previous optimism regarding the ability of this sector to make a difference by providing a more subtle and complex picture of livelihood -biodiversity linkages (e.g. Arnold, 2002a; Belcher, 2005; Koziell, 2001; Lawrence, 2003; Ros-Tonen and Wiersum, 2005; Scherr et al., 2004; Wunder, 2001). Central to these new enquiries is a more perceptive and nuanced appreciation of (a) what is by poverty alleviation in relation to natural resources, distinguishing between poverty prevention or mitigation and poverty elimination; (b) the links between natural resource dependence and the potential of natural product trade to provide pathways out of poverty; and (c) the extent to which opportunities associated with natural product production and sale can be made more pro-poor and thus contribute to the efforts to combat poverty and vulnerability (FAO, 2003; Koziell, 2001). A key area of debate is whether the trade in natural products can assist in improving livelihoods and income, or alternatively, whether it offers limited options serving only as a last resort, possibly contributing to persistent poverty (Belcher, 2005; Ros-Tonen and Wiersum, 2005; Wunder, 2001).

Yet most literature on natural product commercialization focuses on international value chains for charismatic products with high levels of external intervention (Taylor, 1999). In addition, few case comparisons to date (e,g, Belcher et al., 2005; Marshall et al., 2006) have sought to systematically investigate the livelihood and poverty outcomes from the commercialization of a range of products

produced and marketed within the same geographic setting. Moreover, most of the referred studies examine total forest or environmental income which includes critical subsistence products such as medicinal plants and wild foods (see e.g. Jodha, 1986; Reddy & Chakravarty, 1999; Belcher et al., 2005; Babulo et al., 2009). In addition, although there are a number of commercial non-timber forest products in Ethiopia¹⁶, there is no systematic study to our knowledge that investigates whether commercialization of forest products enhances the welfare of the rural poor.

Moreover, although several studies have explored the role of forest income in reducing poverty and inequality and established empirical regularities about the positive impact of forests on the welfare of rural households (e.g. Jodha 1986; Reddy & Chakravarty, 1999; Belcher et al., 2005; Babulo et al., 2009; Lopez-Feldman, 2011), they have some methodological shortcomings. One such limitation is that none of the studies examine the question of whether participants and nonparticipants have inherent different income potentials, all things being equal. Their method provides a direct and simple measure of how forest income contributes to the total income under the simplifying assumption that the differences in income between households who are participating in forest environmental resource commercialization and those households that did not participate in forest commercialization could be due to observed heterogeneity. They therefore shed little light on the important policy issue of who might gain or lose from potential changes in forest policy.

To provide more accurate and policy-relevant results it is important to estimate what the households' income situation would have been in the absence of participation –the counterfactual income level- and then to compare that with the actual household income situation with forest

¹⁶The most commercial non-timber forest products in Tigray include, honey, bee wax, gums and resins, cactus, bamboo and charcoal.

income included as part of total income. Previous studies on the contribution of forests in Ethiopia and elsewhere have neglected this issue, choosing instead to compare actual income excluding forest income with income including forest income rather than estimating the counterfactual without-participation scenario. This could result in overestimation of the contribution of forests to household income.

Indeed, not distinguishing between the causal effect of forest commercialization and the effect of unobserved heterogeneity could lead to misleading policy implications (Di-falco and Veroseni, 2010;Carter and Milon, 2005). To this end, we used a simultaneous equation model with endogenous switching by full information maximum likelihood estimation (Di-falco and Veroseni, 2010;Di-Falco et al. 2011, Asfaw and Shiferaw, 2010;Carter and Milon, 2005) to account for the endogeneity of the participation in forest environmental resource commercialization. For the model to be identified, we use selection instruments such as distance to forest, distance to market and awareness of climate change. We have access to a particularly rich data set, which contains both households that did and did not participate in forest resource commercialization plus a very large set of control variables.

Our results are interesting in a number of respects. First, they show that the choice of methodology can have a significant impact on the estimated impacts and conclusions made about the impact of participation in forest environmental resource commercialization on the welfare of rural households. Second, the estimated heterogeneity revealed that, independent of participation, participant households would have less income on average than non- participants, implying that the group of farm households that are actually participating in forest environmental resource commercialization has systematically different characteristics than the group that did not participate. Third, the treatment effect analysis revealed that participation decreased income for

non- participant groups in the counterfactual case that they did participate, while it improved the income of households that actually participated. This suggests that participation in the forest environmental resource commercialization seems to be particularly important for the most vulnerable group of farm households by helping them to fill the income gap with the less vulnerable households in the same locality.

The following section presents an overview of forest conservation and poverty alleviation in rural Tigray, Ethiopia. Section 5.3 presents a brief description of the study area and descriptive statistics. Section 5.4 introduces the econometric model and estimation procedure. We describe a "treatment effects" approach (Heckman 2001; Di-Falco et al 2011) to evaluate the contribution of forest environmental resources commercialization on the household income per adult equivalent. Section 5.5 describes the estimated econometric results followed by discussion. We conclude with a summary and observations on the contribution of forest environmental resource commercialization to the household income and its implications for development policy in Ethiopia and similar developing countries.

5.2. Overview of forest Conservation and poverty Alleviation in Rural Tigray

Environmental degradation in Tigray has been well documented in the literature see for example (Chisholm 2000 and 2004; Gebremedhin et al., 2000; Howard and Smith, 2006). Forest resources of the region were overexploited and today forests and woodlots cover less than 2 percent of the region's total area (TFAP, 1996 cited in Babulo, 2007). Efforts to contain the problem of land degradation have been made at several levels (Nedessa et al., 2005).Closing degraded land areas from human and livestock intervention to promote natural regeneration of plants, commonly termed as enclosures, is among the major constraint efforts practiced in the highland areas of Tigray, northern Ethiopia (Tewelde-Berhan, 2002; Babulo, 2007). In addition to enclosures, the

regional government created restrictions on the use of certain tree species and the restrictions apply to all common land areas, regardless of whether they are open access or enclosed (FAO, 2006). This strategy is meant to benefit the environment and is a conservation effort that the community itself is operating (Gebrehiwot and van der Veen, 2010). The immediate positive environmental effects of the enclosures and the species restrictions are nearly universally acknowledged both by researchers and by local populations (Howard and Smith, 2006) and the international community alike.

However, the adoption of restrictive use rules and the limitation in allowable harvests have led many local residents to view enclosures as a less favoured land use option (Babulo, 2007). There is also consensus in the relevant literature that benefit streams from enclosures to date are largely environmental and economic benefits for most people are so far minimal (FAO, 2006). However, the arguments from both sides have been based essentially on qualitative narratives of the pros and cons of enclosure establishment and have not been backed up by quantitative reasoning and indicators (Babulo, 2007). This paper is therefore an attempt to quantitatively measure the welfare impacts of forest environmental resources commercialization for rural households living in Tigray and to provide some policy recommendations that can promote a win-win for conservation of forest environmental resources and poverty reduction in the region.

5.3. Data Description and Study Sites

A household survey was conducted in 2010/11 on 254 households in Tigray region of northern Ethiopia. However, because of missing data on some key variables for 3 households, our final estimating sample consists of 251. A two stage sampling design was made in the study. The primary sampling units (PSUs) were tabias. Sample tabias were selected on the basis of secondary information collected from all the Woredas. In selecting the sample tabias, a number of factors

that affect socio economic condition such as nearness to market, geographical location, the availability of both rain fed agriculture and irrigation, size of tabias based on population etc were considered so as to make the sampled tabias representative. In this category, a total of ten tabias namely Arato, Derga _ajen, Hugumrda, Meswaeti, Kara_adishawo, WorebayuKal_amin, Kelisha_emni,andFelege_woini were selected for the survey. The tabias selected are representative of the three different agro-ecological zones of the region identified on the basis of altitude. Areas with altitude ranging from 1500- 2300 m.a.s.l. are locally termed as *woinadouga*i.e midland areas, areas above 2300 m.a.s.l. are locally known as *douga*i.e highland areas and areas with altitude less than 1500 m.a.s.l. are termed as *kola* i.e., lowland. A multi-purpose questionnaire was used to gather information on household income, expenditure, off-farm income, household assets and local institutions alongside a host of other information related to production and sales.

<< INSERT FIGURE 2.1 ABOUT HERE >>

The dependent variable used in this study is a dummy variable that takes the value of one, if the household participated in forest environmental resources commercialization, and a value of zero, if no participation was recorded. The outcome variable used is the log transformed total per capita income in adult equivalent units. Tables 5.1, 5.2 and table 5.1A in the appendix present the sample statistics of the variables used in the analysis, the mean separation test for participants and non-participants and their definitions respectively.

<<INSERT TABLE5.1, 5.2 AND TABLE 5.1A ABOUT HERE>>

Out of the 251 households in the data set, about 33% of the households were participants during the year 2009/10. The average age of sample households is 46.76 years and 85% are male headed. There is also a significant difference observed in the age and gender of participants and non-

participants. In addition participants seem to have significantly lower number of students and lower average grade of educational level than their counterpart non-participant households. The average land holding in tsimdi (=0.25 hectare) for participants is 3.75 while it is about 4.7 for non-participants, the difference is statistically significant. Moreover, households that are participating in forest commercialization are living far away from markets and are also prone to shocks such as weather and illness of a household member.

5.4. Econometric Model and Estimation Procedure

Households who are involved in forest environmental resource commercialization are likely to differ from households who do not. A household may participate in forest resource commercialization if he/she anticipates that the decision to participate in forest commercialization is worthwhile rather than entering by random assignment. Unobservable characteristics of households and their family may affect the participation decision in forest commercialization and the level of income, resulting in inconsistent estimates of the effect of participation income (Difalco et al, 2010). Accordingly, we used a simultaneous equations model with endogenous switching by full information Maximum likelihood (FIML) to determine the counterfactual income of forest resource commercialization following Di-falco et al. (2010) as shown below.

We specify the selection equation for participating in forest environmental resource commercialization as

(5.1)
$$A_i^* = Z_i \alpha + \eta_i$$
 with $A_i = \begin{cases} 1 & \text{if } A_i^* > 0 \\ 0 & \text{otherwise} \end{cases}$

That is, households will choose to participate in forest extraction ($A_i = 1$) if $A^* > 0$, 0 otherwise, where A^* represents the expected benefits of participating in forest extraction with respect to not participating, Z is a vector of observed household characteristics or variables that are influencing the decision to participate in forest commercialization or not. To complete the system of simultaneous equations model for household income, conditional on participation in forest environmental resource commercialization, we specify two equations following Di-falco et al., (2010), where (1) participation of households in forest commercialization and (2) non-participation in forest commercialization are defined as follows:

(5.2a) Regime 1:
$$y_{1i} = X_{1i}\beta_1 + \varepsilon_{1i}$$
 if $A_i = 1$

(5.2b) Regime 2:
$$y_{2i} = X_{2i}\beta_1 + \varepsilon_{2i}$$
 if $A_i = 0$

Where y_i is the income and food security level in regimes 1 and 2, X_i represents a vector of explanatory variables mentioned above.

Finally, the error terms are assumed to have a trivariate normal distribution, with zero mean and covariance matrix Σ , i.e., $(\varepsilon_1, \varepsilon_2, \eta_1) \blacksquare N(0, \Sigma)$ with

$$oldsymbol{\Sigma} = egin{bmatrix} \sigma_n^2 & \cdot & \cdot \ \sigma_{1\eta} & \sigma_1^2 & \cdot \ \sigma_{2\eta} & \cdot & \sigma_2^2 \end{bmatrix},$$

Where σ_n^2 is the variance of the error term in the selection equation (5.1), (which can be assumed to be equal 1 since the coefficients are estimable only up to a scale factor), σ_1^2 and σ_2^2 are the variances of the error terms in the income function (5.2a) and (5.2b), and $\sigma_{1\eta}$ and $\sigma_{2\eta}$ represent the covariance of η_i , ε_{1i} and ε_{2i} . Since y_{1i} and y_{2i} are not observed simultaneously the covariance between ε_{1i} and ε_{2i} is not defined (Maddala, 1983 cited inDi-falco et al., 2010)). An important implication of the error structure is that because the error term of the selection equation (5.1) η_i is correlated with the error terms of the income equation (5.2a) and (5.2b) (ε_{1i} and ε_{2i}), the expected values of ε_{1i} and ε_{2i} conditional on the sample selection are non zero:

$$E[\varepsilon_{1i}|A_i=1] = \sigma_{1\eta} \frac{\Phi(Z_i\alpha)}{\Phi(Z_i\alpha)} = \sigma_{1\eta}\lambda_{1i}, \text{ and } E[\varepsilon_{2i}|A_i=0] = -\sigma_{2\eta} \frac{\Phi(Z_i\alpha)}{1-\Phi(Z_i\alpha)} = \sigma_{2\eta}\lambda_{2i},$$

Where $\phi(.)$ and $\phi(.)$ are the standard normal probability function, $\Phi(.)$ the standard normal cumulative density function respectively, and $\lambda_{1i} = \frac{\phi(Z_i\alpha)}{\Phi(Z_i\alpha)}$, $\lambda_{2i} = \frac{\phi(Z_i\alpha)}{1-\Phi(Z_i\alpha)}$. Thus, if the estimates of the associated coefficients to the inverse Mill's ratios are found to be statistically different from zero, the hypothesis of the absence of sample selectivity bias can be rejected (Féres et al., 2007). This model is defined as a "switching regression model with endogenous switching" (Maddala and Nelson, 1975 cited in Di-falco et al., 2010). The full information maximum likelihood (FIML) is an efficient method to fit the endogenous switching regression model (Di-falco and Veroseni, 2010; Linh et al., 2014).

The logarithmic function given the previous assumption regarding the distribution of the error term as presented by Di-falco et al., (2010) is

(5.3)
$$\ln L_{i} = \sum_{i=1}^{N} A_{i} \left[\ln \phi \left(\frac{\varepsilon_{1i}}{\sigma_{1}} \right) - \ln \Phi + \ln \Phi(\theta_{1i}) \right] + (1 - A_{i}) \left[\ln \Phi \left(\frac{\varepsilon_{2i}}{\sigma_{2}} \right) - \ln \sigma_{2} + \ln (1 - \Phi(\theta_{2i})) \right],$$

Where $\theta_{ji} = \frac{(z_i \alpha + \rho_j \varepsilon_{ji} / \sigma_j)}{\sqrt{1 - \rho_j^2}}$, j = 1, 2, with ρ_j denoting the correlation coefficient between the error

term η_i of the selection equation (1) and the error term of ε_{ji} of equation (5.2a) and (5.2b), respectively.

5.4.1. Expected outcome and Treatment Effects

The expected outcomes with and without participation in forest environmental resource commercialization can be used to estimate the expected treatment effects in the two regimes. Table

5.3 shows the expected income and treatment effects of the household who have participated in forest environmental resource commercialization and those who did not. For example, (a) and (b) in table 5.3 show the actual or observed income in the sample , while (c) and (d) show the expected income without participation in forest environmental resource commercialization and this represents the counterfactual income. The conditional expectations for income status of the households are presented following Di-falco et al., (2010) below:

(5.4a) $E(y_{1i}|A_i = 1) = X_{1i}\beta_1 + \sigma_{1\eta}\lambda_{1i}$ (5.4b) $E(y_{2i}|A_i = 0) = X_{2i}\beta_2 + \sigma_{2\eta}\lambda_{2i}$ (5.4c) $E(y_{1i}|A_i = 1) = X_{2i}\beta_2 + \sigma_{2\eta}\lambda_{1i}$ (5.4d) $E(y_{1i}|A_i = 0) = X_{1i}\beta_1 + \sigma_{1\eta}\lambda_{2i}$

<< TABLE 5.3 ABOUT HERE >>

In addition, following Heckman et al., (2001) and Di-Falco et al., (2011) we calculate the heterogeneity and treatment effects as shown in equations 5.5 – 5.8; thus we seek to determine whether the effect of participation in forest resources commercialization is smaller or larger for participants who actually participated and those who did not participate relative to their counterfactual case. The effect of participation in forest environmental resource commercialization for those households who actually participated is the "effect of treatment on the treated" (TT) and the effect of participate, but could participate defined as the "effect of the treatment on the untreated"(TU) is calculated as follows (Di-Falco et al., 2011; Di-falco and Veroseni;2010;Carter and Milon, 2005):

(5.5)
$$TT = E(y_{1i}|A_i = 1) - E(y_{1i}|A_i = 0) = X_{1i}(\beta_1 - \beta_2) + (\sigma_{1\eta} - \sigma_{2\eta})\lambda_{1i},$$

(5.6)
$$TU = E(y_{2i}|A_i = 0) - E(y_{2i}|A_i = 1) = X_{2i}(\beta_1 - \beta_2) + (\sigma_{1\eta} - \sigma_{2\eta})\lambda_{2i},$$

The policy relevant treatment effects can also be distinguished from heterogeneity effects (Carter and Milon (2005). For example, those households who participated in forest environmental resource commercialization may have higher (lower) income regardless of their participation in forest environmental resource commercialization due to other unobservable factors that affect household income. This "base heterogeneity effect" (Carter and Milon, 2005; Di-falco and Veroseni, 2010;Di-Falco et al., 2011) for households that are actually participated in forest environmental resource commercialization and that did not is given in equations (5.7) and (5.8) respectively

$$(5.7)BH_{1} = E(y_{1i}|A_{i} = 1) - E(y_{1i}|A_{i} = 0) = (X_{1i} - X_{2i})\beta_{1i} + \sigma_{1\eta}(\lambda_{1i} - \lambda_{2i}).$$

$$(5.8)BH_{2} = E(y_{2i}|A_{i} = 1) - E(y_{2i}|A_{i} = 0) = (X_{1i} - X_{2i})\beta_{2i} + \sigma_{2\eta}(\lambda_{1i} - \lambda_{2i}).$$

Moreover, the income of the households who have participated in forest environmental resource commercialization may be higher (lower) because of their participation in forest environmental resource commercialization that those who didn't participate, if they had participated. This "transitional heterogeneity effect" (TH) (Carter and Milon, 2005; Di-falco et al, 2010;Di-Falco et al., 2011) is the difference between the effect of participation in forest environmental resource commercialization for the group of households who participated in forest environmental resource commercialization (TT) and those who did not (TU).

5.5. Results and Discussion

Estimated parameters for income function with endogenous switching due to participation in forest environmental resource commercialization are presented in table 5.4. For comparison, a single equation income function with no switching was estimated. A result from OLS estimates for the dummy variable (NTFC_C) equals 1 if the household participated in forest environmental resource commercialization or not is presented in the first column of table 5.4. The estimates of the selection equation (5.1) and for the income function for the group of households that have participated in forest environmental resource commercialization and those that did not participate is presented the last three columns of table 5.4.

<<INSERT TABLE 5.4 ABOUT HERE >>

Parameter estimates in table 5.4 revealed that the main factors that influence households' decision to participate in forest environmental resource commercialization include weather related shocks, sex of the household head being male, number of adult household numbers in the household and access to private forest. This finding is in line with the forest and livelihoods literature. For example with more adult members in the household, households may have access to labour that could be involved in forest environmental resource commercialization. In addition our result indicated that male headed households participate more in forest environmental resource commercialization. Our finding is similar to Adhikari (2005) because households headed by a woman might be at a disadvantage to participate in some income generating activities (e.g., women might not be allowed to work in agriculture) but they might also be restricted in their access to natural resources (Lopez –Feldman et al., 2011). Similarly, when insurance markets are not existent or incomplete households rely on different strategies to cover themselves against

covariate shocks. In this sense, it has been argued that environmental resources might provide households with a "natural insurance" (Pattanayak and Sills, 2001; Takasaki et al., 2004). Our finding also shows that households experienced with weather related shocks are more likely to participate in forest environmental resource commercialization. The positive sign in the variable distance to market also supports the hypothesis that households in isolated villages have less productive alternatives and are more likely to take part in low return activities like resource extraction and depend more on them as income generating activities (Lopez-Feldman et al., 2011).

Conversely, variables such as education, access to wage income and own business, and access to more land are found to be negatively correlated with forest environmental resource commercialization. With more education, households may have access to a broad variety of livelihood activities, have higher opportunity cost of time (Adhikari et al., 2004) and may disregard the collection and sales of forest resources. Similarly households with more assets may easily access more profitable livelihood activities and may give less priority to environmental resource incomes which are often considered "employment of the last resort" (Angelsen and Wunder, 2003).

We now turn to the contribution of participation in forest environmental resource commercialization to the household income. The simplest approach to investigate the effect of forest resource commercialization on the household income is an estimation of the income function via OLS using the participation in forest environmental resource commercialization as an explanatory variable as shown in the first column of table 5.4. The estimated result shows that participation in forest environmental resource commercialization affects household income positively though it is insignificant. However, this may be a misleading conclusion as OLS estimates do not explicitly account for potential structural differences between the income function

of households (Di-falco and Veroseni, 2010) that participated in forest environmental resource commercialization and the groups of households that did not. The likelihood ratio test indicates that the two equations are not independent (Prob> 0.00, table 2 bottom row)).

The endogenous switching in the income function estimates is presented in column 3 and 4. As shown in the 2nd row from bottom of the same table, the parameter coefficients of the terms ρ_j are statistically different from zero and have positive signs in the two equations implying (1) participation in forest environmental resource commercialization significantly increases income among participants; and (2) income levels are significantly higher among non-participants. This implies that the decision to participate in forest environmental resource commercialization and the income of households given the participation decision is affected by observed and non-observed factors. Moreover, results also show that there is self-selection in the participation of forest environmental resource commercialization as shown by the significance of the coefficient of correlation between the participation equation and the income of households that participated in forest environmental resource commercialization and the income of households that participation equation and the income of households that participated in forest environmental resource commercialization.

Finally, the income function of households that participated in forest environmental resource commercialization and those that did not is significantly different at 1 % level of significance. This implies the presence of heterogeneity in the sample (refer to the last two columns of table 5.4)

. <<INSERT TABLE 5.5 ABOUT HERE>>

Table 5.5 presents the expected income per capita under the actual and counterfactual conditions. Considering first the observed differences in the average household income per capita between the two groups of households as shown in cells (a) and (b) along the diagonal, the group of households that participated earned 3,808.88(26.3%) Ethiopian Birr less, on average, than the group of households that did not participate in forest environmental resources commercialization. This simple comparison is misleading however because it does not account for other unobserved factors (Di-falco and Veroseni, 2010; Carter and Milon, 2005) that may have affected income.

The treatment effect of participation in forest environmental resource commercialization is presented in the last column of table 5.5. For the group of households that participated in forest environmental resource commercialization, the first entry in the last column indicates that the mean effect of participation in forest resource commercialization (TT in table 5.3 and equation (5.5)) was an increase in per income of 1619.062 (17.9%) Ethiopian Birr. Similarly, if the group that did not participate in the forest environmental resource commercialization had participated (TU in table 5.3 and equation (5.6)), the mean effect would be a decrease in per capita income of 3630.48(25%) Ethiopian Birr. However, the effect of participate in forest resource commercialization is larger for the group that did not participate in forest resource commercialization, resulting in a positive value of the transitional heterogeneity effect (TH in table 5.3). The estimated treatment effects indicate that both groups, if they participated in forest environmental resource commercialization of participation in forest environmental resource commercialization in forest environmental resource commercialization in forest environmental both groups, if they participated in forest environmental resource commercialization in forest environmental resource commercialization, would decrease the contribution of participation in forest environmental resource commercialization to household income.

The base heterogeneity which shows the differences in the expected income per capita among households who participated in forest environmental resource commercialization and those that did not is presented in the last row of table 5.5. With the counterfactual condition that the non-participant group of households participated in forest environmental resource commercialization (BH1 in table 5.3 and equation (5.7)), the group who actually participated in forest environmental resource commercialization and every second to earn 174.8 (1.6 %) Ethiopian Birr less on

average. However, this difference is not statistically significant. Similarly, with the counterfactual condition that the group of households that participated did not participate in forest resource commercialization (BH2 in table 5.3 and equation (5.8)), the group of participants would still significantly earn 5427.942 (37.5 %) Ethiopian Birr less per year. Under both counterfactual conditions, the group of households that actually participated in forest environmental resource commercialization would earn less. These differences reflect systematic sources of variation between the two groups that could not be fully accounted for in the observable determinants of the income model.

5.6. Conclusions and Policy Implications

This chapter investigates the factors affecting households' decision to participate in forest environmental resource commercialization and how participation in forest environmental resource commercialization impacts on the income of rural households living along the peripheries of protected forests in northern highlands of Ethiopia. We use household level data from 251 sample households living along the margins of protected forests found in northern highlands of Ethiopia to estimate a without-participation counterfactual household income scenario, against which to compare actual with-participation income.

Our results are interesting in a number of respects. First, they show that the choice of methodology can have a significant impact on the estimated impacts and conclusions made about the impact of participation in forest environmental resource commercialization on the welfare of rural households. We compare the estimated impacts using the counterfactual approach discussed above, with the more simplistic and more commonly used approach which treats forest income as an exogenous addition to other sources of household income. We find that the contribution of forest environmental resource commercialization to the overall income is lower when the methodologically superior counter factual estimation is used.

Second, the estimated heterogeneity effects revealed that, independent of participation, participant households would have less income on average than non-participants, implying that the group of farm households that participated in forest environmental resources commercialization has systematically different characteristics from the group that did not participate. This systematic difference in income between the two groups could not be explained with observable determinants of the income model and indicates a more complex relationship between participation in forest environmental resource commercialization and household income, which could not be taken into account by the estimation of OLS regression.

Third, participation in forest environmental resource commercialization increases the income for the group of households who actually participated. However, the treatment effect analysis revealed the counterintuitive results that participation decreased income for non-participant groups in the counterfactual case they did participate. However, if this group of households had participated in forest environmental resource commercialization, they would have earned the same as the farm households that actually participated. Therefore, participation in forest environmental resource commercialization in forest environmental resource commercialization is participated for the group of farm households most vulnerable to shocks, those who have no access to other high yielding activities such as off farm work or adequate land, by helping them to fill the income gap with the less vulnerable group of households in the same locality. While the result is surprising, it is consistent with the previous studies on the forest – poverty nexus that the poor households are more dependent on forests than the rich (Jodha ,1986; Reddy and Chakravarty, 1999 ; Cavandish 2000; Mamo et al.,2007; and Babulo et al.,2008), that most NTFPs produce low returns (particularly per ha and often, also per

labour unit), are primarily used for subsistence and often to fill income gaps (Byron and Arnold, 1999; Newmann and Hirsch, 2000).

Finally, to the extent the results from this sample are representative of other areas and development policies, further research inquiries are also needed to evaluate the effects of policies that promote forest resource commercialization. In this regard, the distinction made by Fisher (2003) between low return forest resources (LRFR) and high return forest activities (HRFA) would be important.

The results from this analysis particularly raise several issues for the forest-based livelihoods research and policies. First, predictions based on the assumption that participation in forest environmental resource commercialization increases income over all households living along the margins of forests may overestimate the impacts of forests' contribution in the livelihood of people. In this regard public policies and participation of NGOs could help in providing alternative livelihood and income generating options and minimizing the dependence of households on forest environmental resources. The introduction of crop insurance and health related insurance are other possible policies. Providing access to off farm work and enhancing the existing productive safety net programs are paramount in order for households to take risks and invest in high risk and high return activities rather than depending too much more on forest environmental resources.

	Tota	1	Participants		Non-par	ticipants
	Mean	S. D	Mean	S. D	Mean	S. D
Variable name						
Dependent variables						
NTFP_C (1/0)	0.331	0.471	0.000	0.000	0.000	0.000
Log_inc_aue	9.161	0.670	9.200	0.554	9.142	0.749
Explanatory variables						
HH characteristics						
Hhh_sex	0.848	0.359	0.92	0.279	0.82	0.389
Age_hh_1	46.76	12.52	44.08	11.56	48.1	12.79
Family_size	5.745	2.114	5.843	2.334	5.964	2.002
Edu_hhh	1.167	2.184	0.964	1.685	1.268	2.391
Stu_number	1.509	1.435	1.12	1.334	1.71	1.445
Average_g	4.212	2.388	3.488	2.297	4.578	2.355
Adult_hhm	2.865	1.286	2.903	1.393	2.845	1.233
Aware_cc	0.355	0.479	0.337	0.476	0.363	0.482
Asset Holdings						
P_size _tsimdi	4.356	2.927	3.749	2.496	4.656	3.082
Tlu	3.107	2.585	3.107	2.252	3.142	2.740
Private_forest	0.450	0.499	0.482	0.503	0.435	0.497
F_saving	0.251	0.412	0.193	0.397	0.226	0.419
Access to factor marke	et and inputs	5				
Wage_income	1844.1	2252.455	1533.9	1405.503	1997.5	2560.003
Own_business	0.355	0.479	0.157	0.366	0.405	0.492
Access_credit	0.375	0.485	0.373	0.487	0.375	0.486
Access_irrigation	0.135	0.343	0.120	0.328	0.143	0.351
Log _dis_market	1.243	1.335	1.777	1.083	0.978	1.370
Log_dis_forests	1.708	0.828	1.637	0.786	1.742	0.847
Shock dummies						
Weather_s	0.797	0.403	0.892	0.034	0.75	0.434
Death_ill_s	0.227	0.420	0.289	0.456	0.196	0.398
Price_s	0.323	0.468	0.349	0.479	0.309	0.464
Loss_live_s	0.454	0.499	0.469	0.502	0.446	0.498
Sample Size	251		83		168	

Table 5. 1 Descriptive statistics of sample households

	Participants		Non-particip	oants	
	Mean	S. D.	Mean	S. D	P-Value
Variable name					
Income					
Log_inc_aue	9.200	0.608	9.142	0.058	0.5354
Explanatory					
variables					
HH characteristics					
Hhh_sex	0.92	0.279	0.82	0.389	0.0373**
Age_hh_1	44.08	11.56	48.1	12.79	0.0168**
Family_size	5.843	2.334	5.964	2.002	0.6055
Edu_hhh	0.964	1.685	1.268	2.391	0.3005
Stu_number	1.12	1.334	1.71	1.445	0.0017***
Average_g	3.488	2.297	4.578	2.355	0.0006***
Adult_hhm	2.903	1.393	2.845	1.233	0.7358
Aware_cc	0.337	0.476	0.363	0.482	0.6898
Asset Holdings					
P_size _tsimdi	3.749	2.496	4.656	3.082	0.0206**
Tlu	3.107	2.252	3.142	2.740	0.7606
Private_forest	0.482	0.503	0.435	0.497	0.4796
F_saving	0.193	0.397	0.226	0.419	0.5463
Access to factor mar	ket and inputs				
Wage_income	1533.9	1405.503	1997.5	2560.003	0.1251
Own_business	0.157	0.366	0.405	0.492	0.0001***
Access_credit	0.373	0.487	0.375	0.486	0.9816
Access_irrigation	0.120	0.328	0.143	0.351	0.6277
Log _dis_market	1.777	1.083	0.978	1.370	0.0000***
Log_dis_forests	1.637	0.786	1.742	0.847	0.3425
Shock dummies					
Weather_s	0.892	0.034	0.75	0.434	0.0086***
Death_ill_s	0.289	0.456	0.196	0.398	0.0998***
Price_s	0.349	0.479	0.309	0.464	0.5269
Loss_live_s	0.469	0.502	0.446	0.498	0.7268
Sample Size	83		168		

Table 5. 2 Mean separation test of participants and non-participants

*, **, *** significant at 10, 5 and 1 percent level of significance.

Table 5. 3 Definition of Expected income and treatment effects

	Decision				
Sub-Sample	Participation	Non Participation	Treatment Effect		
Households that participated	(a) $E(y_{1i} A_i = 1)$	(c) $E(y_{2i} A_i = 0)$	TT		
Households that did not participate	(d) $E(y_{1i} A_i = 1)$	(b) $E(y_{2i} A_i = 0)$	TU		
Heterogeneity effects	BH_1	BH_2	TH		
Notes: (a) and (b) represent observed expected income: (C) and (d) represent counterfactual expected					

Notes: (a) and (b) represent observed expected income;(C) and (d) represent counterfactual expected income

 $A_i = 1$ if the farm households participated in forest sales; $A_i = 0$ if farm households did not participate Y_{1i} : income if the household participate in forest resource commercialization

 Y_{2i} : income if the household did not participate in forest resource commercialization

TT: The effect of the treatment (i.e., participation) on the treated (i.e., farm households that participate

TU: The effect of the treatment (i.e., participation) on the untreated (i.e., farm households that did not participate)

 BH_i : the effect of base heterogeneity for farm households that participate in forest resource commercialization (i = 1), and did not participate (i = 2);

TH = (TT - TU), i.e., transitional heterogeneity.

Source : Di-falco and Veroseni (2010)

	(1)	(2)	(3)	(4)
		End	ogenous Switching Reg	gression ¹⁷
			Participation =1	Participation $= 0$ (if the
			(farm households	households did not
			that participate in	participate in forest
			forest	commercialization)
Model	OLS		commercialization	
	Annual	Participation	Annual Household	Annual Household
Dependent Variable	household	1/0	income in Ethiopian	income in Ethiopian
	Income in		Birr	Birr
	Ethiopian			
	Birr			
Participation 1/0	0.122			
	(0.078)			
Sex of the household head (1if male	0.331***	0.656**	0.289	0.433***
0, otherwise	(0.104)	(0.309)	(0.194)	(0.130)
Education of the hh head	-0.009	-0.007	0.368	-0.014
	(0.017)	(0.500)	(0.332)	(0.020)
Average education level in the hh	-0.018	-0.121**	-0.069**	-0.032
	(0.017)	(0.049)	(0.032)	(0.021)
Number of Adult hh Members	0.060**	0.300***	0.103**	0.087**
	(0.029)	(0.090)	(0.047)	(0.039)
Livestock in TLU	0.065***	0.048	0.128***	0.063***
	(0.016)	(0.047)	(0.029)	(0.020)
Plot size in Tsimdi	0.019	-0.131***	-0.203	0.006***
	(0.014)	(0.043)	(0.026)	(0.019)
Access to Irrigation	0.385***	-0.463	-0.008	0.486
	(0.102)	(0.289)	(0.165)	(0.130)
Access to Credit	-0.007	0.185	-0.044	0.0/4
F' '10 '	(0.076)	(0.208)	(0.119)	(0.100)
Financial Saving	0.441^{***}	0.255	0.093	0.548***
A () X / ·	(0.085)	(0.249)	(0.135)	(0.10/)
Access to Wage income	0.001^{***}	-0.001*	-1.960	0.001^{***}
Weather valated sheels	(0.000)	(0.000)	(0.000)	(0.000)
weather related shocks	-0.280^{++++}	(0.039^{***})	0.038	-0.188
Number of Students in the hh	(0.094)	(0.299)	(0.191)	(0.122)
Number of Students in the nn		-0.205		
Log distance o forest		(0.075)		
Log distance o torest		-0.143		
Log distance to Market		(0.113)		
Log distance to Market		(0.096)		
		(0.080)		

Table 5. 4 OLS and Endogenous Switching Regression estimates

¹⁷Estimation of full information maximum likelihood at household level.

Own Business		-0.746^{***}		
Access to Private forest		(0.243) 0.586*** (0.195)		
Constant	8.412***	-0.445	8.233***	8.426***
	(0.142)	(0.676)	(0.297)	(0.183)
σ_i			0.501***	0.583***
			(0.061)	(0.045)
$ ho_i$			0.655**	0.719***
			(0.187)	(0.149)
Adjusted R2	0.40			
Likelihood Ratio (LR) test of indepen	ndent Equation	s: chi2(2) = 1	10.40 $\text{Prob} > \text{chi2}$	= 0.0013
*,** and *** denotes signific	ant at 10%; 5%	b, and 1 % resp	ectively.	

Sample size: 251. σ_i denotes the square-root of the variance of the error terms ε_{ij} in the outcome equation (2a) and (2b), respectively; ρ_j denotes the correlation coefficient between the error term η_i of the selection equation(1) and the error term ε_{ij} of the outcome equations (2a) and (2b), respectively.

Table 5. 5 Expected income and treatment effects

	Forest Commercia		
Sub-Sample	Participation	Non Participation	Treatment Effect
Households that participated in forest	(a) 10673.91	(c)9054.848	TT = 1619.062**
commercialization	(568.001)	(536.196)	(394.436)
Households that did not participate in	(d) 10852.31	(b)14482.79	TU = -3630.48***
forest commercialization	(612.223)	(546.392)	(820.586)
Heterogeneity effects	$BH_1 = -178.4$	<i>BH</i> ₂ = -5427.942***	TH = 5249.542***
	(874.212)	(952.636)	(295.334)

See notes of table 5.2 Standard Errors in parenthesis *** and ** denotes significant at 1% and 5% respectively

Appendix

Table 51AVariables definition

Variable name	Definition					
Dependent variables						
NTFP_C (1/0)	Dummy = 1 if the household participated in forest resource					
	commercialization, 0 otherwise					
Log_inc_aue	Log income per adult equivalent unit in Ethiopian Birr					
Explanatory variables						
HH characteristics						
Hhh_sex	= 1 if the head of the household is male, 0 otherwise					
Age_hh_1	Household head's Age in years					
Family_size	Household size					
Edu_hhh	Educational level of the household head in years					
Stu_number	Number of Students in the household					
Average_g	Average educational grade level in the household					
Adult_hhm	Number of adult household members					
Aware_cc	Dummy = 1 if the household head is aware of climate change in his/her					
	locality, 0 otherwise					
Asset Holdings						
P_size _tsimdi	Plot size owned by the household in tsimdi (= 0.25 hectare)					
Tlu	Number of livestock owned in Tropical Livestock Unit (TLU)					
Private_forest	Dummy = 1 if the household owned private forest, 0 otherwise D_{i}					
F_saving	Dummy = 1 if the household has financial savings \mathbf{U}					
Access to factor marke	et and inputs					
Wage_income	Amount of wage income earned by the household in Ethiopian Birr					
Own_business	Dummy = 1 if the household engaged in his/her own business ,0 otherwise					
Access_credit	Dummy = 1 if the household has access to credit form micro finance 0.otherwise					
Access_irrigation	Dummy =1 if the household has access to irrigation, 0 otherwise					
Log _dis_market	Log transformed distance from home to market in minutes					
Log_dis_forests	Log transformed distance from home to nearest forest in minutes					
Shock dummies						
Weather_s	Dummy = 1 if the household experienced weather related shocks last					
	year, 0 otherwise					
Death_ill_s	Dummy =1 if the household experienced death or illness of a household member last year. 0 otherwise					
Price_s	Dummy = 1 if the household experienced price related shocks last year 0 otherwise					
Loss_live_s	Dummy = 1 if the household experienced shocks related to loss of $\frac{1}{2}$					
	livestock last year, 0 otherwise					

CHAPTER SIX

ENVIRONMENTAL RESOURCES, POVERTY AND INEQUALITY IN THE SEMI-ARID REGION OF TIGRAY, ETHIOPIA

6.1. Introduction

Poverty reduction is one of the international imperatives of the new millennium – an imperative stressed by the Millennium Development Goals (MDGs), the 2002 World Summit on Sustainable Development and, most recently, at the 2005 UN World Summit (Roe .D, and Elliot. J., 2005). With rural poverty accounting for some 75% of world poverty, meeting this goal, together with the high levels of spatial concurrence between regions rich in biodiversity and the majority of the world's poor, has compelled scholars and practitioners operating at the environment – development interface to seek solutions to poverty that include natural resource – based activities.

The underlying reasoning of poverty reduction through biodiversity conservation initiatives is multifaceted. One aspect is the premise that many tropical forests have a greater long-term economic value if they are left standing. Second, local forest communities will tend towards careful, sustainable management of the forest if they receive direct economic benefits from harvesting its resources. Third, poverty in local forest communities is both a cause and result of deforestation and environmental degradation. If poverty can be alleviated through the harvesting of forest resources, then deforestation pressures will be reduced (Newmann and Hirsch, 2000).

A literature has grown rapidly in recent years focused on examining the potential for forest-based poverty alleviation. Among the key authors are: Byron and Arnold (1999); Cavendish (2000); Arnold (2001); Wunder (2001); Scherr *et al.* (2002); Angelsen and Wunder (2003); Lopez –Feldman (2007); Mamo et al., (2007) and Babulo et al. (2009). Views diverge on whether there should be grounds for optimism or pessimism for forest-based poverty alleviation. Scherr *et al.* (2002) can be viewed as leaning in the optimistic direction, whereas Wunder (2001) clearly favours a pessimistic view (Sunderlin and Thu, 2005). A recurrent theme in this new literature is the need for additional research as there are many questions that remain to be answered (Angelsen and Wunder 2003).

Reducing income inequality has also become a major public policy challenge among development agencies and poverty reduction experts. In this regard the impact of Non Timber Forest Products (NTFPs) on income distribution is another important aspect. For example Griffin and Ickowitz (1997) stated "the importance of natural capital in the total stock of capital tends to vary inversely with the level of income per head". That is, the poorer the country, the more significant is natural capital likely to be in determining the overall distribution of wealth.

Inequality is also important for poverty discussion in many ways. One aspect is that people living in and around forests, just like other people, do not only care about their own absolute incomes:their welfare perceptions also depend on their income status "vis-a- vis "the joneses" – the individuals, households or communities of reference that they would like to keep up with" (Angelson and Wunder 2003). Second, it is also generally accepted that one of the causes of poverty is unequal distribution of assets and income and that those with few assets and income tend to over-exploit their resources. Third, where NTFPs already provide significant household incomes, the distribution of that income may be highly uneven and complex (Newman and Hirsch, 2000). Fourth, levels of income inequality can also reflect inequalities in rights over assets and resources that are used to generate income (Anderson et al. 2006). More research can elucidate these patterns, and in particular suggest how, why and under what circumstances inequality exists, and with what consequences to marginalized groups and individuals engaged in NTFP extraction. Yet, quantitative studies of the relationship between natural resources, poverty and inequality are scarce (Lopez 2007). With few exceptions, discussions about poverty reduction strategies and the role of forests often focus almost exclusively on income growth, neglecting the potential roles of income re-distribution and inequality.

Jodha (1986) seems to be amongst the first stream of researchers who attempted to rigorously shed more light on the distributional implications of forest environmental income on poverty and inequality. He found that the Gini coefficient in dry regions in India increases by as much as 34 percent when income derived from forest gathering is ignored in Gini estimation. Also in India Reddy and Chakravarty (1999) found that when forest income is set to zero in poverty calculations, poverty increases by as much as 28 percent. However, the inequality effect of ignoring forest income was very marginal. Similarly, Lopez-Feldman et al.(2007) in their study of rural Mexico and the Lacandona Rainforest community area of Mexico, observed that when forest income is ignored in poverty calculations, the severity of poor people increases more at the regional and community levels (17.1% and 18.4% respectively), than at national level(10.8%). The headcount and poverty gap measures revealed a similar pattern of greater sensitivity of poverty at the regional and community levels than at the national level. In their inequality calculations, it was also observed that when forest income is increased by 10 percent, the Gini coefficient reduces by as much as 0.36 and 0.11 percent, respectively, at the national and community levels. In the World Bank meta-study (Vedeld et al., 2004), the average increase in the Gini-coefficient when forest income was excluded was 0.13 (from 0.36 to 0.49). The same could be deduced from studies in Southern Malawi (Fisher 2004), Uganda (Aryal, 2002), Ethiopia (Mamo et al., 2007 and Babulo et al., 2009) and Nigeria (Fonta et al., 2011).

However, although the above studies explored the role of forest income in reducing rural poverty and inequality and tried to establish empirical regularity about the poverty reduction and equalizing effect of forest income, they suffer from methodological shortcomings. Their method provides a direct and simple measure of how forest income contributes to the total income under the simplifying assumption that forest income is exogenous and not a substitute for the participants' labour earnings on-farm and in other works. The focus of their analysis was principally on the decomposition techniques and the estimation of inequality indicators, using household income including and excluding forest incomes, without considering the interdependency of forest extraction and other household activities. Indeed, not distinguishing this interdependency could lead to biased decomposition of poverty and inequality indicators and misleading policy implications. In other words, these earlier studies did not attempt to estimate the opportunity costs of labour participation in forest extraction; that is, what the household members who participated in forest extraction would have contributed to household income had they not been involved in forest extraction. Neither were the indirect effects of forest income on other sources of income included in the analysis.

The implicit assumption of their approach is that forest income is an exogenous income transfer, which does not have any effect on labour supply, and other income sources of the household. This assumption implies that, for a given household, with a given level of farm income, an increase in forest income raises total income by the same amount. This could be true if the participation in forest resource extraction was to compensate a short- term shock, such as a bad harvest or drought/flood. But, more often than not, participation in forest extraction is a long term alternative choice of participation in farm activities for households such as those in rural Tigray. In fact Gopalakrishnan et al., (2005) tested the hypothesis of competitive /complementary labour allocation between NTFP gathering and agriculture in their study of the Sinharaja rain forest region of SriLanka. The results of their analysis showed that labour requirement for tea production is negatively and significantly related to NTFP, implying that NTFP gathering and agriculture are competing for labour and that the household members' time in forest commercialization would contribute to the household in alternative ways if they did not participate in forest commercialization.

In addition the decision to participate in forest extraction and the subsequent income also affect the household's exposure to income risks, as well as its investment and production decisions. For example, the income from forest extraction might provide insurance and relieve the household's budget constraint which in turn might lead the household to adopt riskier or more costly production techniques with higher potential returns. On the downside, forest extraction might also affect the household's labour participation and supply decision or forest extraction may lead to deforestation which boosts the time allocated to fuel wood collection and thereby reduces agricultural output by shifting household members' time away from agricultural work.

As a result, when the indirect effects of insurance, investment and liquidity prevail, the marginal impact of forest commercialization on the total household income is hypothesized to be greater than unity. In contrast when the disincentive effects prevail, the marginal impact will be less than unity. Inter temporal, long -run effects might also be at play if households decide to invest their

forest income in productive assets, which then provide an additional contribution to households' total income in subsequent years.

The task of the research then is to estimate what households' per capita income situation would have been in the absence of forest commercialization (due to restrictive policy such as area closures or resource depletion) - the counterfactual income level - and then to compare with the actual household per capita income with forest income included as part of total income. Previous studies on the areas of forest income and inequality elsewhere have neglected this issue, choosing instead to treat forest income as an addition to what the household would have otherwise earned. In other words, rather than estimating a counterfactual without forest extraction scenario, previous studies have simply compared actual income excluding forest income on both the level of household income (and poverty alleviation) and the distribution of income. In this paper we will compare the results following both methodological approaches, in order to allow us a clear estimation of the extent of over-estimation or under- estimation of these impacts in previous works.

Our results contribute a meaningful addition to the poverty-environment debate and Ethiopia's current conservation policy in many respects: First, we show that the choice of methodology can have a significant impact on the estimated contribution of forests and conclusions made about the role of forests on poverty and distribution of income. Second, we find that the contribution of forests to reduce the incidence of poverty is considerably weaker, and might increase the severity of poverty when the counterfactual estimation method is used. Third, in relation to income distribution, we find that after using the counterfactual approach, the impact of forest income on distribution is negative as against a positive contribution in the conventional approach. Fourth, these apparently contradictory findings, where forest income appears to reduce substantially the

poverty and inequality indicators when we use the conventional approach, but has little or negative impact on poverty and inequality indicators under the counterfactual approach, could be explained by the fact that participation in forest activities might have a disincentive effect on incomes from other activities which are not captured when forest income is treated as an exogenous addition to the existing household income.

The chapterr is outlined as follows: Section 6.2 provides an overview of forest management, poverty and distributional aspects in rural Tigray. Section 6.3 presents the data description and study sites. Section 6.4 outlines the econometric model and estimation procedure. Section 6.5 presents the results and discussion while the final section provides concluding remarks and policy implications.

6.2. Overview of Forest Management, Poverty and Distributional Aspects in Tigray

As late as the 1950s, nearly half of Tigray region's land area was covered in woodlands and forests, while less than 30 years later, in 1979 nine percent remained (Wolde-Giorgis, 1996 cited in Howard and Smith, 2006). Accelerated deforestation has led to severe soil erosion in regions where people are dependent on marginal, rain fed agriculture. Since 1991, the Ethiopian government has embarked on an economic development strategy known as Agricultural Development–Led Industrialization (ADLI), which places greater emphasis on agricultural development. Conservation programs also gained top priority in economic development in Tigray, which focuses on conservation of natural resources and community participation. Closing degraded land areas from human and livestock intervention to promote natural regeneration of plants, commonly termed as enclosures , is among the major constraint efforts practiced in the highland areas of Tigray, northern Ethiopia (Tewelde-Berhan, 2002; Babulo, 2007).
In addition to enclosures, the regional government created restrictions on the use of certain tree species and the restrictions apply to all common land areas, regardless of whether they are open access or enclosed (FAO, 2006). This strategy is meant to benefit the environment and is a conservation effort that the community itself is operating (Gebrehiwot and van der Veen, 2010). The immediate positive environmental effects of the enclosures and the species restrictions are nearly universally acknowledged both by researchers and by local populations (Howard and Smith, 2006) and the international community alike¹⁸. Nevertheless(Nedessa et al., 2005) called into question whether the results of the current conservation policy are effective or sustainable from an economic or environmental point of view. There is also a steadily growing body of literature questioning the impact of this policy in terms of its contribution to income and equity of distribution (Tesfay, 2006; Gebremedhin et al., 2004; (Chisholm, 2000; and Nedessa et al., 2005).

Local communities use woodlots/enclosures to cut and collect grass for animal feed and other purposes, to collect fruits and seeds, and to practice beekeeping .However, although these practices are widespread, they generate relatively small economic benefits (Gebremedhin et al., 2004), the economic benefits and costs of enclosures are distributed inequitably (Chisholm, 2000) and the rich households benefit triply (Nedessa et al., 2005). However, the arguments from both sides have been based essentially on qualitative narratives of the pros and cons of exclosure establishment and have not been backed up by quantitative reasoning and indicators (Babulo, 2007). For example, they evaluate the impacts of such policy on incomes and equity using only simple tabulations, without verifying if the reported regularities are significant using poverty and inequality decompositions. This indicates that there is untapped research potential to provide

¹⁸One of the communities in Tigray was awarded the 2012 UNDP Equator Prize for good management of the environment.

clarity under which conditions the current forest management contributes to poverty and distribution of incomes quantitatively. With the exception of (Babulo, 2007), current literature does not provide economic data that relates the contribution of the existing forest management to the distribution of income in the semi-arid region of Tigray.

6.3. Data Description and Study Sites

A household survey was conducted in 2010/11 on 254 households in Tigray region of northern Ethiopia. However, because of missing data on some key variables for 3 households, our final estimating sample consists of 251 households. A two stage sampling design was made in the study. The primary sampling units (PSUs) were tabias. Sample tabias were selected on the basis of secondary information collected from all the Woredas along the margins of the two protected forests in the region. In selecting the sample tabias, a number of factors that affect socio economic condition such as nearness to market, geographical location, the availability of both rain fed agriculture and irrigation, size of tabias based on population etc were considered so as to make the sampled tabias representative. In this category, a total of ten tabias namely Arato, Derga _ajen, Hugumrda, Meswaeti, Kara_adishawo, WorebayuKal_amin, Kelisha_emni,andFelege_woini were selected for the survey. The tabias selected are representative of the three different agroecological zones of the region identified on the basis of altitude. Areas with altitude ranging from 1500- 2300 m.a.s.l. are locally termed as *woinadouga* i.e midland areas, areas above 2300 m.a.s.l. are locally known as *douga*.e. highland areas and areas with altitude less than 1500 m.a.s.l. are termed as *kola* i.e., lowland. To determine the extent to which households in rural Tigray use forest environmental resources for their livelihood, we follow (Narrain et al., 2008) to calculate the income that each household obtained from major sources.

6. 4. Econometric Model and Estimation Procedure

Households who are involved in forest activities are likely to differ from households who are not. A household may participate in forest extraction if he/she anticipates that the decision to participate in forest related activity is worthwhile rather than entering by random assignment. Unobservable characteristics of households and their family may affect the participation decision in forest activities and the level of income, resulting in inconsistent estimates of the effect of participation on income (Di-falco et al, 2011). Accordingly, we used a simultaneous equations model with endogenous switching by full information Maximum likelihood (FIML) to determine the counter factual income of forest resource extraction following Di-falco et al., (2011) as shown below.

The selection equation for participating in forest activities is specified as:

(6.1)
$$P_i^* = z_i \alpha + \eta_i \text{ with } P_i = \begin{cases} 1 \text{ if } P_i^* > 0\\ 0 \text{ otherwise} \end{cases}$$

That is, households will choose to participate in forest extraction ($A_i = 1$) if $A^* > 0$, 0 otherwise, where A^* represents the expected benefits of participating in forest extraction with respect to nonparticipation in forest related activities, Z is a vector of observed household characteristics or variables that are influencing the decision to participate in forest commercialization or otherwise. To complete the system of simultaneous equations model for household income, conditional on participation in forest environmental resource commercialization, we specify two equations following(Di-falco et al., 2011) where (1) participation of households in forest extraction and (2) non-participation in forest activity are defined as follows:

(6.2a) Regime 1: $y_{1i} = X_{1i}\beta_1 + \varepsilon_{1i}$ if $P_i = 1$

(6.2b) Regime 2:
$$y_{2i} = X_{2i}\beta_1 + \varepsilon_{2i}$$
 if $P_i = 0$

Where y_i represents the income and food security level in regimes 1 and 2, X_i represents a vector of explanatory variables mentioned above.

Finally, the error terms are assumed to have a trivariate normal distribution, with zero mean and covariance matrix $\boldsymbol{\Sigma}$, i.e., $(\varepsilon_1, \varepsilon_2, \eta_1) \equiv N(0, \boldsymbol{\Sigma})$ with

$$\boldsymbol{\varSigma} = \begin{bmatrix} \sigma_n^2 & \cdot & \cdot \\ \sigma_{1\eta} & \sigma_1^2 & \cdot \\ \sigma_{2\eta} & \cdot & \sigma_2^2 \end{bmatrix},$$

Where σ_n^2 is the variance of the error term in the selection equation (6.1), (which can be assumed to be equal 1 since the coefficients are estimable only up to a scale factor), σ_1^2 and σ_2^2 are the variances of the error terms in the income function (6.2a) and (6.2b), and $\sigma_{1\eta}$ and $\sigma_{2\eta}$ represent the covariance of η_i , ε_{1i} and ε_{2i} . Since y_{1i} and y_{2i} are not observed simultaneously the covariance between ε_{1i} and ε_{2i} is not defined (Maddala, 1983 cited inDi-falco et al, (2011)). An important implication of the error structure is that because the error term of the selection equation (6.1) η_i is correlated with the error terms of the income equation (6.2a) and (6.2b) (ε_{1i} and ε_{2i}), the expected values of ε_{1i} and ε_{2i} conditional on the sample selection are non zero:

$$E[\varepsilon_{1i}|A_i=1] = \sigma_{1\eta} \frac{\phi(Z_i\alpha)}{\phi(Z_i\alpha)} = \sigma_{1\eta}\lambda_{1i}, \text{ and } E[\varepsilon_{2i}|A_i=0] = -\sigma_{2\eta} \frac{\phi(Z_i\alpha)}{1-\phi(Z_i\alpha)} = \sigma_{2\eta}\lambda_{2i},$$

Where $\phi(.)$ and $\phi(.)$ are the standard normal probability function, $\phi(.)$ the standard normal cumulative density function respectively, and $\lambda_{1i} = \frac{\phi(Z_i\alpha)}{\phi(Z_i\alpha)}$, $\lambda_{2i} = \frac{\phi(Z_i\alpha)}{1-\phi(Z_i\alpha)}$. Thus, if the estimates of the associated coefficients to the inverse Mill's ratios are found to be statistically different from zero, the hypothesis of the absence of sample selectivity bias can be rejected (Féres et al., 2007). This model is defined as a "switching regression model with endogenous switching" (Maddala and Nelson, 1975 cited in Di-falco et al., 2010). The full information maximum likelihood (FIML) is

an efficient method to fit the endogenous switching regression model (Di-falco et al, 2011; Linh et al., 2014).

The logarithmic function given the previous assumption regarding the distribution of the error term as presented by Di-falco and Veroseni (2010) is

(6.3)
$$\ln L_{i} = \sum_{i=1}^{N} A_{i} \left[\ln \phi \left(\frac{\varepsilon_{1i}}{\sigma_{1}} \right) - \ln \Phi + \ln \Phi(\theta_{1i}) \right] + (1 - A_{i}) \left[\ln \Phi \left(\frac{\varepsilon_{2i}}{\sigma_{2}} \right) - \ln \sigma_{2} + \ln (1 - \Phi(\theta_{2i})) \right],$$

Where $\theta_{ji} = \frac{(Z_i \alpha + \rho_j \varepsilon_{ji} / \sigma_j)}{\sqrt{1 - \rho_j^2}}$, j = 1, 2, with ρ_j denoting the correlation coefficient between the error

term η_i of the selection equation (1) and the error term of ε_{ji} of equation (6.2a) and (6.2b), respectively.

6.4.1. Expected outcome and Treatment Effects

The aforementioned endogenous switching regression model can be used to compare the expected income of the farm households that participate in forest extraction (6.4a) with respect to the farm households that did not participate (6.4b), and to investigate the expected income in the counterfactual hypothetical case (6.4c) that the participant farm households did not participate. The conditional expectations for income status in the four cases are defined following Di-falco et al. (2011) as follows:

(6.4a)
$$E(y_{1i}|P_i = 1) = X_{1i}\beta_1 + \sigma_{1\eta}\lambda_{1i}$$

(6.4b)
$$E(y_{2i}|P_i=0) = X_{2i}\beta_2 + \sigma_{2\eta}\lambda_{2i}$$

(6.4c) $E(y_{1i}|P_i = 0) = X_{1i}\beta_2 + \sigma_{2\eta}\lambda_{1i}$

Cases (6.4a) and (6.4b) represent the actual income expectations observed in the sample for participants and non-participants respectively while case (6.4c) represents the counterfactual income of participants in the hypothetical case that the participants did not participate in commercialization of forests. Having simulated the income obtained if the household did not participate in forest commercialization, we can study the effect of forest income on rural households' poverty and inequality. We calculate respectively poverty measured by class of P_{α} indices (Foster et al., 1984) and inequality measured by Gini Coefficient of:

- (1) Observed income, which is the sum of (6.4a) and(6.4b)
- (2) Observed income, without forest income
- (3) Observed income for non-participants and the counter factual income for participants under the scenario of non-participation, which is the sum of (6.4b) and (6.4c).

If P_{α} under (1) which is the actual income including forest income is lower than P_{α} under (2) which is the observed income excluding forest income and the counterfactual scenario (3), forest incomes reduce poverty, and vice versa. Following the same idea, we study the impact of forest income on inequality. Overall three basic scenarios are analysed as follows.

In the first scenario- observed income without forest income – the indicators are estimated using observed income excluding forest income for all households. This scenario is used by all the previous studies in forest poverty analysis including those of Cavendish (2000); Botha (2003); Fisher (2004); Lopez-Feldman (2007) and Babulo et al (2009). This scenario treats forest income as an exogenous addition to income from other sources.

The second scenario – counterfactual income – uses the basic counterfactual methodology discussed in section 6.4.1 to estimate what the poverty indicators would be without participation in forest resource commercialization. The FGT indices are calculated using observed income for non-participants, and the counterfactual income for participants under the scenario of non-participation.

Finally, the third scenario uses the counterfactual methodology and the FGT index without forest income methodology to see whether previous studies such as those by Cavendish (2000); Botha (2003); Fisher (2004); Lopez-Feldman (2007) and Babulo et al (2009) were overestimating or underestimating the contribution of forest environmental resources on poverty and inequality.

6.4.2. Measuring Poverty

To analyse the contribution of forest incomes to poverty and inequality, we start from the standard Foster–Greer-Thorbecke (FGT) class of measures that incorporates the three most common poverty indexes. If we let $y = (y_1, y_2, ..., y_k)$ represent household income in increasing order and z > 0 denote the predetermined poverty line, the FGT poverty line is defined by:

(6.5)
$$P(y;z) = \frac{1}{nz^{\alpha}} \sum_{k=1}^{q} g_k^{\alpha}$$

Where *n* is the total number of households, q = q(y; z) is the number of poor households, $g_k = z - y_k$ is the income short fall (the gap between the household's income and the poverty line) of the k-th (poor) household, and α is a parameter.

Three variants of the FGT poverty index are used to estimate the impacts of income from forests on rural poverty: the head-count measure $\alpha = 0$, $P_H(y; z) = \frac{q}{n}$ measures the incidence of changes in forest income on rural poverty (i.e., the share of the population living below the poverty line);

the poverty gap $\alpha = 1$, $P_G(y; z) = \frac{1}{nz} \sum_{k=1}^{q} z - y_k$ measures the depth of poverty, or how far below the poverty line the average poor household's income falls; and finally, the squared poverty gap $\alpha = 2$, $P_{SG}(y; z) = \frac{1}{nz^2} \sum_{k=1}^{q} (z - y_k)^2$ measures the severity of poverty and is sensitive to changes in the distribution of income among the poor (Taylor et al., 2005).

6.4.3 Measuring inequality

To explore the impact of forest income on rural income inequality, we use Gini indices. Although various indices that can be used to measure inequality exist, the Gini is probably the most intuitive and allows for decomposition by income sources (Taylor et al., 2005). If we let $y = (y_1, \dots, y_l)$ represent *I* components of household income, and if we define total income as $= \sum_{i=1}^{l} y_i$, we can write the extended Gini coefficient for village incomes as a function of the covariance between income and its cumulative distribution:

$$(6.6)G_Y = \frac{2 \operatorname{cov}[Y,F(Y)]}{\overline{Y}}$$

where \overline{Y} is the mean of total income Y and F(Y) is the cumulative distribution of total incomes in the village. If we use the properties of covariance, we can re-write equation (10) as:

$$(6.7)G_Y = \frac{2\sum_{i=1}^{1} cov \left[y_{i}, F(Y)\right]}{\bar{Y}} = \sum_{i=1}^{1} R_i G_i S_i$$

In equation (6.7), S_i is the share of income from source *i* in total income, $S_i = \overline{Y}_i / \overline{Y}$, R_i is the Gini correlation of income from source *i* with the distribution of total income (CBO, 2011), and G_i is the Gini index corresponding to the distribution of income from source *i*.

$$(6.8)R_{i} = \frac{cov\left[y_{i,F(Y)}\right]}{cov\left[y_{i,F(Y_{i})}\right]}$$

In equation (6.7) the aggregate Gini coefficient G_Y that measures the total income inequality in a given population is expressed as the sum of the product's i^{th} income component in our case forest income, relative to its own Gini, and its correlation with the total income (Stark et al., 1986). One of the basic rationales for decomposing the total Gini to component income sources is to investigate how changes in a particular income source (in our case forest income) will affect the total income inequality, holding income from other sources constant.

Taking household labour and production decisions as given, if we increase an income source j by a factor e such that $y_i(e) = (1 + e)y_i$, the marginal effect on the Gini of total income is

$$(6.9)\frac{\partial G_Y}{\partial e} = S_j (R_j G_j - G_Y)$$

where S_j , R_j , G_j and G_Y are measured prior to the marginal income change, and the relative effect is given by

$$(6.10)\frac{\partial G_Y/\partial e}{G_Y} = \frac{S_j G_j R_j}{G_Y} - S_j$$

If R_j which is the Gini correlation between forest income and total income is negative or zero, an increase in forest income would decrease inequality. However, if R_j is positive, then the impact of an increase in forest income on inequality depends on the sign of $R_jG_j - G_Y$. Inequality will increase if the inequality of forest income exceeds the inequality of the total household income: $G_j > G_Y$ (Stark et al., 1986).

6.5. Results and Discussion

6.5.1 Descriptive Statistics

Out of the 251 households in the data set, about 33% of the households were participating in forest commercialization during the year 2009/10. The average age of sample households is 46.76 years and 85% are male headed. There is also a significant difference observed in the age and gender of participants and non-participants. In addition participants seem to have significantly lower number of students and lower average grade of educational level than their counterpart non-participant households. The average land holding in tsimdi (=0.25 hectare) for participants is 3.75 while it is about 4.7 for non- participants and the difference is statistically significant. The average walking distance to nearby market is significantly higher for participants.

<< INSERT TABLE 6.1.ABOUT HERE>>

6.5.2 Poverty and inequality Analysis

Using the conditional expectations and treatment effects discussed in section 6.4.1, we estimated the expected income of participants in the counterfactual hypothetical case that the participant group of households did not participate. This counterfactual income, which assumes participation in forest-related activity to be zero, is used to calculate per capita household income adjusted by adult equivalent scales. For comparability purposes, we use the FGT poverty index for the poverty analysis, while the Gini coefficient is used as a measure of inequality as discussed in section 6.4.3. We set the income poverty line 2100 Birr (an inflation-adjusted poverty line of the official baseline (poverty line of Birr 1075 set in 1995/96) as a measure of welfare corresponding to some minimum acceptable standard of living in Ethiopia(MOFED, 2006). Further more, we undertook ordinal poverty comparisons using stochastic dominance tests to check the robustness of the poverty ordering using the Distributive Analysis Stata Package (DASP).

Tables 6.2, 6.3 and 6.4 present the results for the analysis of poverty indicators, which were estimated using the household level data weighted by household size. The tables show the estimated FGT indices under each scenario, as well as their percentage change when compared against each indicator obtained using actual household income, including forest income.

A. Estimated Poverty indicators: Conventional vis-a-vis Counterfactual Scenario

Comparing the role of forest income in poverty and inequality indicators under each scenario shows some interesting similarities and differences with previous studies in Ethiopia (Babulo et al.,2009 and Mamo et al.2007) and elsewhere as discussed below.

<< INSERT TABLE 6.2 ABOUT HERE >>

Results from table 6.2 which uses the conventional approach of estimating the contribution of forests to poverty and distribution of income indicates that forest incomes lead to a decline in the incidence of poverty (P_0) from 61.6 % to 43.4%¹⁹, in the depth of poverty (P_1) from 28.6% to 16.1 % and in the severity of poverty (P_2) from 17.1% to 8.1%. The strong impact on the depth and severity of poverty suggests that participation in forest environmental resource commercialization reduces the income gap among the poor and the gains in poverty reduction due to participation in forest environmental resource commercialization go disproportionately to the poorest households. This finding is consistent with Reddy and Chakravarty (1999); Cavandish (1999); Fisher (2004); Lopez – Feldman (2007) and Babulo et al (2009).

<< INSERT TABLE 6.3 ABOUT HERE >>

¹⁹In its five year (2010/11-2014/15) regional GTP, the regional government reported that the incidence of poverty to be 41% in the year 2009/10. In this regard our estimate reflects the real condition of the study area.

Table 6.3, which uses the predicted income of households that participate in forest environmental resource commercialization under the condition that they did not participate - the counterfactual approach, indicates that participation in forest environmental resource related activities still leads to a decline in the incidence of poverty (P_0) from 61.6 % to 53.4 %. These results provide strong evidence that participation in forest environmental resource commercialization and the subsequent income alleviate poverty in rural Tigray. However, under this scenario, the poverty headcount ratio decreases by 22.9 % while it declines by 44.2 % under the first scenario. Moreover, unlike the first scenario the depth of poverty (P_1) and the severity of poverty (P_2) increase from 15.8% to 16.1% and from 6.2% to 8.1% respectively. Even though it is not significant, the negative impact on the depth of poverty suggests that participation in forest related activities increases the income gap among the poor. And the negative impact on the severity of poverty, which assigns higher weights to the poorest of the poor, suggests that participation in forest activities did not improve the wellbeing of the poor disproportionately. In other words, the gains in poverty reduction due to participation in forest environmental resource commercialization did not significantly go disproportionately to the poorest households. In this regard our finding differs from those of Reddy and Chakravarty (1999); Lopez - Feldman (2007) and Mariara and Gachoki(2008), who found that forest income has positive impacts on poverty, depth of poverty and severity and that the impact is greater in terms of lessening dire poverty than its effect in lifting poverty in India, Mexico and Kenya respectively.

<< INSERT TABLE 6.4 ABOUT HERE >>

Comparing the results of the counterfactual approach and the conventional approach in table 6.4, we observe that the contribution of forest income to poverty is substantially overestimated when the methodology, which simply treats forest income as an addition to the household income, is

used. The estimated contribution of forests is substantially lower when we use the more complex methodology in which forest incomes are assumed to be endogenous in the hypothetical, no participation in forest-related activity scenario. Using this counterfactual approach the contribution of forests to the incidence of poverty (P_0) is decreased by 0.0836, and the depth (P_1) and Severity (P_2) of poverty by 0.1284 and 0.1086 respectively. To check the robustness of our results we also compared the differences on poverty indexes between the conventional approach and the counterfactual approach vis-à-vis the poverty indices including forest income as shown in figures 6.1, 6.2 and 6.3. Our results unambiguously confirm that the contribution of forests to the incidence of poverty (P_0) is still positive but lower when we use the counterfactual scenario than the conventional without forest scenario regardless of the poverty line used. In contrast we observe that forests decrease the severity of poverty (P_2) if we use the conventional approach, while the severity of poverty (P_2) increases under the counterfactual approach. This is again robust regardless of the poverty line used. However, the difference on the contribution of forests to the depth of poverty(P_1) between the two approaches shows that it is lower under the counterfactual approach, yet comparing the result to the poverty indices that include forest income we observed that the result is inconclusive and sensitive to the poverty line used.

These results confirm that the choice of methodology for computing the contribution of forests on poverty has a significant impact on the results and the conclusions drawn, and that care should be taken to use appropriate methodology before reaching any conclusion and policy recommendations on the forest –poverty nexus.

<< INSERT FIGURES 6.1, 6.2 AND 6.3ABOUT HERE >>

B.Estimated Gini index: Conventional Vis-à-vis Counterfactual Scenario

<< INSERT TABLE 6.5 ABOUT HERE >>

When forest incomes are excluded from the total household income, i.e. under the conventional approach shown in table 6.5 above, the average increase in the Gini coefficient is 0.05 (from 0.34 to 0.39). This finding is again consistent with previous studies on the distributional impact of forest income in Zimbabwe (Cavendish 2000), Malawi(Fisher, 2004), Uganda (Aryal, 2002), Mexico (Lopez- Feldman 2007), Ethiopia (Babulo et al.2009 and Mamo et al.2007) and in the World Bank meta-study (Vedeld et al., 2004).

<< INSERT TABLE 6.6 ABOUT HERE >>

When the hypothetical counterfactual estimation method is used, the Gini coefficient shows a substantial decline by 0.06 (from 0.34 to 0.28) suggesting that forest income increases inequality, which is contrary to the findings of the conventional approach. Comparing the Gini index using both approaches as shown in table 6.7 below, we observed that the contribution of forests to income distribution is again overestimated under the conventional approach. Under the conventional approach, the Gini index is estimated to be 0.39 while the same index under the counterfactual approach is 0.28 shows a significant and substantial difference of 0.11.

<< INSERT TABLE 6.7 ABOUT HERE >>

The gap in the contribution of forests to income distribution when we use the conventional approach and counterfactual approach scenario, against which to compare actual with forest income, is further illustrated using the Lorenz curve in figure 6.4. The figure shows that the addition of forest income to household income reduces measured income inequality. This result accords with the previous studies in the literature. However, under the hypothetical counterfactual

estimation technique that we use it shows that participation in forest-related activity increases inequality.

<< INSERT FIGURE 6.4 ABOUT HERE >>

Although it is based more on conjecture and not supported by inequality decomposition, Narain et al. (2009) cautioned that the distributional outcomes of forest extraction would be worse if we are to include biodiversity resources in our analysis of inequality. Moreover, our results are similar to Adhikari et al., (2004) and Sapkota and Oden (2008).

These apparently contradictory findings, where forest income appears to reduce substantially the poverty and inequality indicators when we use the conventional approach that has been used in the forest-poverty nexus, but has little or negative impact on poverty and inequality indicators under the counterfactual approach, could be explained by the fact that participation in forest activities might have a disincentive effect on incomes from other activities which have higher returns to labour, which are not captured when forest income is treated as an exogenous addition to household income. For example participation in forest commercialization might affect the household's labour participation and supply decision or forest commercialization may lead to deforestation which boosts the time allocated to fuel wood collection and thereby reduces agricultural output by shifting household members' time away from agricultural work.

6.6 Conclusions and Policy Implications

This research investigates the contribution of forest commercialization to poverty and income distribution in semi-arid region of the Tigray regional state, Ethiopia. We use household survey data to estimate without forest income and counterfactual household income scenario, against which to compare actual with forest income.

Our study contributes to the study of the Poverty –Environment nexus in many respects.First, it provides an overview of forest management, poverty and distributional issues in the rural Tigray region of Ethiopia. Literature can be found on the measurement of poverty and inequality in relation to forests in Tigray but none has focused on the counterfactual income scenario.

Second, our results show that the choice of methodology can have a significant impact on the estimated contribution of forests and conclusions made about the role of forests on poverty and distribution of income. We compare the estimated contribution of forests using the counterfactual approach discussed in section 6.4, with the simpler and more commonly used method which treats forest income as an exogenous addition to other sources of household income. We find that the contribution of forests on poverty measures is considerably weaker, and despite forests decreasing the incidence of poverty they might increase the severity of poverty when the counterfactual estimation method is used. This is a new finding that has not been documented in previous studies. However, it has important policy implications. If forests increase the severity of poverty and if this forest management policy is left unresolved, the poor remain poorer than their counterparts and will eventually be less able to escape from the poverty trap. Our finding is consistent with the previous studies and conclusions made that the existing conservation policy in Ethiopia, particularly in the region has only environmental benefits and that economic benefits are yet to emerge.

Third, in relation to income distribution, we find that after using the counterfactual approach, the impact of forest income on distribution is negative as against a positive contribution in the conventional approach.

Fourth, these apparently contradictory findings, where forest income appears to reduce substantially the poverty and inequality indicators when we use the conventional approach, but has

159

little or negative impact on poverty and inequality indicators under the counterfactual approach, could be explained by the fact that participation in forest activities might have a disincentive effect on incomes from other activities which are not captured when forest income is treated as an exogenous addition to the existing household income.

Over all, this study makes a meaningful addition to the poverty-environment debate and Ethiopia's current conservation policy to better distribute resource rights, both property and procedural. The improved management of trees and forests cannot be pursued in isolation, through sectoral efforts and requires careful consideration of biological, economic, socio-cultural and institutional factors. Equally, policy should facilitate and strengthen linkages between agriculture and income diversification which provides alternative livelihood opportunities and pathways out of poverty. Finally inequality of access to natural assets must be addressed to warrant a win –win for conservation of the environment and poverty reduction.

	Households participate	that did not	Households participate	that	
Variable name	Mean	Std. Dev.	Mean	Std. Dev.	P- Value
Log_inc_aue	9.142	0.749	9.200	0.554	0.5354
Hhh_sex	0.82	0.389	0.92	0.279	0.0373**
Age_hh_1	48.1	12.79	44.08	11.56	0.0168**
Family_size	5.964	2.002	5.843	2.334	0.6055
Edu_hhh	1.268	2.391	0.964	1.685	0.3005
Stu_number	1.71	1.445	1.12	1.334	0.0017***
Average_g	4.578	2.355	3.488	2.297	0.0006***
Adult_hhm	2.845	1.233	2.903	1.393	0.7358
Aware_cc	0.363	0.482	0.337	0.476	0.6898
P_size _tsimdi	4.656	3.082	3.749	2.496	0.0206**
Tlu	3.142	2.740	3.107	2.252	0.7606
Private_forest	0.435	0.497	0.482	0.503	0.4796
F_saving	0.226	0.419	0.193	0.397	0.5463
Wage_income	1997.5	2560	1534	1406	0.1251
Own_business	0.405	0.492	0.157	0.366	0.0001***
Access_credit	0.375	0.486	0.373	0.487	0.9816
Access_irrigation	0.143	0.351	0.120	0.328	0.6277
Log _dis_market	0.978	1.370	1.777	1.083	0.0000***
Log_dis_forests	1.742	0.847	1.637	0.786	0.3425
Weather_s	0.75	0.434	0.892	0.034	0.0086***
Death_ill_s	0.196	0.398	0.289	0.456	0.0998***
Price_s	0.309	0.464	0.349	0.479	0.5269
Loss_live_s	0.446	0.498	0.469	0.502	0.7268
Sample Size	168		83		

Table 6. 1 Mean separation test of participants and non-participants

*, **, *** significant at 10, 5 and 1 percent level of significance.

Table 6. 2 Poverty decomposition: Scenario 1

	Without forest income		With fore	st income		
	Estimates	S.E	Estimates	S.E	Difference	t-value
FGT (0)	0.6175	0.0307	0.4343	0.0314	0.1832	7.49***
FGT (1)	0.2864	0.0188	0.1606	0.0149	0.1258	11.6***
FGT (2)	0.1709	0.0143	0.0813	0.0098	0.0896	9.91***

***,** and * represent 1, 5 and 10 percept level of significance respectively

Table 6. 3 Poverty decomposition: Scenario 2

	Counter factual income		With fore	st income		
	Estimates	S.E	Estimates	S.E	Difference	t-statistic
FGT (0)	0.5339	0.0316	0.4343	0.0314	0.0996	3.06***
FGT (1)	0.1580	0.0122	0.1606	0.0149	-0.0026	-0.19
FGT (2)	0.0623	0.0063	0.0813	0.0098	-0.0190	-2.23***

***,** and * represent 1, 5 and 10 percept level of significance respectively

	Counter factual income		Without	forest income		
	Estimates	S.E	Estimates	S.E	Difference	t-statistic
FGT (0)	0.5339	0.0316	0.6175	0.0307	-0.0836	-2.68***
FGT (1)	0.1580	0.0122	0.2864	0.0188	-0.1284	-8.66***
FGT (2)	0.0623	0.0063	0.1709	0.0143	-0.1086	-9.33***

***,** and * represent 1, 5 and 10 percept level of significance respectively

Table 6. 5 Gini Index: Scenario 1

	Without forest income		With fore	est income		
	Estimates	S.E	Estimates	S.E	Difference	t-statistic
Gini Index	0.3920	0.0198	0.3425	0.0164	0.04955	5.36***

***, ** and * represent 1, 5 and 10 percept level of significance respectively

Table 6. 6 Gini Index: Scenario 2

	Counterfactual Income		With fore	est income		
	Estimates	S.E	Estimates	S.E	Difference	t-statistic
Gini Index	0.2817	0.0139	0.3425	0.0164	0.0607	5.36***

***, ** and * represent 1, 5 and 10 percept level of significance respectively

Table 6. 7 Differences in Gini index under counterfactual and conventional scenarios: Scena	rio	3
---------------------------------------------------------------------------------------------	-----	---

	Counter factual income		Without	forest income		
	Estimates	S.E	Estimates	S.E	Difference	t-statistic
Gini Index	0.2817	0.0139	0.3920	0.0198	0.1103	5.19***

***, ** and * represent 1, 5 and 10 percept level of significance respectively



Figure 6. 1 FGT Curves for Incidence of Poverty



Figure 6. 2 FGT Curves for Depth of Poverty



Figure 6. 3 FGT Curves for Severity of Poverty



Figure 6. 4 Lorenz Curve of Forest Income Distribution, Different Scenarios

CHAPTER SEVEN

IMPACT OF THE PRODUCTIVE SAFETY NET PROGRAM (PSNP) IN ETHIOPIA IN CONSERVATION OF FORESTS

7.1. Introduction

Poor people often depend on biodiversityboth for their livelihoods and as a safety net against deeper poverty (IIED, 2010). In addition, of the 1.2 billion people estimated to live on less than US\$ 1 a day (i.e. those that are the target of MDG1), 70% live in rural areas with a high dependence on natural resources for all or part of their livelihoods (Roe, D., 2003).. Moreover, direct dependence on services from ecosystems is highest amongst people living in arid and semi-arid lands such as Tigray where alternative livelihood options areoften limited and environments are particularlyfragile and risky (MA 2005b cited in Shackleton et al.2008).

However, this may constitute a poverty trap as a result of a tragedy of the commons (Hardin 1968). Too many households in rural Tigray depend on these resources as part of their livelihood and the resources cannot provide enough to properly insure all the population. They face thus the classic poverty-environment nexus, where poor people depend on the environment, and over use it (Delacote, 2007). The resultant increased pressure often has a negative feedback on the capacity of the ecosystems to deliver these services and creates a downward spiral of increasing poverty and ecosystem degradation (Shackleton et al., 2008).

Accordingly, finding synergistic solutions has been on local, national, and international agendas for decades, in the hope that humandevelopment and biodiversity conservation can be found to be less a zero-sum game of trade-offs and more a set of mutually reinforcing goals (Timmer, 2005). Yet there is little evidence that increasing incomes or poverty reduction affects the way in which poor people exploit natural resources (Roe and Elliott, 2005). To this end research is needed under which win-win solutions can be found to advance poverty reduction and conservation biodiversity such as forest environmental resources. The objective of this paper is therefore to investigate whether the Productive Safety Net Program (PSNP) which is one of the largest social protection programs in Africa provides an appropriate framework to create synergies for poverty reduction and conservation of the environment.

The Productive Safety Net Program (PSNP) is a development-oriented social protection program launched in Ethiopia in 2005 (Debela et al., 2014). The PSNP has a short term objective ofprotecting the poor against shocks (consumption smoothing) as well as a long term objective of poverty reduction, growth enhancement and natural resource conservation (Nega et al., 2008). The PSNP operates as a safety net, targeting transfers to poor households in two ways, through Public Works and direct Support (Gilligan et al., 2008).

Given the importance of the program at national level, a rigorous impact evaluation design has been followed since the very early stages of the program and lauded for its success in achieving its objectives such as reducing chronic and transient poverty (Fredu et al., 2010); consumption; food security and assets (Gilligan and Hoddinot 2007; Gilligan et al., 2008); livestock and tree holding (Andersson et al., 2011); human capital formation and child labour (Woldehana 2010; Hoddinot et al., 2009); and crop portfolios (Andersson, 2010). More than eight years in the program, however, there has not been any attention on the potential of the PSNP for exacerbating or mitigating forest pressure. On the one hand, the government explicitly states that the PSNP has a short term objective of protecting the poor against income shocks (income smoothing) as well as a long term objective of poverty reduction, growth enhancement and natural resource conservation. Therefore, the government clearly has an expectation that PSNP will result in improved environmental outcome. We focus here on forests as the environmental outcome of interest which is a key local resource and global public good.

However, even if we limit the scope to the relationship between public safety nets and forests, previous empirical results are mixed. Alix –Garcia et al., (2010) investigate the impact of large income transfers in Mexico called Oportunidades. They find that the income transfer increases deforestation, at least in the population that is just below the poverty level required to be able to receive payments. On the other hand, Fisher and Shively (2005) find that household forest use responds negatively to a receipt of a positive income shock (technology assistance package)in Malawi. Moreover, many of the empirical literatures on income-deforestation links have been hampered by concerns about the endogenieity of income growth (Alix –Garcia et al., 2010).

The objective of this study is therefore to analyse whether the interests of the global community, the Ethiopian government and local farm households can be reconciled through the PSNP. If the PSNP generates higher and more stable income while minimizing pressure on forests, it could be argued that such a program promotes conservation of forest biodiversity, i.e. renders development compatible with sustainability. The findings of this analysis bear implications for current discussions and initiatives seeking to promote "win-win" strategies for regions where poverty reduction and conservation of biodiversity are all high priorities and to the global efforts to limit greenhouse gas emissions from degradation and deforestation of forest ecosystems.

Analysing Ethiopia's PSNP allows us to make two contributions to the existing empirical literature. First, the implementation of the PSNP creates an exogenous source of variation in income, allowing for clean identification of causal effects that may be significant in predicting the impact of policy changes. Second, it contributes to the program evaluation literature by showing how a class of widely implemented public safety nets has important positive/negative environmental externalities.

We use propensity score matching methods to control the two important selection biases in the program²⁰, i.e. targeting of the program to recipients based on the characteristics unobservable to the researcher and self-selection into the program by eligible receipts (Gilligan and Hoddinot, 2007).

The chapter is outlined as follows: section 7.2 presents a brief introduction of the PSNP in Ethiopia. Section 7.3 provides the theoretical framework used for the study. Section 7.4 presents the empirical strategy. Section 7.5 presents empirical results and discussions, while the final section presents concluding remarks and policy implications.

7.2. Overview of the Productive Safety Nets Program (PSNP) in Ethiopia

Chronic food insecurity has been a defining feature of the poverty that has been affected millions of Ethiopians for decades (Hoddinot et al., 2009). One of the most important and tenacious problems that is argued to either cause or aggravate the drought-poverty-famine nexus in the plough based cereal farming system in the highlands is severe natural resource degradation, particularly manifest in soil erosion, deforestation, and degradation (Howard and Smith,

²⁰The two most important sources of selection bias include "program placement" bias, resulting from effective targeting of the program to poor households, and self-selection bias, resulting from the fact that households that choose to participate in the program may be different from households with access to the program that choose not to participate.

2006).These factors often interact with one another resulting in a reinforcing cycle of the "poverty, food insecurity and natural resource degradation trap". This problem is most acute in Ethiopian highlands (where over 85% of the country's population lives), which is affected by recurrent drought and famine affecting millions of people (FAO, 2003).

Between 1994 and 2003, an average of five million Ethiopians were declared "at risk" and in need of emergency assistance every year (RHVP, 2007). As a result Ethiopia has been the largest recipient of food aid in Africa and one of the largest recipients in the world in the last two decades (Little, 2008 cited in Andersson et al., 2011). However, decades of large –scale food aid deliveries have done little to resolve the problem. Instead, dependency of food aid has steadily increased over time, as has the number of "chronically food insecure" Ethiopians. As a result, the number of individuals in need of emergency food assistance rose from approximately 2.1 million people in 1996 to 13.2 million in 2003, before falling back to 7.1 million in 2004 (Gilligan et al., 2008).

To address this problem of food insecurity and dependence on the previous system of emergency appeals, the Ethiopian government and a consortium of donors that include European Commission(EC), World Bank (WB), Development cooperation of Ireland (DCI), United States Agency for International Development (USAID), Canada International Development Agency (CIDA) and the UK Department for International Development (DFID) launched a new social protection program called the Productive Safety Net Programme (PSNP) in 2005.

The main objective of PSNP is reducing household vulnerability, improving households' resilience to shocks and breaking the cycle of dependence on food aid by bridging production deficits in chronically food insecure farming households, protecting household assets from distress sale and creating community assets (Devereux et al., 2006) using a multi annual, predictable resource framework for predictable problems (RHVP 2007). With an annual budget of nearly US\$ 500 million, the PSNP is the second largest social protection program in Africa, reaching more than 7 million Ethiopians (Gilligan et al., 2008).

Food aid targeting in Ethiopia has a long history of relying on community-based targeting systems which have been seen as effective. The PSNP adopted this system while further refining the targeting criteria to capture chronic food insecurity—defined as a three months food gap or more and receiving food aid for three consecutive years. The PSNP uses a mix of geographic and community-based targeting to identify chronically food-insecure households in chronically foodinsecure woredas. After determining PSNP eligibility based on these criteria, households are assigned to public works or direct support: eligible households with able-bodied adults receive transfers for their participation in public works projects, while those households that cannot provide labor or other means of support receive unconditional transfers. Most beneficiary households participate in public works (90% of all PSNP transfers); a much smaller proportion receives direct support. In 2008, the program operated in the 290 most food-insecure of the approximately 670 woredas in rural Ethiopia (Coll-Black et al, 2013). Despite some regional variations PSNP has superior targeting in comparison to any of the African safety net programs by international standards (Coady et al, 2004). Generally, the targeting principles laid out in the Project Implementation Manual (PIM) are being followed and households do not generally perceive that religious affiliation, ethnicity, or personal connections affect the likelihood of selection into public works, the one exception being Oromiya (Coll-Black et al, 2013).

The PSNP operates as a safety net, targeting transfers to the poor households in two ways, thorough Public Works and Direct Support (Gilligan et al., 2008).Public works are labour intensive community-based activities which are designed to provide employment for chronically food

171

insecure people who have labour (MoARD, 2006), while the direct Support component provides assistance to households who are labour-poor and do not have reliable support (Woldehanna, 2009). Social transfers (direct support) contribute to risk coping and public works has elements of risk reduction, risk mitigation and risk coping, depending on what kinds of public work activities are undertaken (Devereux and Guenther 2007).

There are several ways in which we expect the PSNP to reduce the risk of vulnerability. It can (1) reduce income fluctuations because it increases income irrespective of shocks and thus have the same insurance properties as a permanent income that allow participants to save for the future and bad times; (2) displace non- desirable coping strategies such as child labour, depletion of assets and depletion of forest ecosystems. However, while the impact of PSNP with regard to child labour and depletion of assets has been undertaken there is no systematic study on the impact of PSNP on forest ecosystem conservation. In fact empirical studies on the role of public safety nets and forests have been mixed (see for example Monica and Shively, 2005 and Alix et al. 2010).

7.3. Conceptual Framework for Labour Allocation

The productive Safety Net Program (PSNP) combines a short term objective of protecting the poor against shocks (consumption smoothing) with a long term objective of poverty reduction, growth enhancement and natural resources conservation. With the exception of minor direct benefits for some households, the overall design for the PSNP in Ethiopia is directly related to employment of adults in beneficiary households. This public work requirement means that the program is not neutral in terms of adult labour supply and work incentives. To investigate the impact of PSNP on forest use, we draw on a farm household model and empirical studies of household labour allocation in developing countries (Abdulai and Delgado 1999; Jacoby 1993; Rosenzweig 1980; Fisher et al., 2002) similar to that presented by Fisher et al., (2005).

The model explicitly accounts for the fact that farm households in Ethiopia are both producers and consumers of agricultural and forest goods, and that markets for key factors and products are weak or absent. As a result, production decisions are influenced by consumption needs, and so production and consumption decisions in the model are assumed to be made jointly in response to changes in input and output prices.

We assume households allocate family labour across four major activities: Agricultural production, forest use, PSNP, and self employment. The household seeks to maximize household utility:

Max
$$U = U(C, F, O, L; H)$$
 (7.1)

Where utility is derived from consumption of a representative agricultural $\operatorname{crop}(C)$, a composite forest $\operatorname{product}(F)$, leisure (*L*) and other $\operatorname{goods}(O)$. We assume household and individual characteristics (*H*) influence preferences. Utility is maximized subject to production functions for crop and forest products, a full income constraint, a time constraint, and non-negativity constraints:

$$Q = Q[L, X, A, A(L; K; I)]$$
(7.2)

$$Q = A(L; K, I) + f(L; K)$$
 (7.3)

$$Y = p (Q - C) + p (Q - F) + p L + p L + R - p O - p X$$
(7.4)

$$T - L$$
 (7.5)

$$F, C, O, X, Q, Q, L, \ge 0$$
 (7.6)

Equation (7.2) describes smallholder production of crops which is assumed to be a function of labour(L), purchased inputs such as fertilizer and seed (X), the household's land endowment (A)

and additional land acquired through land clearing, represented by a function A(*). Cultivated area in the model is endogenously determined. Note that crop production can occur either through intensification (*via X*) or extensification (*via A*(*)), or both. Although customary land ownership implies that land markets are generally absent in much of rural Ethiopia, land can be "purchased" by using labour (*L*) and capital (*K*), e.g. an axe to clear uncultivated and possibly forested land. The existence of forest management institutions(*I*) also enters as an argument in *A* reflecting the potential for institutions to restrain forest clearing.

Equation (7.3) describes production of forest goods. The production function A(*) illustrates that when forest is cleared for agricultural expansion, forest products arise as a joint product. The technology f(*) describes forest "thinning" activities in which household labour is used to extract products from the forest, but land is not cleared in the process. Note that the existence of forest management institutions (*I*) appears as an argument in A(*) but not f(*). This is consistent with the patterns of forest management in Ethiopia, which tends to be more effective at restraining forest clearing than limiting collection of forest products.

Equation (7.4) defines the household's full income. Prices and net hourly returns to labour are denoted by a vector of prices p. Households earn income from four sources: agriculture, forest use, PSNP and other non-/off-farm employment. Households also receive remittances(R^*), defined here as money received from relatives. Households make expenditures on crops (C), forest products (F), other goods (O) and agricultural inputs (X). A positive (negative) sign for (Q - C) and (Q - F) indicates the household is a net-seller (net-buyer) of crop and forest products. Equation (7.5) describes the household's time constraint. Finally, a set of non-negativity constraints (7.6) completes the model.

Two important assumptions inherent in the model should be noted. One, we assume that households sell but do not hire labour. Two, we assume that households do not engage in production of cash crops – beyond sales of surplus crop production. While the assumptions are strong, and not appropriate in the context of rural Ethiopia as a whole, they are reasonable within the context of the sample. Most sample households are net purchasers of food, constrained in both cash and crop and thereby rarely able to hire labour (often paid either with cash or in kind). Only a few sample households engaged in cash crop production during the sample period, partly because coffee, Ethiopia's main cash crop, has historically been produced outside the study area.

The Lagrangian for the household's maximization problem is:

$$L = U(C, F, 0, T - L; H)$$

$$-\lambda \begin{bmatrix} Y - p \{Q[L, : X, A, A(L; K, I)] - C\} \\ -p \{[A(L; K, I) + f(L; K)] - F\} \\ -p L - p L - R + p 0 + P X \end{bmatrix}$$
(7.7)

After some rearrangement of terms, first-order conditions for the problem can be expressed as:

$$\frac{\partial U}{\partial L} = \lambda p \frac{\partial Q}{\partial L} \quad (7.8a)$$

$$\frac{\partial U}{\partial L} = \lambda p \frac{\partial Q}{\partial A} \frac{\partial A}{\partial L} + \lambda p \frac{\partial A}{\partial L} + \lambda p \frac{\partial f}{\partial L} \quad (7.8b)$$

$$\frac{\partial U}{\partial L} = \lambda p \quad (7.8c)$$

$$\frac{\partial U}{\partial L} = \lambda p \quad (7.8d)$$

$$p \frac{\partial Q}{\partial x} = p \quad (7.8e)$$

211

$$\frac{\partial U}{\partial c} = \lambda p \quad (7.8f)$$

$$\frac{\partial U}{\partial F} = \lambda p \quad (7.8g)$$

$$\frac{\partial U}{\partial 0} = \lambda p \quad (7.8h)$$

$$Y = p (Q - C) + p (Q - F) + pL + pL + R - pO - pX(7.8i)$$

Equations (7.8a) through (7.8d) indicate that, at the optimum, households allocate labour across activities so as to equate the marginal value of household leisure with that of time spent on each productive activity, i.e. value of marginal product or net hourly returns to labour. Equations (7.8e) thorough (7.8h) equate marginal values with prices. Equation (7.8i) recovers the full income constraint.

Expressions of labour supply, input demand, and commodity demand can be derived as functions of all exogenous variables.

$$\left. \begin{array}{l} L\\L\\L\\L \end{array} \right\} = g\left(p, p, p, p, A, H, K, I, T \right)$$

$$(7.9)$$

Consider the Slutsky equation relating a change in the net hourly returns from PSNP to the forest labour share. This is:

$$\frac{\partial L}{\partial p} = \frac{\partial L}{\partial p} \left| \pm \frac{\partial L}{\partial Y} \right|$$
(7.10)

where the first term on the right hand side of equation (7.10) is a substitution effect and is unambiguously non-positive. The second term is an income effect. While *L* is non-negative, the sign of $\partial L/\partial Y$ may be positive or negative. With rising income the demand for leisure (*L*) should increase if leisure is a normal good, but for some reason the demand for forest products (F) should also increase. More leisure should mean a lower forest labour share. However, higher consumption of forest products could imply an increase in the forest labour share. This would be the case, for example, if the household collected rather than bought additional forest goods. Such behaviour might be expected for a household that is a net seller of forest products. In sum, the net effect of a change in the returns to PSNP on forest labour share is ambiguous. A negative relationship whereby higher income from PSNP reduces forest pressure is plausible and could arise under several different scenarios: if forest products are inferior goods, if forest products are normal goods but the income – induced demand for leisure outweighs that for forest products, if forest products are normal goods but the household buys rather than collects forest products, or if a negative substitution effect dominates a positive income effect. A positive relationship between p and L could arise if the income induced demand for forest goods outweighs that for leisure, the household is a net-seller of forest goods and the income effect dominates the substitution effect. The combination of these effects implies that the impact of PSNP on the pressure on forests is ambiguous from a theoretical point of view. The presence of any impact, and its direction, is ultimately an empirical question.

7.4. Methodological Framework

Let the impact of the PSNP on forest outcomes of interest *Y* for treated households be given by:

$$Y_i^T - Y_i^C \tag{7.11}$$

Where *T* refers to the outcome of treated households and *C* denotes the counterfactual, the outcome for the same households had they been untreated. The above formalization assumes that each household is characterised by a pair of potential outcomes: Y_i^T for the outcome under the active

treatment (participation in PSNP) and Y_i^C for the outcome under comparison treatment (non-participation in PSNP).

Since it is impossible to observe the household treatment effect – because we cannot observe both treated and untreated situations for the same household- we aim to infer the impact of PSNP through its average effect in the population. If data on the outcome(s) of interest are available for treated (participants in PSNP) and untreated (non- participants in PSNP), we can then use the differences in outcome of treated and untreated households to obtain a (naive) estimate of the program impact:

$$E[Y_i^T - Y_i^C] \tag{7.12}$$

In this case, the average outcome of untreated households is intended to act as a substitute for the unobservable counterfactual. However, treated households are usually different in a set of observable household characteristics – such as income, food security, and asset- from those households who are not targeted to the programme. This problem will be magnified if households self-select into the programme (for instance, with the aim of securing additional income, or due to their social connectedness etc), meaning that unobservable factors such as motivation are a key determinant freatment assignment. This makes it difficult to isolate the differences between both groups which are due to already existing distinctions before the treatment – the selection bias-from those which are due to solely to the program's impact, as it can be seen by expanding equation (7.12) above:

$$E[Y_i^T - Y_i^C] = E[Y_i^T | T] - E[Y_i^C | C]$$
$$= E[Y_i^T | T] - E[Y_i^C | T] + E[Y_i^C | T] - E[Y_i^C | C]$$

$$= E[Y_i^T - Y_i^C|T] + E[Y_i^C|T] - E[Y_i^C|C]$$
(7.13)

The first term in equation (7.13) represents the average treatment effect on the treated, usually the parameter of interest we want to isolate (which will be formally defined below). This parameter will only be defined if the selection bias, represented by the second and third terms above, equals zero; this in turn, will only happen if there are no systematic differences in the average untreated outcomes between treatment and comparison groups.

Equation (7.13) guides us also regarding the direction of the bias. If households which participated in PSNP are on average more dependent on forests than their counterparts in the untreated group to begin with, then the selection bias term will be positive and the impact of PSNP on forest dependency will be overestimated. Conversely, if the comparison groups (untreated) households are more dependent on forests on average than those treated, then the selection bias term will be negative and the estimated treatment effect will underestimate the true impact of PSNP on forest dependency.

An evaluation design in which the selection bias problem tends to disappear is that in which treatment and comparison groups are randomly selected from a large population of potential beneficiaries. In other words, it may be assumed that, on average, those treated by PSNP are not different from those who did not participate either regarding observable characteristics (food security, education, age) or unobservable ones. Consequently, any statistically significant difference in the outcome variable (forest dependency) can be reliably attributed to PSNP.

In most real situations, nevertheless, public safety net programs such as the PSNP have been purposely implemented by governments – for instance by targeting households who are very poor on average than the average poverty level- and/or require households to self-select into the
programme by taking up the benefits. Various quasi-experimental and non-experimental methods have been used to address the bias problem (for details, see Heckman et al., 1998, Rosenbaum and Rubin, 1983 and Smith and Todd, 2001). One of the most commonly used quasi-experimental methods is Propensity Score Matching (PSM), which selects PSNP beneficiaries and nonbeneficiaries who are as similar as possible in terms of observable characteristics expected to affect participation in PSNP as well as outcomes under certain assumptions. The differences in outcomes between the two matched groups can be interpreted as the impact of the program on the beneficiaries (Smith and Todd, 2001). These assumptions and the formal definitions of alternative matching estimators for investigating the impact of PSNP on forest dependency will be explained in the next section.

7.4.1. Definitions of Average Treatment Effects

The empirical literature on program evaluation has traditionally focused on estimating three main average treatment effects that include the Average Treatment of the Treated (ATT) or the participation effect, the Average Effect of the Untreated (ATU) and the Average Treatment Effect (ATE) when assessing the impact of single treatment such as PSNP. Becker and Ichino (2002) indicate that the parameter of interest in the estimation of propensity score is the ATT. However, under the assumption of unconfoundedness, Titus (2007) argued that an appropriate approach of evaluating policy–relevant outcomes in a counterfactual framework is to examine not only the ATT, but also the ATU, which captures the effect of treatment on households who do not participate in PSNP. The definitions and identification conditions of each of these treatment effects will be discussed for the same context of evaluating the impact of PSNP on the outcome variable of interest for both participants and non-participants.

Average Treatment of the Treated (ATT): This parameter represents the average impact of PSNP on the outcome variable of interest (forest dependency) for those households who have been targeted for it. Formally, it is defined as:

$$ATT = E[Y_i^T - Y_i^C | T] = E[Y_i^T | T] - E[Y_i^C | T]$$
(7.14)

The second term after the last equality in equation (7.14) is the counterfactual to be estimated. The ATT is a measure of the average gain from PSNP to the treated households rather than to any households in the sample.

Average Treatment Effect on the Untreated (ATU): This alternative estimation represents the expected average impact of PSNP on the outcome variable (forest dependency) among those who have not been treated. In formal terms

$$ATU = E[Y_i^T - Y_i^C | C] = E[Y_i^T | C] - E[Y_i^C | C]$$
(7.15)

The first term after the last equality in equation (7.15) cannot be observed and must be estimated. The ATU parameter recovers the expected impact of PSNP on forest dependency on households randomly drawn from the sub-population of non-participants of PSNP, and is potentially useful to assess the impact of a program expansion to initially untreated households.

Average Treatment Effect (ATE): This is the third treatment effect corresponding to the average impact of PSNP for the entire population, whether or not a particular household has been treated, formally:

$$ATE = ATT * P(T) + ATU * P(C)$$

$$ATE = E[Y_i^T - Y_i^C | T] * P(T) + E[Y_i^T - Y_i^C | C] * P(C)$$
(7.16)

Where P(T) and P(C) are the probabilities of belonging to the treatment and comparison groups, respectively. In the sample these probabilities correspond to the sample frequencies of treated and untreated households. As can be seen, counterfactuals must now be estimated for both components of the *ATE*. This parameter is relevant for poverty reduction interventions that could be universally expanded, addressing the question of what the treatment gain would be to randomly selecvted households of the population.

Estimation of the Average Treatment Effects using matching methods relies on two key assumptions. The first is the conditional independence assumption (CIA), which states that assignment to treatment group is independent of the outcomes but based on observable characteristics (selection on observables). Matching on every covariate is difficult when the set of covariates is large. To solve this problem, we estimated the *propensity score*²¹ given as follows:

$$p(X) = p(D = 1 | X = x) = E[D|X = x]$$
(7.17)

The second assumption is the common support or overlap assumption. The common support is the area where the balancing score has positive density for treatment and comparison groups. No matches can be made to estimate the Average Treatment Effects when there is no overlap between the treatment and control groups.

²¹The propensity score is defined as the conditional probability the household will participate in PSNP conditional on observed characteristics(x), Where D = 1 when the household participates in PSNP, and D=0 otherwise.

Several methods are possible for selecting matching observations. We used kernel matching method (using the normal density of Kernel), which uses a weighted average of "neighbours" (within a given range in terms of the propensity score) of a particular observation. Unlike the nearest–neighbour method, using a weighted average improves the efficiency of the estimator (Smith and Todd, 2011).

7.5. Empirical Results

7.5.1. Descriptive Statistics

A household survey was conducted in 2010/11 254 households in the Tigray region of northern Ethiopia. However, because of missing data on some key variables for 3 households, our final sample consists of 251 households (for details of the survey please refer section 2.4 of the thesis).

The dependent variable used in the study is a dummy variable that takes the value of one, if the household participated in the Productive safety net Program, and a value zero, if no participation was recorded. The outcome variables used in this study are one binary indicator of whether the household participates in forest environmental collection or not (forest_ext), forest dependency, measured in terms of the ratio of forest income to the total household income (forest_dep) and number of trips to forests by the household in the survey year (forest_trips). These outcome variables employed were chosen to be in line with those suggested in the literature (see for example, Vira and Kontoleon, 2010).

<<< INSERT TABLE 7.1 ABOUT HERE >>>

Descriptive statistics of the variables used in the analysis are also presented in table 7.1 and 7.2. About 67% of the sample households were participating in the PSNP, while the remaining 33% did not participate. The survey also collected extensive information on several factors, including household characteristics, asset endowments, income and expenditure, household food security and vulnerability to shocks.

In many of the variables, we observed non-participants to be in a better position than the participant households. For example, non-participants had an average higher number of months food stock available, higher income per adult equivalent, better access to irrigation and transfer than their participant counterparts. Conversely, participants were vulnerable to weather and health related shocks. This is not however surprising, given the objective of the program is to protect the poor against shocks (consumption smoothing) as well as a long term objective of poverty reduction.

<< INSERT TABLE 7.2 ABOUT HERE>>

With regard to the outcome variables, our descriptive statistics did not show significant difference between participants and non-participants. These findings suggest that PSNP may not have a role in the outcome variables, but given that there are two important sources of selection bias in the PSNP - i.e. targeting of the programme to recipients based on characteristics unobservable to the researcher and self-selection into the programme by eligible recipients (Gilligan and Hoddinot, 2007) – a simple comparison of the outcome variables of the participants and non-participants has no causal interpretation. Therefore, we needed to conduct careful multivariate analysis to manage this econometric problem and test the impact of PSNP on the outcome variables.

7.5.2. Propensity Score Estimation Results

The propensity scores which were estimated with a probit model are reported in table 7.3. Propensity scores help as a device to balance the observed distribution of covariates across the treated and untreated groups (Lee, 2008). A detailed interpretation of the propensity score estimates is not undertaken in this study. However, the results shown in table 7.3 indicate that most of the variables included in the estimators have the expected signs.

<< INSERT TABLE 7.3 ABOUT HERE>>

The density distribution of the propensity score for participants and no participants in the program shows a good overlap (see figure 7.2) suggesting that the common support condition is satisfied. The bottom half of the graph shows the propensity score distribution for the non-participants (untreated), while the upper –half refers to participants (treated).

<< INSERT FIGURE 7.2 ABOUT HERE>>

Table 7.4 demonstrates how matching restricts the control sample in order to increase the similarity of the subsample of control cases that are directly compared with treated cases, in order to estimate the consequences of treatment. Table 7.4 presents the balancing information for the propensity scores and for each covariate before and after matching. We used the standardized bias differences between treatment and control samples as a convenient way to quantify the bias between treatment and control samples. In many cases we found that sample differences in the unmatched data significantly exceed those in the samples of matched cases. The process of matching thus creates a high degree of covariate balance between the treatment (participants in PSNP) and control groups (non participants) that are used in the estimation procedure.

The imbalances between the treatment and control samples in terms of the propensity score had been more than 100 % before matching as shown in the table 7.4. This bias was significantly reduced to a level of 3.1% after matching. The same table also shows that before matching, several variables exhibit statistically significance differences, while after matching the covariates are balanced.

The low pseudo R^2 and the insignificance likelihood ratio tests also support the hypothesis that both groups have the same distribution in covariates *X* after matching (see table 7.5). These results clearly show that the matching procedure is able to balance the characteristics in the treated and the matched comparison groups. We therefore used these results to evaluate the impact of PSNP among groups of households having similar observed characteristics. This allows us to compare outcomes for participants with those of a comparison group showing a common support.

7.5.3. Average Treatment Effects

The results of treatment effects (ATT, ATU and ATE) all estimated by Kernel matching method are presented in table 7.6. Starting from the ATT, the matching estimates reported in table 7.6 generally indicate that participation in PSNP exerts negative and significant effect in all of the outcome variables. Specifically the ATT effect of 0.0438 indicates that participation in PSNP result in lowering the probability of households' participation in forest environmental resource extraction (Forest _ext) by 4.38 %.

<< INSERT TABLE 7.5 AND TABLE 7.6 ABOUT HERE>>

The effect of PSNP on forest dependency was also found to be negative and statistically significant, again implying that PSNP helped in lowering the dependency of households by 11.8 %. Thus, PSNP appears to be crucial in minimizing the pressure on forest ecosystems. The magnitude of the coefficients of the treatment effects indicate that the average treatment effect of the treated (ATT) is higher than the Average treatment effect of the untreated (ATU) and the average treatment effect of the entire sample (ATE). This finding reveals that households that have a higher probability of participating in PSNP are able to reduce their dependency on forests over and above those the households that have less probability of participating in PSNP. The implication of this

result is thus, the gains in terms of reducing forest dependency as the result of PSNP are higher for households with higher probability of participating PSNP than those of households with lower chances of involving in the program. Also presented in table (7.6) are the critical levels of gamma (Γ) called the Rosenbaum test obtained using Γ bounds command in STATA 10. The Rosenbaum test is proposed by Rosenbaum (2002) to test if our causal inference about the impact of PSNP on the outcome variables would be altered by unobservable factors that affect both treatment and outcome variables. The p-critical values represent the lower bound of the p-value. Given that the estimated treatment effect is negative, the upper bounds under the assumption that the true treatment effect has been overestimated were less interesting (Becker and Caliendo, 2007). As a result table 7.6 shows that the negative effect of PSNP is not sensitive to selection bias due to unobserved variables, even if we allow participants and non-participants to differ by as much as 40 % in terms of unobserved covariates. The critical value of Γ , at which point we would have to question our conclusion of a negative impact of PSNP, starts from $\Gamma = 1.4$. However, Becker and Caliendo (2007) argued that these sensitivity results are worst case scenario. Overall the critical values in the table 7.6 clearly indicates that even unobserved heterogeneity would not alter the influence about the treatment effects, suggesting that the findings are generally insensitive to hidden bias. As a result we can conclude that the treatment effects estimate in table 7.6 are a pure effect of PSNP.

7.6.Conclusions and Policy implications

This paper investigates the links between the Productive Safety Net Programme (PSNP) and pressure on forests in North highlands of Ethiopia. We estimated the propensity score matching model to account for self-selection bias that normally occurs when unobservable factors influence

both participation in PSNP and outcomes such as participation in forest environmental resource extraction, forest dependency and number of trips to forests that the household had.

Results of the propensity matching score suggest a negative and significant effect of PSNP and the outcome variables. These findings are generally consistent with the widely held view that programs designed to reduce economic vulnerability of low income households can improve human welfare and reduce forest pressure. Our results are similar with the findings of Fisher and Shively (2005), that households who were involved in the Starter Pack program in southern Malawi had lower forest extraction as compared to households without the program.

Our study complements a small amount of literature that highlights on the intended and unintended impacts of programmes designed for poverty reduction on conservation of forest ecosystems and biodiversity. However, the need for further research is indisputable because households are likely to respond differently to income changes from poverty reduction programs that are perceived to be substantial and permanent versus small and temporary (Alex- Garcia et al, 2010). Further research is thus suggested using panel data spanning several years and in different geographical locations. If the results of our study are confirmed through additional analysis, then they indicate that the growing interest of policy makers in Ethiopia in pursuing the twin goals of sustainable development – raising human living standards and conservation of forest biodiversity – are moving in the right direction. Moreover, our study has important implications for Reducing Emissions from Deforestation and Forest Degradation (REDD) in developing countries using programs such as the Productive Safety Net program in Ethiopia.

PSNP has elements of risk reduction, mitigation and risk coping; as a result it can reduce income fluctuations because it increases income irrespective of shocks and thus has the same insurance

188

properties as permanent income which allows beneficiaries to save for the future and bad times, and in the end displace non-desirable coping strategies such as depletion of forest ecosystems.

Given that the poor households are most dependent on forest ecosystems as source of livelihood and coping mechanisms, which exerts pressure on common pool resources such as forest biodiversity and ecosystems, policy measures could target them to lower their dependency on forest environmental resources. Finally, a wider set of policies should aim to reduce poor households' vulnerability to shocks and improve access to alternative livelihood strategies by providing improved access to micro credit, agriculture and health extension services..

	Total Sample		Participants	3	Non-participants		
	Mean	Std. Dev.	Mean	Std. Dev.	Mean	Std. Dev.	
Variable name							
Treatment Variable							
PSNP (1/0)	0.67	0.33	0.00	0.00	0.00	0.00	
Outcome Variables							
Forest_extraction(1/0)	0.85	0.36	0.86	0.34	0.81	0.40	
Forest_dependency	0.25	0.20	0.26	0.21	0.23	0.21	
Forest_trips	74.8	51.1	73.7	52.2	76.9	49.1	
Explanatory variables							
HH characteristics							
Hhh_sex	0.85	0.36	0.86	0.35	0.83	0.38	
Age_hh_1	46.7	12.5	45.2	11.6	50.0	13.8	
Edu_hhh	1.17	2.18	1.21	2.30	1.07	1.94	
Family_size	5.75	2.11	5.89	2.08	5.47	2.17	
Dep_ratio	0.48	0.20	0.49	0.18	0.46	0.24	
N_dep	2.88	1.59	2.30	1.51	2.66	1.72	
Asset Holdings and acco	ess to input	s and factor	market				
P_size _tsimdi	4.36	2.92	4.20	2.87	4.68	3.04	
Tlu	3.11	2.59	2.97	2.38	3.39	2.96	
Private_forest	0.45	0.50	0.48	0.50	0.38	0.49	
Fooda_months	5.70	3.42	5.30	3.27	6.52	3.61	
Access_irrigation (1/0)	0.14	0.34	0.08	0.27	0.25	0.43	
M_hh_ext_pack	0.17	0.37	0.15	0.36	0.19	0.39	
Transfer (1/0)	0.30	0.46	0.25	0.43	0.42	0.50	
I_per_aue	2209	1527	2059	1263	2513	1928	

Table 7. 1 Descriptive statistics of sample households

T_asset_v	1316	2100	1323	1617	1301	2849
I_per capita						
Dummy_own _b	0.32	0.47	0.34	0.47	0.29	0.46
fol	0.50	0.50	0.52	0.50	0.46	0.50
Shock Dummies						
Weather_s	0.80	0.40	0.84	0.37	0.71	0.46
Death_ill_s	0.22	0.41	0.17	0.38	0.31	0.47
Loss_live_s	0.45	0.50	0.46	0.50	0.43	0.49
Sample Size	251		168		83	

	Participants		Non-partic		
	Mean	Std. Dev.	Mean	Std. Dev.	P-Value
Variable name					
Forest_extraction(1/0)	0.86	0.34	0.81	0.40	0.200
Forest_dependency	0.26	0.21	0.23	0.21	0.301
Forest_trips	73.7	52.2	76.9	49.1	0.642
Hhh_sex	0.86	0.35	0.83	0.38	0.593
Age_hh_1	45.2	11.6	50.0	13.8	0.004***
Edu_hhh	1.21	2.30	1.07	1.94	0.629
Family_size	5.89	2.08	5.47	2.17	0.148
Dependency_ratio	0.49	0.18	0.46	0.24	0.309
N_dep	2.30	1.51	2.66	1.72	0.127
P_size _tsimdi	4.20	2.87	4.68	3.04	0.222
Tlu	2.97	2.38	3.39	2.96	0.230
Private_forest	0.48	0.50	0.38	0.49	0.149
Fooda_months	5.30	3.27	6.52	3.61	0.008***
Access_irrigation (1/0)	0.08	0.27	0.25	0.43	0.000***
M_hh_ext_pack	0.15	0.36	0.19	0.39	0.449
Transfer (1/0)	0.25	0.43	0.42	0.50	0.005***
T_asset_v	1323	1617	1301	2849	0.940
I_per_aue	2059	1263	2513	1928	0.026**
Dummy_own _b	0.34	0.47	0.29	0.46	0.426
fol	0.52	0.50	0.46	0.50	0.327
Weather_s	0.84	0.37	0.71	0.46	0.017**

Table 7.2 Mean separation test of participants and non-participants in PSNP

Death_ill_s	0.17	0.38	0.31	0.47	0.011**
Loss_live_s	0.46	0.50	0.43	0.49	0.649
	168		83		

* significant at 10%; ** significant at 5%, *** significant at 1 %

Table 7.3	Estimation of propensity score: probit model	

Dependent Variable : PSNP (1/0)			
Variable name	Coefficient	Standard Error	P- Value
Hhh_sex	-0.246	0.292	0.933
Age_hh_1	-0.014	0.008	0.075*
Edu_hhh	0.045	0.049	0.355
Family_size	0.198	0.108	0.066*
Dep_ratio	-0.673	0.927	0.469
N_dep	-0.186	0.195	0.339
P_size _tsimdi	-0.073	0.378	0.053*
Livestock in Tlu	-0.030	0.046	0.511
Private_forest	0.518	0.216	0.016**
Fooda_months	-0.056	0.029	0.065**
Access_ irrigation (1/0)	-0.874	0.276	0.002***
M_hh_ext_pack	-0.043	0.244	0.862
Transfer (1/0)	-0.442	0.196	0.024**
T_asset	0.001	0.001	0.774
I_per capita	0.001	0.001	0.774
Dummy_own _b	0.180	0.215	0.399
fol	0.134	0.186	0.473
Weather_s	0.790	0.269	0.003***
Death_ill_s	-0.526	0.239	0.027**
Loss_live_s	0.058	0.206	0.777
constant	0.309	0.696	0.657
Number of observations	251		
Pseudo R^2	0.18		
LR-chi-square	58.52***		
Log likelihood	-130.03795		

Dependent Variable : PSNP (1/0)

* significant at 10%; ** significant at 5%, *** significant at 1%

Table 7.4 Propensity score and covariate balances

		Ν	Iean	% red	uction	t- te	est
Variable	Sample	Treated	Control	% bias	bias	t	p > t
Propensity Score	Unmatched	0.7441	0.5178	108.5		8.42	0.000
	Matched	0.7353	0.7287	3.1	97.1	0.33	0.740
Hhh_sex	Unmatched	0.8571	0.8313	7.1		0.54	0.593
	Matched	0.8519	0.8508	0.3	96.1	0.03	0.980
Age_hh_1	Unmatched	45.167	50.000	-38		-2.92	0.004
	Matched	45.29	44.712	4.5	88	0.42	0.672
Edu_hhh	Unmatched	1.2143	1.0723	6.7		0.48	0.629
	Matched	1.2284	1.4234	-9.2	-37.3	-0.81	0.417
Family_size	Unmatched	5.881	5.4699	19.3		1.45	0.148
	Matched	5.8148	5.6328	8.6	55.7	0.78	0.434
Dep_ratio	Unmatched	0.4888	0.4613	12.9		1.02	0.310
	Matched	0.4863	0.4812	2.4	81.2	0.23	0.816
N_dep	Unmatched	2.9881	2.6627	20.1		1.53	0.127
	Matched	2.9506	2.8276	7.6	62.2	0.69	0.490
P_size _tsimdi	Unmatched	4.1971	4.6778	-16.3		-1.22	0.222
	Matched	4.2603	4.6277	-12.3	24.2	-1.06	0.288
Tlu	Unmatched	2.9695	3.3859	-15.5		-1.20	0.231
	Matched	2.9978	2.9916	0.5	98.5	0.02	0.981
Private_forest	Unmatched	0.4821	0.3855	19.5		1.45	0.149
	Matched	0.4691	0.4457	4.7	75.7	0.42	0.673
Fooda_months	Unmatched	5.2976	6.5181	-35.4		-2.69	0.008
	Matched	5.4136	5.7345	-9.3	73.7	-0.88	0.378
Access_irrigation	Unmatched	0.7738	0.2530	-48.4		-3.93	0.000
	Matched	0.0803	0.0593	5.8	88.1	0.74	0.460

(Note: Figures in bold are significant variables)

M_hh_ext_pack	Unmatched	0.1548	0.1928	-10.0		-0.76	0.450
	Matched	0.1605	0.2178	-15.1	-50.7	-1.32	0.189
Transfer	Unmatched	0.2500	0.4217	-36.8		-2.81	0.005
	Matched	0.2593	0.2386	4.4	87.9	0.43	0.668
T_asset	Unmatched	1322.6	1301.4	0.9		0.08	0.940
	Matched	1328	1295	1.4	-58.0	0.13	0.895
I_per capita	Unmatched	2058.9	2512.7	-27.8		-2.23	0.026
	Matched	2061.7	2254.5	-11.8	57.5	-1.21	0.228
Dummy_own_b	Unmatched	0.3393	0.2892	10.6		0.80	0.426
	Matched	0.3272	0.3408	-2.9	72.9	-0.26	0.796
fol	Unmatched	0.5238	0.4578	13.2		0.98	0.327
	Matched	0.5062	0.4901	3.2	75.7	0.29	0.774
weather_shock	Unmatched	0.8393	0.7108	31.0		2.40	0.017
	Matched	0.8333	0.7811	12.6	59.4	1.19	0.235
Death_ill_shock	Unmatched	0.1726	0.3133	-33.1		-2.56	0.011
	Matched	0.1790	0.1790	0.00	100	0.00	0.999
Loss_livestock	Unmatched	0.4643	0.4337	6.1		0.46	0.649
	Matched	0.4630	0.4623	0.1	98.0	0.01	0.991

Indicator	Sample	
Pseudo R2	Unmatched	0.18
	Matched	0.04
LR X2 (p-value)	Unmatched	59.35(0.000)***
	Matched	18.35 (0.685

 Table 7. 5 Other covariate balance indicators before and after matching

* significant at 10%; ** significant at 5%, *** significant at 1 %

Outcome	PS	М		Critical	,	Treated	Untreat	ted
Indicators	ATT	ATU	ATE	(hidden bias)	On support	Off- support	On support	Off- support
Forest_ext	-0.044**	0.108	0.020	1.4	150	18	3	80
	(-1.94)	(2.17)	(2.43)					
Forest_dep	-0.118***	-0.006	-0.081**	1.7	150	18	3	80
	(-2.41)	(-0.13	(-1.69)					
Forest_trip	-8.410*	0.414	-0.341	1.5	150	18	3	80
	(-2.75)	(0.15)	(-0.21)					

Table 7. 6 Treatment effect and sensitivity analysis

* significant at 10%; ** significant at 5%, *** significant at 1 %

Figure 7. 1 Distribution of propensity scores



Appendix

 Table 71ADefinition of Variables

Variable name	Definition
Treatment Variable	
PSNP (1/0)	Equals 1 if the household is participating in the Productive Safety Net Program (PSNP), 0 otherwise
Outcome Variables	
Forest_extraction (1/0)	Dummy =1 if the household participated in forest environmental resource extraction, 0 otherwise
Forest_dep	Ratio of income from forest environmental resources to the overall household income
Ftrips_year	Number of trips to forests by the household during the survey year.
Explanatory variables	
Sex _hhh (1/0)	Equals 1 if the household head is male, 0 otherwise
Age_hh_1	Household head' age in years
Edu_hhh	Educational level of the household head in years
Family_size	Family size of the household
Dep_ratio	Dependency ratio in the household
N-dep	Number of Dependents in the household
P_size _tsimdi	Plot size owned by the household in tsimdi (= 0.25 hectare)
Tlu	Number of livestock owned in Tropical Livestock Unit (TLU)
Private_forest (1/0)	Dummy = 1 if the household owned private forest, 0 otherwise
Fooda_months	Number of months in the year that the household had enough food stock
Access_irrigation (1/0)	Dummy = 1 if the household had access to irrigation, 0 otherwise
M_hh_ext_pack	Dummy =1 if the household is a beneficiary of the household extension package during the survey year, 0 otherwise
Transfer (1/0)	Dummy =1 if the household had access to transfer, 0 otherwise
T_asset	Total value of assets owned by the household in Ethiopian Birr

I_per capita	Household income per adult equivalent in Ethiopia Birr
Dummy_own _b	Dummy = 1 if the household has his/her own business, 0 otherwise
fol	Dummy = 1 if the household is a member of any social group institution , 0 otherwise
Weather_s	Dummy = 1 if the household experienced weather related shocks last year , 0 otherwise
Death_ill_s	Dummy $=1$ if the household experienced death or illness of a household member last year, 0 otherwise
Loss_live_s	Dummy = 1 if the household experienced shocks related to loss of livestock last year, 0 otherwise

CHAPTER EIGHT

DISCUSSIONS AND CONCLUSIONS

8.1. Forest management, Shocks and Poverty reduction

Poor people often depend on biodiversityboth for their livelihoods and as a safety net against deeper poverty (IIED, 2010). Moreover, direct dependence on services from ecosystems is highest amongst people living in arid and semi-arid lands such as Tigray, where alternative livelihood options areoften limited and environments are particularly fragile and risky (MA 2005b cited in Shackleton et al., 2008). The resultant increased pressure often has a negative feedback on the capacity of the ecosystems to deliver these services and create a downward spiral of increasing poverty and ecosystem degradation (Shackleton et al., 2008) and this dependence of the poor on the fragile open access may confirm the suggestion in some recent literature (e.g. Angelsen and Wunder, 2003; Lavange, 2005; Delacote, 2007) of a resource based "poverty trap."

Accordingly, finding synergistic solutions has been on local, national, and international agendas for decades, in the hope that humandevelopment and biodiversity conservation can be found to be less a zero-sum game of trade-offs and more a set of mutually reinforcing goals (Timmer, 2005). In the same way Howard and Smith (2006) identified two over arching problems of Common Pool Resources Management (CPR) in Tigray: the possibility to halt degradation and sustainably regenerate vegetative resources, and the need to generate livelihood resources that can help to alleviate poverty, food insecurity and continued degradation in the area (Howard and Smith, 2006).

In line with the national development strategy framework known as Agricultural Development Led Industrialization (ADLI), the regional government of Tigray followed a Conservation based Agricultural Development Strategy (CBAD) since 1994. One of the major strategies of environmental rehabilitation includes area enclosures and establishment of community woodlots for ecological restoration. The immediate positive environmental effects of the enclosures and the species restrictions are universally acknowledged by researchers, local populations(Howard and Smith, 2006) and the international community as well. However, there is also a steady growing body of literature questioning the impact of such policy interventions in terms of economic benefits to farm households who are dependent on the forest resources for insurance and source of income, suggesting the gains in the conservation of forest biodiversity in the region require a corresponding focus on the economic and social issues of the intervention. To this end clarity is needed under which conditions the current forest management contributes to the livelihood of farm households in the region.

This study has addressed five major issues of relevance to attaining win –win solutions for human development and conservation of forest biodiversity in the region. *The first issue examined how forest environmental resources provide a "natural insurance" for households affected by idiosyncratic health related shocks*. It examines the difference in the number of trips to forests, forest sales and forest dependency between households that are affected by health related shocks and that were not, and the factors that explain households' participation in forest environmental resources for coping with health related shocks.

Refocusing on the role that forest plays in livelihoods and poverty reduction, Wunder (2001) and Angelsen and Wunder (2003) also argued that optimism about a win-win development and conservation outcome is unwarranted and that NTFP extraction is generally a low-return activity that may even lead to a "poverty trap". Thus, *the second issue addressed in this research deals with forest –food security linkages and its implication for the long term livelihood outcomes of*

smallholder farming households in Tigray. It tries to explicitly focus on food security indicators as outcome variables to investigate the role of forest environmental resources on these indicators. In addition, this section also tries to identify whether a risk management strategy or diversification strategy explain households' choice of participation in forest environmental resource extraction in rural Tigray.

Full understanding of the links between forests and welfare of households requires the ability to make causal inferences about a counterfactual (CBD, 2010). *The third issue of this study is therefore to investigate the welfare impact of forest environmental resource commercialization in rural Tigray using counterfactual income analysis*. It tries to identify conditional expectations, treatment and heterogeneity effects of participation in forest environmental resource commercialization and its implication on the welfare of rural households in Tigray.

Poverty in local forest communities is considered both a cause and result of deforestation and environmental degradation. If poverty can be alleviated through the harvesting of forest environmental resources, then deforestation pressures will be reduced. It is also claimed that the poorer the country, the more significant is natural capital likely to be in determining the overall distribution of wealth. Thus, *the fourth issue that this study addressed is the contribution of forest environmental resources for poverty reduction and distribution of income in rural Tigray.*

Finally, as mentioned in section 8.1., the two overarching problems identified by researchers with the issue of Common Pool Resources (CPRs) in Tigray are the possibility to halt degradation and sustainably regenerate vegetative resources on the one hand, and the need to generate livelihood resources that can help to alleviate poverty, food insecurity and continued degradation on the area on the other (Howard and Smith, 2006). Thus, *the final issue addressed in this research is whether the Productive Safety Net Program (PSNP) in the region provides an appropriate strategy*

for reconciling conservation of the environment and poverty reduction. It identified some of the major problems in the program and policy inputs required to strengthen the role of the program to bring a win-win solution for conservation and poverty reduction.

Discussion on the findings of each research issue and conclusions are presented in subsequent sections. Section 8.2 presents the findings and conclusions on the role of natural insurance for coping with idiosyncratic health related shocks. Section 8.3 presents the main findings and conclusions of forests-food security linkages and their impact on the long term livelihood outcomes of smallholder farming in Tigray. Section 8.4 presents the findings and conclusions on the welfare impact of forest environmental resources along with the conditional expectations, treatment and heterogeneity effects of participation in rural Tigray. Section 8.5 presents a discussion and conclusions on the contribution of forest environmental resources on poverty reduction and income distribution under the existing management system in rural Tigray. Section 8.6 presents a discussion of the findings and conclusions on the role of the productive safety net program in providing win-win environmental conservation and poverty reduction. Finally, in section 8.7 the main policy implications of the overall research and further research needs are discussed.

8.1.1. Idiosyncratic Health Shocks and Forest Environmental Resources

A typical household in rural areas of developing countries is exposed to covariate and/or idiosyncratic risks. Low and volatile incomes coupled with the absence of or poor development of financial and risk sharing institutions in such areas make consumption smoothing an important issue in both theoretical and empirical studies in development economics. In the absence of functioning financial and insurance markets households use different coping mechanisms when

exposed to shocks (Dercon et al, 2005; Dercon & Krishnan, 2005). One of the emergent literatures in relation to consumption smoothing is the issue of "natural insurance" (Angelsen et al., 2007).

However, much of the existing literature on natural insurance deals with covariate shocks such as drought and flooding. Meanwhile, a growing body of empirical evidence suggests that idiosyncratic risk may be as important, or indeed may dominate covariate risks in rural Africa and Asia (Udry, 1990; Townsend, 1995; Deaton, 1997; Libbert et al., 2004; Morduch , 2004; Dercon, 2005; Kaziange and Udry, 2006). In this research we investigated the role of "natural insurance" for coping with idiosyncratic health shocks in rural Tigray, both in the context of the region and in the light of the scholarly debate on the subject.

Controlling for other factors, the empirical results revealed that participation in forest environmental resource extraction and number of trips to forests is positively and significantly associated with health related shocks in Tigray. In the same way, households who are affected by health related shocks are more dependent on forest environmental income. This suggests that forest environmental resources play an important role as a safety net or provide "natural insurance" for the poor households affected by idiosyncratic health related shocks.

8.1.2. Forest-Food Security Linkages

Since the last two decades NTFP-based development was born as a new development paradigm capable of accommodating many conflicting needs - of local livelihoods and of global markets; of balancing regional developmental aspirations with that of national growth; and above all of reconciling environment and development. However, refocusing of the development agenda on poverty has led to recent reassessment of the role that bio-diversity plays in livelihoods and poverty alleviation. A profusion of new commentary has emerged. This poses many fresh questions, and, to some extent, tempers previous optimism regarding the ability of this sector to make a difference

205

by providing a more subtle and complex picture of livelihood – biodiversity linkages (Shackleton, 2006). For example, Neumann and Hirch (2000) present evidence from an array of studies that show that NTFP extraction is an activity of the poor, and Wunder (2001) and Angelsen and Wunder (2003) argue that optimism about a win – win development and conservation outcome is unwarranted, and that NTFP extraction is generally a low-income activity that may even result in a poverty trap. Similarly, Lavange et al., (2005 cited in CBD, 2010) claim that biodiversity resource dependence is a symptom of poverty, and it is only by "leaving the forest" that the poor can hope to escape poverty. A plausible alternative to the downward spiralscenarios is a pattern in which poor rural households convert natural capital into physical and human capital, and use these investments to gradually move out of poverty and reduce their reliance on destructive resource extraction (Angelsen et al., 2007).

The general conclusion in the current literature is that the safety net and poverty trap aspects of NTFPs are linked, in as much as features that make forest products attractive to the poor also limit their potential for generating higher income and escaping their poverty (Shackleton, 2006).

However, much of the existing arguments and conclusions made on the link between poverty and biodiversity suffer from an overload of conjectural and anecdotal assertion rather than evidence (Roy et al., 2010). Full understanding of the links between biodiversity and poverty require the ability to make causal inferences about the counterfactual (CBD, 2010; Vira and Kontoleon, 2010) – and none of the previous studies in the region do this. In a wide ranging review on the research done on the link between poverty and biodiversity, CBD, (2010); Vira and Kontoleon, (2010) also argued that if nature's resources help to temporarily smooth consumption and incomes, their impacts may be better captured through an explicit focus on this temporality as part of our poverty

measure, instead of annualized income or consumption (in which these temporary contributions do not always feature as significant).

In this research we investigated the role of forest environmental resources in food security outcomes of the households in rural Tigray and its implication for long term livelihoods. Three food security outcomes were considered in the analysis: (1) distress sale of assets for immediate consumption, (2) number of months that the household had enough food stock available during the survey year, and (3) total expenditure per adult household equivalent.

The empirical findings show that controlling for other factors, higher educational level and age of the household head, access to off farm employment, and awareness of climate change in the locality decrease the probability of participation in forest environmental resource extraction, while vulnerability to shocks and gender of the household head being male increase the probability of participation in forest environmental resources. Second, we find that participants in forest environmental resource extraction in rural Tigray have significantly lower food security outcomes. Third, we find that the poor and non-poor households that are participating in forest extraction had lower distress sale of assets for immediate consumption than non-participants in the same sample, enforcing the widely held view that forests are important as an economic buffer in hard times. Fourth, households in rural Tigray are not driven into forest extraction by risks in farm output only but also by diversification strategy, suggesting that the problem for local communities has both the characteristics of portfolio analysis and the economics of insurance.

Finally our findings support the widely held view that the poor appear to be linked with naturebased resource use, but these may serve to perpetuate poverty and food insecurity (CBD, 2010; Vira and Kontoleon, 2010).

207

8.1.3. Forest commercialization and Household income

In a wide ranging review, Roe et al. (2010) warned that case studies that reach conclusions about poverty-conservation mechanisms without reference to a counterfactual have to be treated cautiously because income (or any other welfare related measures) may have been affected by other confounding effects. In this section we investigate the contribution of forest environmental resource commercialization on the welfare of rural households in Tigray. We estimated a simultaneous equation model with endogenous switching to account for the heterogeneity in the decision to participate or not, and for unobservable household characteristics.

Our empirical results show that education of the household head, access to wage income and own business enterprise, and higher size of land ownership were found to be negatively correlated to forest related activities, while higher distance to market, vulnerability to shocks and household head being male were found to be positively correlated. Second, the estimated heterogeneity effect revealed that independent of participation, participant households would have less income on average than non-participants, implying the group of farm households that actually participated in forest environmental resource commercialization have systematically different characteristics than the group that did not participate. Third, the treatment effect analysis shows that participate, while it improved the income of households who actually participated. This suggests that participation in forest related activities in rural Tigray seems to be particularly important for the group of farm households most vulnerable to shocks.

8.1.4. Forests income poverty and inequality

Despite a wealth of case studies on the link between biodiversity conservation and poverty reduction, the vast majority of the existing body of work does not use analytical and empirical

methods required to make reliable inferences about the actual impact of conservation initiatives on measurable poverty indicators (CBD, 2010). To appropriately answer these types of impact assessment questions, there is a need for greater use of counterfactual cases. Before and after assessments of conservation initiatives are not sufficient (Roe et al., 2010). In this section we estimated the poverty and inequality indicators using data from rural Tigray taking into account the counterfactual situation. The poverty and inequality indicators were also estimated using the traditional "conventional" approach in order to allow us a clear estimation of the extent of overestimation and/or underestimation of the impacts of forest environmental resources on poverty and inequality in previous works.

The empirical results show that the choice of methodology can have a significant impact on the estimated contribution of forests and conclusions made about the link between forest biodiversity and poverty reduction. Second, we find that the contribution of forests to reduce the incidence of poverty is considerably weaker, and might increase the severity of poverty when the counterfactual estimation method is used.

Third, in relation to income distribution, we find that after using the counterfactual approach, the impact of forest income on distribution is negative as against a positive contribution in the conventional approach. Fourth, these apparently contradictory findings, where forest income appears to reduce substantially the poverty and inequality indicators when we use the conventional approach, but has little or negative impact on poverty and inequality indicators under the counterfactual approach, could be explained by the fact that participation in forest activities might have a disincentive effect on incomes from other activities which are not captured when forest income is treated as an exogenous addition to the existing household income.

8.1.5. PSNP and forest pressure

The findings of our previous sections revealed that forest environmental resources play an important role for households vulnerable to idiosyncratic and covariate shocks. Findings also show that it is the poor and vulnerable households in rural Tigray who are most dependent on forest environmental resources and forest environmental resources provide food security, enforcing the widely held view that forests provide a buffer during economic misfortunes for the poor. Meanwhile, our discussion in section 8.1.2 also warned that dependence of households on forest environmental resources as a" natural insurance" may serve to perpetuate poverty. This implies that forest environmental resources have both the advantage of offering the poor households an activity to survive, with the disadvantage of keeping them in poverty (Delacote, 2007). This suggests the potential existence of risk induced poverty traps, whereby those who can insure their consumption against income shocks can take advantage of the more profitable opportunities and possibly grow out of poverty, while others are stuck with low return, low risk activities, trapping them into poverty, even though their inherent risk preferences may fundamentally be the same (Dercon, 2007). Indeed common resource extraction may constitute a poverty trap as a result of a tragedy of the commons. Too many households are in need of insurance and the resource cannot provide enough to properly insure all the population. They face thus the classic poverty – environment nexus, where poor people depend too much on their environment, and over use it (Delacote 2007).

An implication of the above cases and our findings from the previous chapters is that current consumption of households in rural Tigray is maintained through actions that seriously, sometimes irreversibly, compromise future livelihoods, actions that could have been avoided if households were able to use other alternatives to anchor consumption to average income. In this paper we investigated whether social protection programmes such as the Productive Safety Net Programme (PSNP) in Ethiopia could provide a win –win for poverty reduction and conservation of forests in rural Tigray by reducing the pressure on forests.

The empirical findings revealed that the PSNP decreases the pressure on the forest ecosystem by decreasing trips to forest environmental resource collection and forest dependency. The findings also show that those households that have a higher probability of participating in PSNP were able to reduce their dependency on forests over and above the households that have less probability of participating in PSNP. The implication of this result is thus, the gains in terms of reducing forest dependency as a result of PSNP are higher for households with higher probability of participating in PSNP than those of households with lower chances of involving in the program. These findings are generally consistent with the widely held view that programs designed to reduce economic vulnerability of low income households can improve human welfare and reduce forest pressure, i.e. PSNP could help in promoting a win-win for poverty reduction and forest ecosystem conservation.

8.2. Policy Implications and Further Research

8.2.1. Policy Implications

The issues addressed in this thesis are interrelated and can be seen within the context of bringing overall sustainable development in Tigray. The very nature of sustainable development emphasises the integration of its pillars – economic, social and environmental – and this implies a need not just to focus on one of its pillars to achieve sustainable development but, for example to consider how environment – and natural resource management – can be integrated across the other pillars of economics and society which are dependent on it for health, food security, and income (and

equally to consider how progress towards the other pillars such as human health, poverty reduction and equity might impact on environmental sustainability).

One important policy implication of this study is that policy makers need to simultaneously address the issue of human and environmental health. In this regard, our specific policy proposal to safeguard the forest resource-related needs of households affected by illness in a particular setting could include those that: ensure local values for forest resources (such as medicinal plants) are protected by forest management plans (Colfer et al., 2006) ; promote agro forestry as a means to assisting rural residents to establish and manage tree crops, particularly those possessing medicinal properties (Barany et al., 2000); and examine rural health delivery to determine the environmental consequences of access, or lack of thereof, to health services. Moreover, institutional support through access to credit and creating awareness to climate change could also play a role in minimizing the pressure on forests.

Second, while it is agreed that forest resource conservation is critical, how it happens, what is conserved, and for whom, requires a complex set of trade-offs that the current conservation intervention could not achieve. Ensuring that the forest resources contribute to poverty reduction rather than exacerbating poverty implies a need to adopt different approaches to resource conservation that provide benefits for the poor people and to meet social justice objectives. The existing intervention does have the potential to contribute a win-win for conservation and poverty reduction, but provided that thorough impact assessment is undertaken with full participation of indigenous people and local communities to identify potential negative impacts, provision is made for full and fair compensation or mitigation where appropriate, mechanisms for including local values (based on utility) and global values are introduced in determining conservation priorities; and equitable sharing of rights, responsibilities, costs and benefits between all stakeholders that

212

requires balancing customary and formal norms of institutions, and recognizing historic tenure rights. Over all, government, policy-makers and natural resource managers need to acknowledge the livelihood safety-net role that forests play in rural livelihoods and recognize that environmental protection policies limiting or banning access and use of forest resources can deepen rural poverty, as the poor suffer more from the deprivation of these resources. Therefore, there is a need to invest in the development and promotion of sustainable use forest management practices that allow the poor to utilize forest environmental resources to enhance their economic wellbeing with minimum adverse effects on forest ecological conditions.

Finally, the policy implication that can be drawn from the study result is the importance of and the promotion of public safety nets such as the Productive Safety Net Program (PSNP) for the poor and the promotion of government programmes and policies that increase the productivity of agricultural production and support diversification into off-farm livelihood and income sources to provide positive incentives for forest conservation and sustainable use. This suggests that sustainable forest management has to be integrated with the broader framework of rural development programs aimed at reducing poverty in order to provide the necessary incentives for the poor to adapt sustainable resource management options.

8.2.2. Further Research

Consideration needs to be given to dynamic elements and nature of the issues addressed in this study. For instance, the issue of forests as safety nets or poverty traps extends beyond one year. Poverty trap is a dynamic issue that needs longitudinal data and the poverty trap implications in the cross sectional analyses used here do not necessary imply a poverty trap in the long run.

Moreover, our conclusion on the positive role of the PSNP in pursuing the twin goals of sustainable development i.e. raising human living standards and conservation of forest biodiversity, also needs

213

further research using longitudinal data, because households are likely to respond differently to income changes from poverty reduction programs that are perceived to be substantial and permanent versus small and temporary (Alix- Garcia et al, 2010).

BIBLIOGRAPHY

- ABCG (Africa Biodiversity Collaborative Group) (2002) 'HIV/AIDS and Natural Resource Management Linkages', workshop proceedings, Nairobi, Kenya
- Adbdulai, A. and Delgada, C.(1990). Determinants of nonfarm earnings of farm-based households and wives in Northern Ghana. American Journal of Agricultural Economics 81,117-130
- Abebe Haile Gebremariam, Million Bekele and Ridgewell, A. (2009) *Small and medium forest enterprises in Ethiopia*.IIED Small and Medium Forest Enterprise Series No. 26. FARM-Africa and International Institute for Environment and Development, London, UK
- ADB (2010). Ethiopia's Economic growth Performance: Current Situation and Challenges. Economic Brief, Volume1, Issue 5 17 September, 2010
- Adhikari, B (2005). Poverty, Property Rights and Collective action: Understanding the Distributive Aspects of Common Property Resource Management. *Environment and Development Economics*, 10:7-31.
- Adhikari, B., Di Falco, S., Lovett, J.C., 2004. Household characteristics and forest dependency: evidence from common property forest management in Nepal. Ecol. Econ. 48, 245–257.
- Adhikari, B. (2003). Property rights and natural resources: Socio-economic heterogeneity and distributional implications of common property resource management, *Working Paper 1-03*, South Asian Network for Development and Environmental Economics (Sandee), Kathmandu, Nepal
- Agarwal, B. (1991). Social security and the family: coping with seasonality and calamity in rural India, pages 171–244. Social Security in Developing Countries, Ahmad, E.J. and Dreze, J. and Hills, J. and Sen, A. (eds.), Clarendon Press, Oxford.
- Akkinnifesi, Festus & Waibel, Herman& Mithofer, Dagmar, 2006." The role of food from natural resources in reducing Vulnerability to poverty: a case study from Zimbabwe," Proceedings of the German Development Economics conference, Berlin 2006 28, Verein fur Socialpolitik, Research Committee Development Economics.
- Akin, JS, CC Griffin, DK Guildy, and BM Popkin. 1985. The demand for primary health services in the third world. Rowman & Allanheld Publishers: New Jersey.
- Aldermann, H. and Paxson, C. (1994). Do the poor insure? A synthesis of the literature on risk and consumption in developing countries, volume 4.Economics in a changing world,London, Mcmillan edition.
- Alix-Garcia, Jennifer, McIntosh, Craig, Welch, Jarrod R. and Sims, Katharine R. E., The Ecological Footprint of Poverty Alleviation: Evidence from Mexico's Oportunidades Program (March 10, 2010). Available at SSRN: http://ssrn.com/abstract=1568245 or http://dx.doi.org/10.2139/ssrn.1568245
- Andersson ,C. (2010). Can a Social Safety Net Affect Farmers' Crop Portfolios? A Study of the Productive Safety Net Programme in Ethiopia. Department of Economics, Umeå Economic Studies 01/2010, Sweden
- Andersson ,C. ,Mokennen A., and Stage.J (2011). Impacts of the ProductiveSafetyNetProgram in Ethiopia on livestock and treeholdings of rural households. Journal of Development Economics, <u>Volume 94, Issue 1</u>, January 2011, Pages 119–126
- Anderson .J, Benjamin.C, Campbell.B, and Tiveau.D . (2006). Forests, poverty and equity in Africa: new perspectives on policy and practice. International Forestry Review vol.8 (1), 2006
- Angelsen Arild, Fisher Monica, Jumbe Charles, Shively Gerald and Sserunkuuma Dick (2008). Using Natural Capital to Mitigate Risk and Reduce Poverty. <u>http://crsps.net/wp-</u> <u>content/downloads/BASIS/Inventoried%202.17/13-2008-2-22.pdf</u>
- Angelsen.A, Fisher.M, Jumbe. C, Shively. G, Sserunkuuma. D (2007). Natural Capital and Poverty Reduction, AProposaltotheAMABASISCRSP,28August2007,http://docs.mak.ac.ug/sites/default/files/Nature%20Capital%20and%20Poverty%20Reduction.pdf
- Angelsen, A, Fisher.M, Jumbe.C, Shively .G and Sserunkuuma,(2008). Using Natural Capital to Manage Risk and Reduce Poverty, BASIS BRIEF, Assets and Market Access, CRSP, February 2008
- Angelsen, A., Sunderlin, W., Ahmad Dermawan, D., and Rianto, E. (2001). Economic crisis, small farmer wellbeing, and forest cover change in Indonesia. World Development, 29(5).
- Angelsen A., and Wunder, S. (2003). Exploring the forest-poverty link: key concepts, issues and research implications'. CIFOR Occasional Paper No. 40, Bogor, Indonesia: Center for International Forestry Research
- Arnold, J.E. Michael (2001a). Forestry, Poverty and Aid. CIFOR Occasional Paper No.33. Bogor, Indonesia: Centre for International Forestry Research.
- Arnorld, J.E.M and Riuz –Perez, M. (2001b). Can non timber forest products match tropical forest conservation and development objectives? Ecological Economics 39: 437-447
- Aryal, B., 2002. Are Trees For The Poor? A Study from Budogo Forest, Uganda.Agricultural University of Norway, Department of Economics and Social Science, Ås. MSc Thesis.
- Asfaw, S and Shiferaw. B (2010). Agricultural Technology Adoption and Rural Poverty: Application of an Endogenous Swtiching Regression for Selected East African Countries. Poster Presented at the joint 3rd African Association of Agricultural Economists (AAAE) and 48th Agricultural Economists Association of South Africa (AEASA) Conference, Cape Town, South Africa, September 19-23, 2010
- Au, D.W.H., Crossley, T.F.& Schellhorn, M.(2005), "The effect of health changes and long-term health on work activity of older Canadians', Health Economics 14(10),999-1018
- Babulo, B. (2007). Economic valuation and management of common-pool resources: the case of enclosures in the highlands of Tigray, Northern Ethiopia, Unpublished PhD thesis, Katholieke Universiteit Leuven, 2007

- Babulo. B, Muys.B, Nega.F, Tollens. E, Nyssen. J, Deckers.J and Mathiji.E(2009). The Economic Contribution of Forest resource use to rural livelihoods in Tigray, Northern Ethiopia. Forest Policy and Economics 11(2009) 109-117.
- Baland, J. and Francois, P. (2005). Commons as insurance and the welfare impact of privatization. Journal of Public Economics, 89(2-3):211–231.
- Baland, J-M.and Platteau, J. P. (1996). Halting Degradation of Natural Resources. Is there a Role for Rural Communities?' Oxford: Clarendon Press.
- Barany, M., Hammett, A.L., Sene, A., and Amichev, B. (2001) 'Nontimber Forest Benefits and HIV/AIDS in Sub-Saharan Africa', *Journal of Forestry* 99: 26-41
- Barany, M., Holding Anyonge, C., Kayambazinthu, D. and Sitoe, A. (2005) 'Firewood, food, and medicine: Interactions between forests, vulnerability and rural responses to HIV/AIDS', Proceedings from the IFPRI Conference: HIV/AIDS and Food and Nutrition Security, Durban, South Africa
- Barany, M., Hammett, A.L., Stadler, K.E., and Kengni, E. (2004) 'Non-timber forest products in the food security and nutrition of smallholders afflicted by HIV/AIDS in Sub-Saharan Africa', *Forests, Trees and Livelihoods* 14: 3-18
- Barrett, C.B., Arcese, P., 1998. Wildlife harvest in integrated conservation and development projects: linking harvest to household demand, agricultural production, and environmental shocks in the Serengeti. Land Economics 74, 449–465.
- Barrett, C.B., Marenya, P.P., Mcpeak, J., Minten, B., Murithi, F., Oluoch-Kosura, W., Place, F., Randrianarisoa, J.C., Rasambainarivo, J. and Wangila, J. (2006) Welfare dynamics in rural Kenya and Madagascar. *Journal* of Development Studies 42(2), pp. 248-277.
- Becker, G.S., 1965. A theory of the allocation of time. Economic Journal 75 (299), 493–517.
- Becker S. and Caliendo M. (2007), "Sensitivity analysis for average treatment effects", The Stata Journal, Vol.7, 1, 71-83.
- Becker, S. O. and Ichino, A. (2002). Estimation of average treatment effects based on propensity scores. *Stata Journal* 2: 358–377.
- Becker.O.Sascha & Ichino Andrea(2002).Estimation of average treatment effects based on propensity scores. The Stata Journal(2002),2, Number 4,pp.358-377
- Belcher ,B. and Schrenberg, K.(2007). Commercialization of non-timber forest products: a reality check. Development Policy Review, 25,335-377
- Ben-Houassa,O.Kouame(2011). Adoption and Levels of Demand for Fertilizer in Cocoa Farming in Cote D'ivore: Does Rsik Aversion Matters? Paper Presented

- Berhanu, Mulugeta. 2004. Institutions and collective action in community property resource management: the case of community woodlots in Hawzien, Tigray Region. Masters research paper, Institute of Social Studies, The Hague.
- Besley, T. (1995). 'Nonmarket Institutions for Credit and Risk Sharing in Low-Income Countries', *Journal of Economic Perspectives*, vol. 9(summer), pp. 115–27.
- BOFED (1995). First Five Years Development Plan of the Tigray, Region, Bureau of Planning and Economic Development, Mekelle, Ethiopia (unpublished document).
- Botha, N.B., 2003. Looking for A Path Out of Poverty: The Study of Chimaliro Forestry Reserve, Malawi. Agricultural University of Norway, Department of Economics and Social Science, Ås. MSc Thesis
- ByronNeil & Arnold Michael (1999). What Futures for the People of the Tropical Forests? World Development Vol. 27, No. 5, pp. 789±805, 1999
- Campbell, B., Frost, P., & Byron, N. (1996) Miombo woodlands and their use: Overview and key issues. In: *The miombo in transition: Woodlands and welfare in Africa*, ed. B. Campbell, pp. 1-10. Bogor, Indonesia: Center for International Forestry Research.
- CavendishWilliam (2000). Empirical Regularities in the Poverty-Environment Relationship of Rural Households: Evidence from Zimbabwe. World Development Vol. 28, No. 11, pp. 1979±2003, 2000
- Cavendish, W. (1999). Environmental resource use and the household accounts: Collection, cleaning, accounting, aggregation and headline results, A background guide for the 1993/94 Dataset for Shindi Ward, Zimbabwe
- CBD (2010) Linking Biodiversity Conservation and Poverty Alleviation: A State of Knowledge Review. CBD Technical Series No. 55, Secretariat of the Convention on Biological Diversity ISBN # 92-9225-281-X.
- CBO (2011). Trends in the Distribution of Household between 1979 and 2007, The Congress of the United States, Congressional Budget Office, October, 2011
- Central Intelligence Agency (2009).*The World Factbook. July,* 2009.<u>https://www.cia.gov/library/publications/the-world-factbook/geos/et.html(</u> Accessed on March 2012)
- Central Statistics Authority (2005). *Ethiopia Demographic and Health Survey*. Addis Ababa, Central Statistics Authority, 2005
- Chisholm, Nick(2004). How Should the Commons be Governed? Household Preferences for CPR Management Regimes in Tigray, northern Ethiopia. Paper presented at the Tenth Biennial Conference of the International Association for the Study of Common Property, Instituto de Investigaciones Sociales, Ciudad Universitaria, Oaxaca, Mexico, 10-13 August 2004.

- Chisholm,Nick(2000). Maintaining a Fragile Balance? Community Management of Renewable Natural Resources in Tigray, NE Ethiopia. Paper submitted to Eighth Biennual Conference of the International Association for the Study of Common Property, Bloomington, Indiana, 31 May-4 June 2000.
- Coady, D., M.Grosh, and J.Hoddinott (2004) "Targeting Outcomes Redux" World Bank Research Observer 19(1): 61-85
- Colfer, C.J.P., Sheil, D., and Kishi, M. (2006) *Forests and Human Health: Assessing the Evidence*, CIFOR Occasional Paper No. 45, Bogor: CIFOR
- Coll-Black, S., Gilligan, D. Hoddinott, J.Kumar, N., Seyoum A., and Wisen, W.(2013). Targeting Food Security Interventions : The Case of Ethiopia's Productive Safety Net Programme. International Food Policy Research Institute, 2013
- Commes, T. Oliver, Barham L. Bradford, and Takasaki Yoshito (2004). Targeting conservation-development initiatives in tropical forests: insights from analyses of rain forest use and economic reliance among Amazonian peasants, Ecological Economics 51 (2004) 47–64
- Costanza, R., Folke, C., 1997.Valuing ecosystem services with efficiency, fairness and sustainability as goals. In: Daily, G. (Ed.), Nature's Services: Societal Dependence on Natural Ecosystems. Island Press, Washington, DC, pp. 49–70.
- Daily, G. (Ed), 1997. Nature's Services: Societal Dependence on Natural Ecosystems. Island Press, Washington, DC, 392 pp
- Das Gupta, M. and Maler, K.(1993). Handbook of development economics, volume 3, chapter Poverty and the Environmental Resource Base.University of Cambridge.
- Deaton, A. (1997). The analysis of household surveys: A micro econometric approach to development policy. Baltimore, Johns Hopkins University Press
- Debela.B, Shively.G, Holden.S (2014). Does Ethiopia's Productive Safety Net Program Improve Child Nutrition? Center for Land Tenure Studies Working Paper 01/14, CLTSNMBU, 2014
- Dehejia, R. und Wahba, S. (1999): Causal effects in non-experimental studies: reevaluation the evaluation of training programs. Journal of the American Statistical Association 94, H. 448, S. 1053-1062.
- Delacote, P. (2007). Agricultural expansion, Forest products as a safety nets, and deforestation. Journal of Environment and Development 12: 235- 249, 2007 Cambridge University Press
- Dercon, S., J. Hoddinott, and T. Woldehanna. 2005. Consumption and shocks in 15 Ethiopian Villages, 1999-2004. *Journal of African Economies* 14: 559-585.
- Dercon, S. and P. Krishnan (2005). 'Food Aid and Informal Insurance', in S. Dercon, (ed.) *Insurance Against Poverty*, Oxford and New York: Oxford University Press, pp. 305-329.
- Dercon,S., & Krishnan, P. (1998). Changes in Poverty in Rural Ethiopia 1989-1995: Measurement, Robustness Tests and Decomposition. Working paper, No. WPS 98/7, centre for the study of African economies, University of Oxford.

- Disney, R., Emmerson, C. & Wakefield ,M. (2006)," ill health & retirement in Britain: A panel data- based analysis" Journal of Health Economics
- Devereux.S and Guenther.B (2009).Agriculture and Social Protection in Ethiopia.FAC working Paper No. SP03, January 2009
- Devereux, S., Guenther, B., 2007. Social protection and agriculture in Ethiopia. Report written for the FAO. Institute of Development Studies, Sussex, UK.
- Devereux, S., Sabates-Wheeler, R., Tefera, M., Taye, H., 2006. Ethiopia's Productive Safety Net Programme (PSNP): trends in PSNP transfers within targeted household. Institute of Development Studies and Indak International Pvt. L.C, Sussex, UK and Addis Ababa, Ethiopia.
- Devereux, S., Amdissa Teshomen and Sabates-Wheeler, R. (2005) 'Too Much Inequality or Too Little? Inequality and Stagnation in Ethiopian Agriculture, IDS Bulletin, 36(2): 121-126
- Devereux. S and Sharp.K (2003). Is Poverty Really Falling in Rural Ethiopia? Paper presented at the conference *Staying Poor: Chronic Poverty and Development Policy*' at the University of Manchester, 7 to 9 April 2003Institute of development Studies at the University of Sussex Brighton BN1 9RE
- Di_Falco.S., Veronesi. M., Yesuf. M.(2011). Does Adaptation To Climate Change Provide Food Security? A Micro-Perspective from Ethiopia. Center for Climate Change Economics and Policy working paper No.22 and Grantham Research Institute on Climate Cchange and the Environment, Working Paper 19, 2010.
- Di_Falco.S., Veronesi. M., Yesuf. M.(2011). Does Adaptation To Climate Change Provide Food Security? A Micro-Perspective from Ethiopia. American Journal of Agricultural Economics, 93(3): 829-846
- Dilys Roe, Matt Walpole and Joanna Elliott (2010) Linking Biodiversity Conservation and Poverty Reduction What, Why and How? Summary report of a symposium held at the Zoological Society of London, 28th – 29th April 2010
- Duraiappah, A. K. (1998). Poverty and environmental degradation: A review and analysis of the nexus, *World Development* **26**(12): 2169–2179.
- d.Uva, T.B., Doorslaer, E.V., Lindeboom, M.& O'Donnel, O. (2008), 'Does reporting hetrogeniety bias the measurement disparities?', Health Economics 17(3) 351- 375
- Ellis, F., 1993. Peasant Economics: Farm Households and Agrarian Development. Cambridge University Press, Cambridge. 309 pp.
- Eriksen, S., Brown, K., & Kelly, P. M. (2005). The dynamics of vulnerability: Locating coping strategies in Kenya and Tanzania. The Geographical Journal, 171(4), 287–305.
- Etile, F. &Milcent, C.(2006), "Income-related reporting heterogeneity in self –assessed health: evidence from France", Health Economics 15(9), 965-981
- Fafchamps, M. and S. Lund (2003), 'Risk-sharing networks in rural Philippines', Journal of Development Economics **71**: 261-287.

- Faltermeier, L. and A. Abdulai (2009). The impact of water conservation and intensification technologies: empirical evidence for rice farmers in Ghana. *Agricultural Economics* 40(3):365-279.
- FAO (2003): Integrated Natural Resource Management to enhance food Security: The case of community based approaches in Ethiopia working paper 16, FAO Rome:Forests, Trees and Food Security
- FDRE (2005) Rural Land Administration and Land Use Proclamation No. 456/2005. Addis Ababa.
- FDRE (2007a) Forest Development Conservation and Utilization Proclamation No. 542/2007. Addis Ababa.
- FDRE (2007b) Policy and Strategy on Forest Development, Conservation and Utilization. Addis Ababa
- Federal HIV/AIDS Prevention & Control Office (FHAPO) (2009).*Monthly ART report as of end of September 2009*. Addis Ababa, Federal HIV/AIDS Prevention & Control Office, 2009.
- Federal Ministry of Health(2008). *Health and health-related indictors* 2005-20066 and 2007-2008. AddisAbaba, Federal Ministry of Health, 2008.
- Feoli E, Vuerich LG, Zerihun W (2002) Evaluation of environmental degradation in northern Ethiopia using GIS to integrate vegetation, geomorphological, erosion and socio-economic factors. Agriculture, Ecosystems and Environment 91:313-325
- Féres., J, Reynaud.A, and Thomas. A (2007). Water reuse in Brazilian manufacturing firm. Working Paper Series No.2007-WP2, Latin American and Caribbean Environmental Economics , Program, 2007
- Fisher M., Chaudhury M. and McCusker B. (2010).Do Forests Help Rural Households Adapt to ClimateVariability? Evidence from Southern Malawi. World Development, <u>Volume 38, Issue 9</u>, September 2010, Pages 1241– 1250
- Fisher, Monica and Shively Gerald (2005). Can Income Programs Reduce Tropical Forest Pressure ? Income Shocks and Foresr Use in Malawi, World Development Vol. 33, No. 7, pp.1115–1128, 2005
- Fisher, Monica (2004). Household welfare and forest dependence in Southern Malawi, *Environment and Development Economics*, Volume 9, Issue 02 / May 2004, pp 135 154
- Fonta W.,Ichoku, E.H. and Ayuk E.(2011). The distributional Impacts of Forest Income on Household Welfare in Rural Nigeria. Journal of Economics and Sustainable Development ISSN 2222-2855(online).
- Foster J., J. Greer, J., & Thorbecke, E. (1984).A class of decomposable poverty measures.*Econometrica*.52, 761-765
- <u>Fredu Nega, Erik Mathijs, Jozef Deckers, Mitiku Haile, Jan Nyssen, Eric Tollens</u> (2010).Rural Poverty Dynamics and Impact of Intervention Programs upon Chronic and Transitory Poverty in Northern Ethiopia.*African Development Review*, Vol. 22, issue 1., pp. 92-114,March 2010
- Fu,Y.N., J.Chen, et al. (2009). « The role of Non-Timber Forest Products during agroecosystem shift in Xishuangbanna, Southwestern China", Forest Policy and Economics 11(1)18:25

- Gebreegziabher Z (2007) Household fuel consumption and resource use in rural-urban Ethiopia. PhD Thesis, Wageningen University, Netherlands
- Gebreegziabher Z, Gebremedhin.B and Mekonnen A (2011) Institutions, sustainable land use and consumer welfare: the case of forest and grazing lands in northern Ethiopia. *Environment and Development Economics* 17: 21–40
- Geberehot.T and van der Veen. A (2010).Effect of policy interventions on food security in Tigray, Northern Ethiopia, ITC Working Papers Series, Paper 6-October 2010
- Gebremedhin. B, Pender. J and Tesfay.G (2000). Community natural resource management: The case of woodlots in Northern Ethiopia: EPTD Discussion paper, No 60, Washington, DC : International Food Policy Research Institute.
- Gebremedhin. B, Pender. J and Tesfay.G (2004). Collective action for grazing land management in crop –livestock mixed system in highlands of northern Ethiopia, Agricultural System 82 (3): 273-290
- Gilligan, D.O., Hoddinott, J., 2007. Is there persistence in the impact of emergency food aid? Evidence on consumption, food security, and assets in rural Ethiopia. American Journal of Agricultural Economics 89 (2), 225–242.
- Gilligan, D.O, Hoddinott, J and Taffesse, AS. (2008): "The impact of Ethiopia's Productive Safety Net Programme and its linkages". *IFPRI Discussion Paper 839*. Washington, DC: International Food Policy Research Institute.
- Gilligan, D. O., Hoddinott, J., Kumar,N.R, Taffesse,A.S,2009. Impact of Social Protection on Food Security and Coping mechanisms: Evidence from Ethiopia's Productive Safety Nets Prorgam. Draft report . International Food Policy Research Institute (IFPRI), March 6, 2009
- Godoy. R, Overman. H, Demmer.J, Apaza.L, Byron.E, Huanca.T, Leonard.W, Pe´rez.E, Reyes-Garcı´a.V, Vadez.V, Wilkie.D, Cubas.A, McSweeney.K, Brokaw. N(2002). Local financial benefits of rain forests: comparative evidence from Amerindian societies in Bolivia and Honduras. Ecological Economics 40 (2002) 397–409
- Godoy. R, O'Neil. K, Groff. S, Kostishack. P, Cubas. A, Demmar. J, McSweeney. K and Overman.J(1997).
 Household Determinants of Deforestation by Amerindians in Honduras, *World Development*, Vol. 25, No. 6, pp. 977-987, 1997
- Gody.A and Bawa.S Kamalit (1993).The Economic Value and Sustainable Harvest of Plants and Animals from the Tropical Forest: Assumptions, Hypothesis and Methods, Economic Botany 47(3) pp. 215-219. 1993
- Gomez, P.G., & Nicolas, A.L.(2006), "Health Shocks, Employment and Income in Spanish labor Market", Health Economics 15 (9), 997-1009
- Gopalakrishnan.C, Wickrmashinghe.W.A.R, Gunatilake.H.M, andIllukpitiya.P (2005).Estimating the demand for non-timber forest products among rural communities: a case study from the Sinharaja Rain Forest region, Sri-lanka. Agroforestry Systems, Volume 65 Number 1(2005), 13-22

- Griffin, K and Ickowitz, A., 1997, The distribution of wealth and the pace ofdevelopment. Working paper series, United Nations Development Programme (UNDP) <Http://www.undp.org/poverty/publications>.
- Guthiga Paul and Mburu John(2006).Local Communties's Incentives for Forest Conservation: Case of Kakamega Forest in Kenya, A Paper Presented at the 11th Biannual Conference of International Association for the Study of Common Property (IASCP) Bali, Indonesia June 19th June 23rd, 2006
- Hagos, Fitsum and Stein Holden (2006). "Tenure security, resource poverty, risk aversion, public programs and household plot level conservation investment in the highlands of northern Ethiopia", Agricultural Economics, 34(2):183-196.
- Hagos, Fitsum, Pender. J and Gebresselasie Nega (1999). Land Degradation in the Highlands of Tigray and Strategies for Sustainable Land Management, Issue 25 of socio economic and policy working paper, International Livestock Research Institute(ILRI), Addis Ababa
- Hardin, G. (1968), 'The tragedy of the commons', Science 162(3859): 1243-1248.
- Heckman, J (1979). Sample Selection Bias as a Specification Error. Econometrica. Vol. 47, No.1, 1979
- Heckman.J.James Ichimura Hidehiko and Todd.E. Petra (1997). Matching as an Econometric Evaluation Estimator: Evidence from Evaluating a Job Training Programme, The Review of Economic Studies, Vol.64, No.4, Special Issue: Evaluation of Training and other Social Programmes (Oct.1997),pp.605-654
- Heckman.J. James, Ichimura Hidehiko and Todd.E. Petra (1998). Matching as an Econometric Evaluation Estimator, *The Review of Economic Studies* Vol. 65, No. 2 (Apr., 1998), pp. 261-294
- Heckman James, Tobias.L. Justin and Vytlacil Edward (2001). Four Parameters of Interest in the Evaluation of Social Programs, *Southern Economic Journal* Vol. 68, No. 2 (Oct., 2001),pp.210-223
- Heckman, J., LaLonde, R. and J. Smith (1999). "The economics and econometrics of active labor market programs". In *Handbook of Labor Economics*, Vol. III, Ed.By O. Ashenfelter, and D. Card, pp. 1865-2097. Elsevier, Amsterdam.
- Heckman, J.J., Navarro-Lozano, S., 2004. Using matching, instrumental variables, and control functions to estimate economic choice models. Review of Economics and Statistics 86: 30-57.
- Hettige, Hamamala (2006), "When Do Rural Roads Benefit the Poor and How?" Asian Development Bank, 2006.
- Hoddinott., J. Gilligan,O.D. Taffesse,S.A (2009) . The Impact of Ethiopia's Productive Safety Net Program on Schooling and Child Labor. Social Protection for Africa's Childrened by S. Handa S. Devereux, and D. Webb (London: Routledge).
- Holding Anyonge, C., Rugalema, G., Kayambazinthu, D., Sitoe, A. and Barany, M. (2006) 'Fuel wood, food and medicine: the role of forests in the response to HIV and AIDS in rural areas of southern Africa', *Unasylva* 224: 20-23

- Homma ,A.K.O.(1996). Modernization and technological dualism in extractive economy in Amazonia.In M.R.perez & J.E.M. Arnod (eds), current issues in non-timber forest products research.Bangor,Indonesia: CIFOR
- Howard P. and Smith E. (2006).Leaving two thirds out of development: Female headed households and common property resources in the highlands of Tigray, Ethiopia.FAO LSP WP 40.Access to Natural Resources Sub-Programme.
- Huffman, W.E. (1991). Multiple job holding among farm families. In: *Agricultural Households Survey and Critique*. Edited by J.L. Findeis, M.C. Hallberg and D.L. Lass.Chapter 5, Iowa State University Press, 1991, Ames IA, USA.
- Hujer, R., Caliendo, M. and S.L. Thomsen (2004). New evidence on the job creation schemes in Germany- a matching approach with three fold heterogeneity. *Research in Economics* 58:257-302.
- Hunter, L., Twine, W., and Johnson, A. (2008) Adult Mortality and Natural Resource Use in Rural South Africa: Evidence from Agincourt Health and Demographic Surveillance Site. Working Paper EB2005-0004, Boulder, CO: Institute of Behavioral Science, University of Colorado at Boulder
- Hunter, L., Twine, W., and Patterson, L. (2007) "'Locusts are now our beef'': Adult mortality and household dietary use of local environmental resources in rural South Africa', *Scandinavian Journal of Public Health* 35: 165-174
- IIED (2000). Look both ways: Mainstreaming Biodiversity and Poverty Reduction, IIED briefing, 2010
- Illukpitiya, P., Yanagida, J.F., 2008. Role of income diversification in protecting natural forests: evidence from rural households in forest margins of Sri Lanka. Agroforest. Syst. 74, 51–62.
- IMF (2008). Federal Democratic Republic of Ethiopia: 2008 Article IV Consultation Staff Report. IMF Country Report 08/264. Washington DC: International Monetary Fund.
- IUCN (2003). Sustainable Livelihoods. Media Brief for the World Parks Congress. IUCN, Gland.
- Jacoby, H., J.(1993). Shadow wages and Peasant Familiy Labor Supply: An Econometric Application to the Peruvian sierra, Review of Economic Studies, 60: 901-923
- Jalan, J. and M. Ravallion (2003). Estimating the benefit incidence of an antipoverty program by propensity-score matching. *Journal of Business and Economic Statistics*21(1):19–30.
- Jodha.S.N.(1986). Common Property Resources and Rural Poor in Dry Regions of India, *Economic and Political Weekly* Vol. 21, No. 27 (Jul. 5, 1986), pp. 1169-1181
- John M. Ulimwengu (2009), Farmers' Health Status, Agricultural Efficiency, and Poverty in Rural Ethiopia: A Stochastic Production Frontier Approach, IFPRI Discussion Paper 00868, June 2009
- Joshua Graff Zivin, Maria Damon, and Harsha Thirumurthy(2010). Health Shocks and Natural Resource Management: Evidence from Western Kenya. National Bureau of Economic Research (NBER) working paper series 16594, Cambridge

- Kaimowitz, D., 2004. Conventional wisdom about sustainable forest management and a pro-poor forest agenda. In: Zarin, D.J., Alavalapati, J.R.R., Putz, F.E., Smith, M. (Eds.), Working Forests in the Neotropics Conservation Through Sustainable Management? Columbia University Press, New York, USA, pp. 379– 387.
- *Kant, S., and N.G. Mehta. 1993.* Forest based tribal economy: a case study of Motisingloti village. Forests, Trees and People. 20:34-38.
- Kayambazinthu, D., Barany, M., Mumba, R., and Holding Anyonge, C. (2005) *Miombo woodlands and HIV/AIDS interactions: Malawi country report*, Forestry Policy and Institutions Working Paper 6, Rome: FAO
- Kazianga, H. and C. Udry (2006)."Consumption smoothing?Livestock, insurance and drought in rural Burkina Faso." Journal of Development Economics 79(2): 413-446.
- Klemick.H, Leonard.L, and Masatu.C (2007). Defining Access to Health Care: Evidence on the Importance of Quality and Distance in Rural Tanzania. WP 08-03,Department of Agricultural and Resource Economics, The University of Maryland, USA
- Kochar, Anijin (1999). Smoothing Consumption by Smoothing Income: Hours-of-Work Responses to Idiosyncratic Agricultural Shocks in Rural India, *The Review of Economics and Statistics*, February 1999, Vol. 81, No. 1, Pages 50-61
- Kondrashin AV and VS Orlov. 1989. Migration and Malaria. Chp. 23 in Demography and vector borne diseases. MW Service (ed.) CRC Press, Inc.: Boca Raton, Florida. pp. 353-365.
- Koziell. I.(2002). Diversity not adversity. Sustaining livelihoods with biodiversity. London: International Institute for Environment and Development (IIED)
- Lavang, P.,E.Daunias, et al.(2005)" out of Forest, out of Poverty?" Forests, Trees and Livelihoods 15(2): 211-235
- Lechner, M. 2000. An evaluation and estimation of causal effects of multiple treatments under conditional independence assumption, in Lechner, M. and Pfeiffer, F. (eds) Economic evaluation of labour market policies Heidelberg: Physica-Verlag.
- Lee, L. F., and R. P. Trost. 1978. Estimation of Some Limited Dependent Variable Models with Application to Housing Demand. *Journal of Econometrics* 8: 357–382.
- Lee, W. (2008). Propensity Score Matching and Variations on the Balancing Test. 3rd Conference on Policy Evaluation, ZEW, Mannheim (Germany), 27-28 October
- Little, P.D., 2008. Food aid dependency in North Eastern Ethiopia: myth or reality? World Development 36 (5), 860–874.
- Loksin, Michael and Sajaia Zurab (2004). Maximum likelihood estimation of endogenous switching regression models, The Stata Journal (2004) 4, Number 3, pp. 282–289

- Lopez-Feldman, A. Mora, J., & Taylor, J.E. (2007). Does natural resource extraction mitigate poverty and inequality? Evidence from rural Mexico and a Lacandona Rainforest. *Environment and Development Economics*. 12:251-269.
- Lopez-Feldman, Alejandro; Taylor, J. Edward and Yúnez-Naude, Antonio (2011): *Natural resource dependence in rural Mexico*. Published in: Investigacion Economica , Vol. 278, December 2011: pp. 23-44.
- LWAG. (2002). *Wildlife and Poverty Study*. Livestock and Wildlife Advisory Group.Department for International Development, London.
- Lybbert, T.J., Barrett, C.B., Desta, S. and Coppock, D.L. (2004) Stochastic Wealth Dynamics and Risk Management among a Poor Population. *The Economic Journal*, 114(498), pp. 750-777.
- Mackenbach, J.P. & Bakker, M. (2002), Reducing inequalities in health: a European Perspective, Routledge
- Maddala,G. S., and F. D. Nelson. 1975. Switching Regression Models with Exogenous and Endogenous Switching, Proceeding of the American Statistical Association (Business and Economics Section), pp. 423–426
- Maddala, G. S. 1983. *Limited Dependent and QualitativeVariables in Econometrics*. Cambridge, UK: Cambridge University Press.
- Mahanty Sango, Gronow Jane, Nurse Mike, and Malla Yam (2006). Reducing Poverty through Community Based Forest Management in Asia, Journal of Forest and Livelihood 5(1) February,2006
- Malla,Y.B., Neupane, H.R. & Branney, P.J. 2003. Why aren't poor people benefiting more from community forestry ? Journal of Forest and Livelihood, 3(1): 78-90
- Mamo,Getachew, Sjaastad Espen and Vedeld Pål (2007). Economic dependence on forest resources: a case from Dendi District, Ethiopia, *Forest Policy and Economics* 9 (2007) 916–927
- Mariara-K, J., & Gachoki, C. (2008). Forest dependence and household welfare: empirical evidence from Kenya. CEEPA Discussion paper-2008.
- Marmot, M. (1999), Multi-level approaches to understanding social determinants in L.Berkman & I.Kawachi, eds." Social Epidemiology Oxford': Oxford University Press
- Marshal, E. Schrecknberg, K. and Newton, A.C, (2006).). Commercialization of Non-timber forest products Factors influencing success. Lessons Learned from Mexico and Bolivia and Policy Implications for Decision- makers. UNEP World Conservation Monitoring Center, Cambridge,UK.
- Mayers, J. and S. Vermeulen. (2002). *How Good Forest Governance can Reduce Poverty*. WSSD Opinion Paper. IIED, London.
- McElwee, P.D., 2008. Forest environmental income in Vietnam: household socioeconomic factors influencing forest use. Environ. Conserv. 35, 147–159.
- McMichael, T. (2002) 'The Biosphere, Health, and "Sustainability", Science 297: 1093

- McSweeney. K (2003). Tropical forests as safety nets? The relative importance of forest product sale as Smallholder insurance, Eastern Honduras, Paper presented at The International Conference on Rural Livelihoods, Forests and Biodiversity, Bonn, Germany, 19-23 May 2003
- McSweeney Kendra (2004). Forest Product Sale as Natural Insurance: The Effects of Household Characteristics and the Nature of Shock in Eastern Honduras, Society and Natural Resources, Volume 17, issue1, 2004
- Mekuria, W., E. Veldkamp, M. Haile, K. Gebirehiwot, B. Muys and J. Nyssen. 2009. Effectiveness of exclosures to control soil erosion and local community perception on soil erosion in Tigray, Ethiopia.Afr.J. Agric. Res. 4:365-377
- MOFED(2002). Ethiopia: Sustainable Development and Poverty Reduction Program, Federal Democratic Republic of Ethiopia (FDRE), Ministry of Finance and Economic Development (MOFED), July 2002, Addis Ababa , Ethiopia
- MoARD (2006). Productive Safety Net Programme: Programme Implementation Manual (Revised), Addis Abeba : Ministry of Agriculture and Rural Development, Government of Ethiopia, Addis Abeba , Ethiopia
- MORAD (2009). Tigray Regional Overview and Livelihood Zone Summaries, Livelihoods Integration Unit, Ministry of Agriculture and Rural Development Disaster Management and Food Security sector, January ,2009 Addis Ababa, Ethiopia
- Morduch, J. (2004). 'Consumption Smoothing Across Space: Tests for Village-Level Responses to Risk', In Dercon, S. (ed.), *Insurance against Poverty*, Oxford University Press.
- Myers, N. (1988) Tropical forests: much more than stocks of wood. Journal of Tropical Ecology 4:209-221
- Nega.F, Mathijs. E, Deckers. J, Haile. M, Nyssen. J, Tollens. E (2008). Rural poverty dynamics and impact of intervention programs upon chornic and transitory poverty in Northern Ethiopia. A paper selected for presentation at the 2008 African Economic Conference, 12-14 November 2008 Tunis, Tunisia
- Narain, U., Gupta, S. and van't Veld, K.(2008). Poverty and the Environment: Exploring the Relationship between Household Incomes, Private Assets, and Natural Assets. Land Economics 88(2)
- Nedassa. B, Ali. J, and Nyborg. I(2005). Exploring Ecological and Socio-Economic Issues for the Improvement of Area Enclosure Management. A Case Study from Ethiopia, DCG Report No. 38
- Neumann P. Roderick and Hirsch Eric (2000). Commercialization of Non-Timber Forest Products : Review and Analysis of Research. Center for International Forestry Research (CIFOR), ISBN 979-8764-51-X
- Nyssen J, Poesen J, Moeyersons J et al (2004) Human impact on the environment in the Ethiopian and Eritrean highlands—a state of the art. Earth-Science Reviews 64:273-320
- Owusu V., Abdulai A. and Abdul-Rahman S. (2011); Non- farm work and food security among farm households in Northern Ghana; Food Policy 36(2011) 108-118.

- Parkes. M., Panelli, R., and Weinstein, P. (2003) 'Converging Paradigms for Environmental Health Theory and Practice', *Environmental Health Perspectives* 111: 669-675
- Pattanayk K. Subhrendu, and Sills.O Erin (2001). Do Tropical Forests Provide Natural Insurance? The Microeconomics of Non-Timber Forest Product Collection in the Brazilian Amazon, Land Economics, Vol.77, No. 4 (Nov., 2001), pp. 595-612
- Pattanayak.K, Corey.G, Yewah, Lau.F and Kramer.A (2011).Forest malaria: A microeconomic study of forest protection and child malaria in Flores, Indonesia.
- Paumgarten Fiona(2005). The Role of non-timber forest products as safety-nets: A review of evidence with a focus on South Africa, Geo Journal(2005) 64: 189-197
- Paxson.H, Christina (1992). Using Weather Variability to Estimate the Response of Savings to Transitory Income in Thailand, The *American Economic Review*, *Review*, Vol. 82, No. 1 (Mar., 1992), pp.15-33
- Pender .J and Gebremedhin .B (2007).Determinants of Agricultural and Land Management Practices and Impacts on Crop Production and Household Income in the Highlands of Tigray, Ethiopia. Journal of African Economies , Volume 17, Number 3, PP. 395–450 doi:10.1093/jae/ejm028 online date 24 November 2007
- Peters, C. M., A. H. Gentry, and R. O. Mendelsohn.(1989) Valuation of an Amazonian rainforest.*Nature*339:655–656.
- Plotkin, M. and Famolare, L. (Editors) (1992) Sustainable Harvest and Marketing of Rain Forest Products, Conservation International-Island Press, Washington DC, pp.xiii-xv.
- Rasmus Heltberg and Niels Lund (2009. Shocks, Coping, and Outcomes for Pakistan's Poor: Health Risks Predominate. Journal of Development Studies, Vol. 45, No. 6, pp. 889-910, 2009
- Reddy, C.R.S and Chakravarty.P.S(1999). Forest Dependence and Income Distribution in a Subsistence Economy: Evidence from India, World Development Vol. 27, No. 7, pp. 1141±1149, 1999
- RHVP (2007). Lessons from Ethiopia on a Scaled –Up National Safety Net Programme, Wahenga, briefs No. 14, August 2007
- Riphahn , R.T.(1999), "Income and employment effects of health shocks a test case for German welfare state", Journal of Population Economics 12(3), 363-389

Roe. D and Elliott. J (2005). Poverty - Conservation Linkages: A conceptual Framework. Poverty and

Conservation Learning Group, IIED, November, 2005

- Roe, D and Elliott,J (2005). Poverty Conservation Linkages: A conceptual Framework. Poverty and Learning Group,IIED, November , 2005 pdf.conceptualfarmework/pdf/org.danadeclation.www//:http
- Roe Dilys(2003). The Millennium Development Goals and natural resources management: Reconciling Sustainable Livelihoods and Resource Conservation or fuelling a divide?

http://www.tanzaniagateway.org/docs/MDGs and natural resources management sustainab le livelihoods resource.pdf.

- Rose Elaina (2001). Ex ante and ex post labor supply response to risk in a low-income area, Journal of Development Economics, Vol. 64 2001 371–388
- Rosenbaum, P.R. and D. Rubin (1983). The central role of the propensity score in observational studies for causal effects. *Biometrika* 70:41-45.
- Rosenzweig, M. and Wolpin, K. (1993). 'Credit Market Constraints, Consumption Smoothing, and the Accumulation of Durable Production Assets in Low-income Countries: Investment in Bullocks in India', *Journal of Political Economy*, vol. 101(2), pp. 223–44.
- Rosenzweig, M. (1988).Risk, Implicit Contracts and the Family in Rural Areas of Low-Income Countries. The Economic Journal 98 (393), 1148-1170.
- Ruiz-Perez, M., Belcher, B. et al. (2003). Global patterns and trends in non- timber forest product use. Manuscript, Centre for International Forestry Research (CIFOR), Bangor, Indonesia
- Sapkota I, Oden PC (2008) Household characteristics and dependency on community forests in Terai of Nepal. International Journal of Social Forestry 1(2),123-144.
- Sauerborn R, Adams A and Hein M (1996) 'Household strategies to cope with the economic costs of illness,' *Social Science and Medicine* 43(3):291-301.
- Saxena, N.C. 2003. Livelihood Diversification and Non-Timber Forest Products in Orissa: Wider Lessons on the Scope for Policy Change? ODI Working Paper No. 223. London, U.K.: Overseas Development Institute.
- Scherr, Sarah J., Andy White and David Kaimowitz. 2002. *Strategies to Improve RuralLivelihoods through Markets* for Forest Products and Services. Washington, DC:Forest Trends.
- Schmidt, Ralph, Joyce K. Berry and John C. Gordon (eds.). 1999. *Forests to FightPoverty: Creating National Strategies*. New Haven and London: Yale UniversityPress
- Schultz, T. and A. Tansel (1997). Wage and labor supply effects of illness in Côte d'Ivoire and Ghana: instrumental variable estimates for days disabled. Journal of Development Economics 53 (2), 251-286.
- Shackleton, C., Shackleton, S., 2004. The importance of non-timber forest products in rural livelihood security and as safety nets: a review of evidence from South Africa. South African Journal of Science 100, 658–664.
- Shackleton, C., Shackleton, S., 2006a.Household wealth status and natural resource use in the Kat River valley, South Africa. Ecological Economics 57, 306–317.
- Shackleton, C.M., Shackleton, S.E., 2006b. Household wealth status and natural resource use in the Kat River valley, South Africa. Ecological Economics 57, 306–317.
- Shackleton, S., Kaschula, S., Twine, W., Hunter, L., Holding Anyonge, C. and Petheram, L. (2006c) Forests as safety nets for mitigating the impacts of HIV/AIDS in southern Africa, Bogor: CIFOR

- Shackleton, C, Shackleton, , Gambiza. J, Nel. E, Rowntree.K, and Urquhart.P (2008). Links between Ecosystem Services and Poverty Alleviation: Situation analysis for arid and semi-arid lands in southern Africa. Consortium on Ecosystem and Poverty in sub-Saharan Africa (CEPSA),2008
- Shively, G.E. (1997) Poverty, technology, and wildlife hunting in Palawan. Environmental Conservation 24(1), 57-63
- Sianesi, B. (2004). Evaluation of the active labor market programmes in Sweden. *The Review of Economics and Statistics* 86 (1):133-155.
- Sills, E.O., Lele, S., Holmes T.P, and Pattanayak S.K.2003. Forest in a Market Economy, 259-281, Sills and Abt (eds), Netherlands
- Singh, I., Squaire, L. & Strauss, J. (1986). The basic model: theory, empirical results, and policy conclusions. In: Singh, I., Squire, L. & Strauss, J. (eds.) Agricultural households' models. Johns Hopkins University Press
- Sivakumar, M.V.K., R.P.Motha and H.P.Das.2005. Natural Disasters and Extreme events in Agriculture, Impacts and Mitigation. Heidelberg: Sprinnger-Varlag
- Smith, J. and P. Todd (2005). Does matching overcome LaLonde's critique of non-experimental estimators? *Journal of Econometrics* 1251-1252.
- Smith, J. and P. Todd (2001).Reconciling Conflicting Evidence on the Performance of Propensity Score Matching Methods.*The American Economic Review* 91(2): 112-118.
- Smith, J.P.(1998), "Socio-economic Status and Health", The American Economic Review 88(2) 192-196
- Stark , O., Taylor , J.E., and Yitzhaki, S. (1986). Remittances and inequality. The Economic Journal, 96(383):722-740
- Sunderlin D. and Huynth Thu Ba (2005). Poverty Alleviation and Forests in Vietnam ,Center for International forestry Research (CIFOR), Indonesia , ISBN: 979-3361-57-3
- Taffere, B. 2003. "Efforts for Sustainable Land Management in Tigray: The Role of Extension." In *Policies for Sustainable Land Management in the Highlands of Tigray, Northern Ethiopia:* Summary of Papers and Proceedings of a Workshop Held at Axum Hotel, Mekelle, Ethiopia, 28–29 March 2002, edited by B. Gebremedhin. Issue 54 of Socio-Economics and Policy Research Working Papers. Mekelle University, Mekelle, Ethiopia, International Livestock Research Institute, International Food Policy Research Institute
- Takasaki, Y., B. L. Barham, and O. T. Coomes. 2002. Risk coping strategies in tropical forests: floods, illness, asset poverty, and natural resource extraction. Paper read at 2nd World Congress of Environmental and Resource Economists, 23-27 June, at Monterey, California.
- Takasaki, Y., B.L. Barham, and O.T. Coomes (2004), 'Risk coping strategies in tropical forests: floods, illness, and resource extraction', Environment and Development Economics **9**: 203-224.
- Takasaki Yoshito (2009). Forest and Sea as Insurance among Fujians: Tsukuba Economics working papers N0. 2009-002, University of Tskuba , Department of Economics , Japan

Taylor, D.A.(1999). Requisites for thriving rural non-wood forest product enterprises, Unasylva, 50(198) ,pp.3-8

- Taylor, J.E., Mora, J., Adams, R., Lopez-Feldman, A. (2005). Remittances, inequality, and poverty: Evidence from rural Mexico. ARE working papers 05-003. Davis, CA: University of California, Davis, Department of Agricultural and Resource Economics.
- Tesfay, G. (2006). Agriculture, Resource Management and Institutions: A socioeconomic analysis of households in Tigray, Ethiopia, PhD thesis (2006), Wageningen University ISBN 90-8504-532-0
- Tewolde-Berha.S. Mitlohner.R, Muys.B and Haile. M (2002). Comparison of Vegetation Development of Closed Areas and ancient Forest in Tigray, Ethiopia. Paper presented in Conference on International Agricultural Research for Development, Witzenhausen, October 9-11, 2002, Germany
- Timmer Vanessa (2005). Taking root: biodiversity conservation and poverty reduction come together in the tropics lessons learned from the equator initiative.

https://www.cbd.int/doc/articles/2005/A-00307.pdf

- Timko, J.A., Kozak, R.A. and Innes, J.L. (2010) 'HIV/AIDS and forests in Sub-Saharan Africa: exploring the links between morbidity, mortality, and dependence on biodiversity', *Biodiversity* 11: 45-48
- Titus, M. A. (2007). Detecting selection bias using propensity score matching, and estimating treatment effects: an application to the private returns to a Master's degree.*Research in Higher Education* 48(4):487-521
- Tongruksawattana, T., Schmidt, E. and H. Waibel (2008).Understanding Vulnerability to Poverty of Rural Agricultural Households in Northeastern Thailand.Tropentag, Hohenheim, 07.-09. October 2008.
- Topouzis, D.N. (2007) 'HIV/AIDS: a risk to the social and economic sustainability of forestry in Sub-Saharan Africa?', *International Forestry Review* 9: 558-562
- Torell, E., Tobey, J., Thaxton, M., Crawford, B., Kalangahe, B., Madulu, N., Issa, A., Makota, V., and Sallema, R.
 (2006) 'Examining the linkages between AIDS and biodiversity conservation in coastal Tanzania', *Ocean & CoastalManagement* 49: 792-811
- Townsend, R. (1994). "Risk and Insurance in Village India." Econometrica 62: 539-591.
- TRBPF (2010). Tigray Regional State Five Years (2010/11-2014/15) Growth and Transformation Plan.
- Twine, W. and Hunter, L. (2008) *AIDS Mortality and the role of natural resources in household food security in a rural district of South Africa*, HIV, Livelihoods, Food and Nutrition security: findings from renewal research (2007-2008) Brief 11, Washington, DC: IFPRI
- Udry, C. (1990). Credit markets in Northern Nigeria: Credit as insurance in a rural economy. World Bank Economic Review 4 (3), 251 -269.
- UNDP-GEF (2005) United Nations Development Program—Global Environment Facility (UNDP-GEF). "Sustainable Development of the Protected Area System of Ethiopia" (2005).

- USAID (2008) Ethiopia Biodiversity and Tropical Forests 118/119 Assessment EPIQ IQC: EPP-I-00-03-00014-00, Task Order 02
- Vedeld, P., Angelsen A., Bojo" J. et al. (2007). Forest environmental incomes and the rural poor. Forest Policy and Economics9, 869–879.
- Villarreal, M., Holding Anyonge, C., Swallow, B., and Kewsiga, F. (2006) 'The challenge of HIV/AIDS: Where does agroforestry fit in?' chapter 23 in D. Garrity, A. Okono, M. Grayson, and S. Parrott (eds) World Agroforestry into the Future, Nairobi: World Agroforestry Centre
- Vira B. and Kontoleon A. (2010); Dependence of the poor on biodiversity: which poor, what biodiversity? University of Cambridge June (2010): Paper Presented for the CPRC International Conference 2010, Manchester, and September 8-10, 2010
- Volker, M., Waibel .H, (2010). Do rural households extract more forest products in times of crisis? Evidence from the mountainous uplands of Vietnam : forest Policy and Economics 2010.
- WFP (2009). Food Security and Vulnerability in Selected towns of Tigray Region, Ethiopia: WFP-Ethiopia Vulnerability Assessment and Mapping (VAM) Addis Ababa, Ethiopia September 2009
- Weinberger, K. and J. P. Jutting (2000), "The Role of Local Organizations in Risk Management: Some Evidence from Rural Chad", Quarterly *Journal of International Agriculture*, Vol 39(3), pp.281-299.
- While, A. & Martin, A. (2002). *Who owns the world's forests? Forest tenure and public forests in transition*. Washington DC, Forest trends.
- Wilcox, B.A., Aguirre, A.A., Daszak, P., Horwitz, P., Martens, P., Parkes, M., Patz, J.A. and Waltner-Toews, D. (2004) 'EcoHealth: A Transdisiplinary Imperative for a Sustainable Future', *EcoHealth* 1: 3-5
- Wolde-Georgis, Tsegay. 1996. Survival strategies and coping mechanisms of peasants to famine: the case of Tigray, Ethiopia. PhD Dissertation, University of Denver (Colorado). Ann Arbor, Michigan: University Microfilms International.
- Woldehnna. T(2009). Productive safety net programme and children's time use between work and schooling in Ethiopia . Young lives Working Paper 40, 2009
- World Bank (2011). The Federal Democratic Republic of Ethiopia Joint IDA-IMF staff Advisory note on the Growth and Transformation Plan (GTP) (2010/11-2014/2015), World Bank Report No. 63592- ET, August 15, 2011, Washington DC
- World Bank (2008).World Bank/Global Environment Fund Support Efforts to Reverse Land Degradation in Ethiopia. Press Release No: 2008/288/AFR. Accessed November 20.2011. http://web.worldbank.org/WBSITE/EXTERNAL/NEWS/0,,contentMDK:21750129~pagePK:34370~piP K:34424~theSitePK: 4607,00.html.
- World Bank (2007). Poverty and Environment: Understanding Linkages at Household level, World Bank Report, No 40218
- World Bank (2002). A revised forest Strategy for the World Bank Group., World Bank Group, Washington DC.

- Wunder, S. (2001). Poverty alleviation and tropical forests: What scope for synergies? World Development, 29(11):1817–1833.
- Yesuf, M., S. Di Falco, T. Deressa, C. Ringler, and G. Kohlin. 2008. The Impact of Climate Change and Adaptation on Food Production in Low-Income Countries: Evidence from the Nile Basin, Ethiopia. IFPRI Discussion paper 00828, December 2008
- Zimmerman, F. J., & Carter, M. R. (2003). Asset smoothing, consumption smoothing, and the reproduction of inequality under risk and subsistence constraints. Journal of Development Economics, 71(2), 233–260.

ANNEX A. HOUSEHOLD QUESTIONNAIRE

Name of Enumerator

Household Questionnaire

HOUSEHOLD IDENTIFICATION		NAME	CODE
Household head			
Name of village			
Tabia Name			
Woreda			
Region			
Name of interviewee			Sex $1 = Male$
			2=Female
Distance to DA office from home	Distance to Nursery	Distance to Health post	Distance to Veterinary

	Distance to seasonal road	Distance to School	Distance to input distribution
	Distance to all weather road	Distance to Forest	
Distance to be advected			
Distance to local market			
Distance to Woreda Market			
Date of interview			
	Date://2010/11		Checked by:
	Storet dimon		
	Start ume:		••••••
	Finish time:		Approved:
Reasons for not conducting interview:	•		Household location GPS
			Coordinates:
			N
			E

A. Provide the details of each household member

Me mbe r ID	Name of household member	Sex	Relatio nship with HH head	Marital status	Age	Education			Main occupatio n	How many months did the name live here in the last 12 months	If they left the home when did they live?	How many times did the name face serious illness in the past season for more than 3 weeks.
	Al	A2	A3	A4	A5		A6	I	A7	A8	A9	A10
						Number of years of schoolin g (a6.1)	Highest class attende d (a6.2)	Highest level of education completed (a6.3)	What is the name of proffesion or activity			
01												
02												
03												
04												
05												
06												
07												
08												
09												
10												
11												
12												
13												
14												
Code												

A2 1=female 2=male

A3 1= husband 2= wife 3= son 4= daughter 5= Grandchild 6=Brothet 7=sister 8=neice 9= nephew 10=other relatives (specify)

A4 1=Married 2=Widowed 3=Divorced 4= separated 5=Never married A6.3 0=none 1=std 1-4 2= std 5-8 3= Attend sec 4=MSCE 5=Techn. Colle 7=University A7 0=none 1= Farming 2=bussiness 3=ganyu (labour) 4=Salaried work 5=schooling 6=Unemployed 7=other (specify)

A10 0=none 1=once 2=twice 3=three times 4=whole season

B: Assets ownership

Items	Does your household own the following itoms	How many items do you have?	How much did you (Birr)	pay for it?	When	When did you acquire them? (year)			acquired, item new?	was the ?	If you were to sell them today what will be the price? (Birr)		
	B1 1=yes 0=no (go to D6)	B2	B2.1		В3		B3.1 1=yes			B4			
									0–110				
Metal or Wooden Bed													
Metal or Wooden Table													
Metal washing Pot (Tisti)													
Plastic Washing Pot (lastic Tisti)													
Stone Mills (Methan)													
Saddle (Korecha)													
Fanus (Betromakis)													
Gas Midija													
Blanket (Koberta)													
Wooden Box (satsun)													
Metal box (satsun)													
Bermel													
Jerikan													
Plowing Set (Mesarei)													
Mahresha													
Hoe + Af kutu (Mekuati)													
Spade (badela)													
Sickle (Meatsid)													
Hammer (Martello +													
Medosha)													
Axe + Fas (misar)													
Saw (megaz)													
Bunta													
Miran + Gemed													

Arebia							
Cart (Gari)							
Radio							
Таре							
Wrist Watch							
Silver (grms)							
Gold (grms)							
Beehive (Kefo)							

C: NON-FARM ACTIVITIES AND INCOME C.1 EMPLOYMENT FOR WAGE

1. In the last twelve months, did any of the household members work off the household's land either on someone else's land or in some other employment or against payment in cash/kind? If yes give the following details. Yes ---- | o ----- 2

ID code of	C2. Kind of work	C3. Location of	C4. Did s/he need Qualification/	C5. Is it permanent	C6. Total day	s worked in each	1 season	C7. Tota Birr	l amount	earned in	C8. What is the income from
H.H Member	[code a]	employment	Experience/	(=1) or temporary							this activity used for?
Wielinder		[coue b]	training	(=2)							[code c]
			[code i]	work?							Multiple codes are possible
					1 st season	2 nd season	3 rd season	1 st	2 nd	3 rd	
					(Jan-April)	May- August	Sept- December	season	season	season	

Code (a) Type of employment

Farm Worker (for pay) = 1; Traditional labor sharing = 2

Professional (teacher, government worker, administration, health worker, clerical) = 3

This kushet = 1; Other kushet in the same Tabia = 2;

This Woreda = 3; This neighbouring Woreda = 4; Mekelle = 5;

Migration to another zone = 6; Migration to another state = 7;

Migration to foreign country = 8;

Other = 9, specify

e (b) Location of employment kushet = 1; Other kushet in the same Tabia = 2; Woreda = 3; This neighbouring Woreda = 4; Mekelle = 5; ation to another zone = 6; Migration to another state = 7; ation to foreign country = 8; r = 9, specify
e (c) Use of incomeInvested into the business = 2;ral purchase for the household =1;Invested into the business = 2;mal purchases from the person who runs the business = 3;Expenditure for the children = 4;d = 5;Purchase livestock/other asset = 6;ayment of taxes and contribution = 7;Payment for school = 8;lebt settlement = 9purchase of fertilizer = 10r = 11, specifyPayment for school = 8;
If any member of your household gets additional daily work would s/he work? (YES1 NO2)
. If your answer to Q.9 is no, why did no member(s) seek other/more employment? Put according to their importance [code d].
Reason 1:; Reason 2:; Reason 3:
. Would any of the household members like to work more for wages during planting time; weeding time harvesting season;
Threshing season (yesl; no2)
, HOUSEHOLD MOTIVATION TO WORK ON FARM AND OFF FARM (Multiple codes are possible)
1. Why did you choose to work only on the farm? [codes for the reason to work only on the farm code k]
2. Why did you choose to work also on the farm and/or only off the farm?[codes for the reason to work off the farm code L]

Code (d) Reasons for not seeking other work

No employment opportunities = 1 Needed on farm = 2 Jobs too far away = 3 Wages too low for the kind of work = 4 Just do not want to work off farm = 5 Respect holidays = 6 Others = 7, specify

Code (k) codes for the reason to work only on the farm

Because I don't like and I am not interested working off the farm =1; Because off-farm work is less profitable than farm work = 3; Because I am retired = 5; Because I can not sell or rent my land = 7; Because I can not find any job off the farm = 2; Because working on the farm allows caring for my family = 4 Because I need more labour for on farm work = 6; other = 8

Code (L)codes for the reason to work off the farm

Because no more labor is needed on farm $=1$;	Because my education is outside agriculture = 2;
Because I do not have enough land to support family $= 3$;	Because I do not like/I am not interested to work on farm = 4
Because of farm work is more profitable = 5;	Because farm work would be in sufficient for living $= 6$;
Because I can not buy additional land = 7;	other $= 8$

C13. Are you a beneficiary of Productive Safety Net Program? (PSNP) 1. Yes 2. No

C14. If yes, in which component of PSNP are you involved? 1. Public works 2. Direct Support

C15. If you are involved in public works program, how many of the household members are involved? -----

C16. Are you a beneficiary of household extension packages? 1. Yes 2. No

C17. If yes what agricultural production development input supplies and technologies were you getting? 1. Fertilizer 2. Improved Seed 3. Improved livestock 4.

Modern Beehive 5. Irrigation technology 6. Credit services 7 Others, Please specify ------

C18.Did you use those inputs and technologies prior to the Household extension package program? 1. Yes 2. No

C19. What is the amount of Credit you get from the household extension package? ------

C21. OWN BUSINESS ACTIVITIES

1. Would like to ask you about your income earning activities such as craft, trades, or other business, carried out by any of the household members this year.

If any of the household member	s are involved in such	n activities fill the following.
--------------------------------	------------------------	----------------------------------

		C22			C23			C24		C25
Activities	HH	How much has t	he household ear	ned net? If given	Total days worked by the HH			Total hi	red	what is the
	member	in kind, change	to cash and inclu	de it as payment	-		labour u	sed	income usually	
	responsible	[tirfi Tirah] and	put it in Birr ear	rned in each						used for
	[ID code]	season								General purchase for the household =1; Invested into the business = 2; Personal purchases from the person who runs the business =3; Expenditure for the children = 4;
		1 st season (Jan-April)	2 nd season MarchAugu	3 rd season Sept-Decembe	1 st season (Jan-April)	2 nd season March-Aug	3 rd season Sept-decemb	Total hours worked	Paid wage in Birr	Saved = 5; Purchase livestock/other asset = 6; For payment of taxes and contribution = 7; Payment for school = 8; For debt settlement = 9purchase of fertilizer = 10 Other = 11, specify
Weaving (shimena)										
Milling (metehan)										

Handicraft,					
including pottery					
Trade in					
grain/general					
Trade in livestock					
Traditional healer/					
Religious teacher					
Transport (by pack					
animal)					
Selling Tela, Arequi,					
Teii, Kolo, and					
iniera					
Other, specify					

C26. OWN BUSINESS ACTIVITIES

1. Would like to ask you about your income earning activities such as selling fuel wood and charcoal, carried out by any of the household members this year.

If any of the household members are involved in such activities fill the following.

		C27			C28		C29		C30	
Activities	HH	How much has t	he household ear	ned net? If given	Total days wo	rked by the HE	I	Total hi	red	what is the
	member	in kind, change t	to cash and includ	le it as payment			labour u	sed	income usually	
	responsible	[tirfi Tirah] and	put it in Birr ear	ned in each						used for
	[ID code]	season	•							General purchase for the household -1: Invested into the business - 2:
										Personal purchases from the person
										who runs the business =3; Expanditure for the abildran = 4;
										Saved = 5; Purchase 4 ,
								Total	Paid	livestock/other asset = 6; For payment of taxes and
		1 st season	2 nd season	3rd season	1st season	2 nd season	3rd season	hours	wage	contribution = 7 ; Payment for
		(Jan-April)	MarchAugu	Sept-December	(Jan-April)	March-Aug	Sept-decemb	worked	in	school = 8; For debt settlement =
								workeu	Birr	Other = 11, specify 10^{-10}
Selling Honey										
Selling Cactus										
Thatching grass										

Fuel Wood					
Charcoal					
Leaves					
Crop Residue					
Selling Wood					
animal)					
Other, specify					

C31. TRANSFERS (REMITTANCE AND AID)

C31.1 Type of receipt Remittance 1; Food Aid 2; Gift 3; Inheritance 4; Dowry 5; Other 6 Specify	C32. Person who receive it (ID code)	C33 The person who send the transfer Non residence household member 1; Relatives of household member 2; Friends 3; Gov't/organization 4; NGO 5; Other 6 Specify	C34 Amount received - If it is given in kind change it to cash Amount in Birr	C35 What was this income mainly used for General purchase for the household =1: Invested into the business = 2; Personal purchases from the person who runs the business =3; Expenditure for the children = 4; Saved = 5; Purchase livestock/other asset = 6; For payment of taxes and contribution = 7; Payment for school = 8; For debt settlement = 9purchase of fertilizer = 10 Other = 11, specify
-------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------------------------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------

C36: MIGRATION AND INCOME

1. Has any member left the household to seek work this year?

Yes 1 (Give Details) 2. No

	C37	C38	C39	C40	C41	C42	C43	C44 How much	C45 Did the HH have	C46 did you come
ID code Of HH member	where did s/he go [code b]	date of departure Day/month/ year	Date of return (if s/he has returned)	did s/he get work? Yes 1; No2	Type of work [code a]	How much s/he earn? (in Birr)	How much s/he send back? (in Birr)	did you bring back when you returned (if s/he is returned) (in Birr)	land allocated away while this person is away? if so how many tsemdies? [yes=l, no=2]	back to avoid losing land [yes=l, no=2]

a) Type of employment

Farm Worker (for pay) = 1; Traditional labour sharing = 2

Professional (teacher, government worker, administration, health worker, clerical) = 3

Labourer (skilled i.e. builder, thatcher, hair cutting or dressing) = 4

Trader =5 ; soldier = 6 ; driver/Mechanic= 7 ; unskilled worker = 8

 $Domestic \ servant = (yebet \ agelgay) = 9; Food \ for \ work = 10; \ Others = 11 \ Specify$

b) Where did she/he go or migrate Regional City (Mekelle= 1; Humera/shire =2 Addis Abeba = 3; other region in Ethiopia = 4;

Foregib country = 5; Other = 6 specify

C47. Availability of Food During each month last year (2009/10)

C47.1. During which month of the last year ,did your household had enough or shortage of food

No	Month	Jan	Feb	March	April	May	June	July	August	Sep.	Oct.	Nov.	Dec.
1	enough												
2	Not -enough												

C47.2. Did you sell assets for immediate consumption during 2009/10? 1. Yes 2. No if yes Please specify ?------

L. FOOD CONSUMPTION AND EXPENDITURE

L.1. How many meals per day did your household eat in the last four weeks?

L.2.We would like to ask you about all the food that was bought for consumption and/or was consumed from your own stock, in

this year. Please do not include food bought for resale, even after processing (the sum from the different sources should be

equal to the total amount consumed).

Food type	Code	L3 Total food consumed	l	L4 Consum	ed from	L5 Consume	ed from	L6 Consumed	from gift or	L7 Consumed f	from other
consumed		[KG, Birr]		purchased		own harvest		food aid		sources	
		Amount	Amount Value		Value	Amount	Value	Amount	value	Amount	Value [Birr]
		{ <i>KG</i> !	{Birr!	[KG!	{Birr]	{ <i>KG</i> !	{Birr!	<i>{KG]</i>	[Birr!	{ <i>KG</i> !	

Cereals	Teff						
	Barley						
	Wheat						
	Maize						
	Sorghum						
	Finger millet						
	Karka'eta						
Pulses	Lentils						
	Faba Bean						
	Field peas						
	Chick peas						
	Guaya						
Oil crops	Linseed						
	rape seed						
	sesame						
	sun/sufI						
	flower						
	Nug						

	Food type	Code	L3		L4		L5		L6		L7	
	i oou type	Coue	Total foo	od	Consume	d from	Consume	d from	Consume	ed from	Consume	d from other
	consumed		consume	d [KG,	purchased	1	own harve	est	gift or fo	od aid	source	
			Birr]									
			Amount	value	Amount	value	Amount	value	Amount	value	Amount	Value
			[KGl	[Birr]	[KG]	[Birr]	[KG]	[Birr]	[KG]	[Birr]	(KG]	(<i>KG</i>]
Milk and	milk/vogurt											
animal	mink/yogurt,											
products	cheese											
	beef meat											
	mutton/goat											

	meet						
	Chicken						
	eggs						
Beverage (liters)	Tella						
	Arequi						
	Тејі						
	Birra						
	Soft drink						
	Coffee						
	Honey						
	Sugar						
	Tea						
	Salt						
	Cooking oil						
	karialberbere						
	Onion						
	garlic						
	Other spices						

Food type consumed	Code	L3 Total food [KG, Birr]	consumed	L4 Consumed purchased	from	L5 Consumed f harvest	rom own	L6 Consumed f food aid	rom gift or	L7 Consumed f source	rom other
		Amount [KG]	Value [Birr]	Amount	Value [Birr]	Amount [KG]	Value [Birr]	Amount [KG]	Value [Birr]	Amount [KG]	Value [Birr]
 Bread											
Macaroni											
Potato											

	Sweet potato						
Vegetables	Green leaf						

L8. Has the household purchased any prepared foods, or eaten elsewhere against payment in the year?

[Yes....1; no....2]_____ If yes, total expenditure in this year [in Birr]_____

L. Non – Food Expenditure

Which of the following items		How much	Which of the following items		How much
did you buy or pay for in the last 7	Yes=1	did you	did you buy or pay for in the last month?	Yes=1	did you pay
days?	No=0	pay for it?		No=0	for it?
	L1a	L1b	L2a	L2a	L2b
L1a					
shoes, fabric for ADULTS (MEN AND			Soan, Omo		
WOMEN)			20 4P , 0110		
Clothes/ shoes, fabric for CHILDREN			Linens (sheets, towel, blankets)		
(BOTH BOYS AND GIRLS)					
Cosmetic (butter) for MEN, WOMEN,			Furniture and Lamp/torch		
BOYS AND GIRLS			I I I I I I I I I I I I I I I I I I I		
Entertainment by MEN			Transport materials		
Entertainment by WOMEN			Building material for house		
Entertainment by BOYS and GIRLS			Ceremonial expense		
Kitchen equipment (cooking pots, Medija)			Contribution to EDIR		
Energy consumption (Kerosene, Fuel wood,			Contribution to Associations (women, youth,		
charcoal, match)			farmers association)		
Modern medical treatment and medicine			Donation to organization (TDA, TPLF etc)		
Traditional medical treatment and medicine			Donation to Community church/Mosque		
School fees			Taxes and contributions to Tabia		

Characteristics of Plots owned by the household in 2009/2010 (2001)

Plo	Name of	Distanc	What is	Is it	What is the	What is	What is the	How did you	Is the plot	What was the primary
t	plot	e from	the size	Irrigated	general texture	the slope	general	acquire this	titled? (1=	use of the plot during
ID	_		of your	type? (1=	of the soil?	_	_	plot?	yes, 2= No)	

	F1	home to the plot	plot in tsimdi ? F3	yes ,2= No) F4	F5	of the plot?	fertility of the plot?	F8	F9	Belg2001 and Meher 2002 F10
		F2				F6	F7			
1										
2										
3										
4										
5										
6										
7										
8										
					1_condy	1 flat	1 yany fantilia			1= annual crops
					1=sandy 2=loam	2-slight	2-average			2= permanent crops
					3=clay	3-steep	3-not fertile			4=left fallow $5=$ other

Codes F8= 1=granted by local leaders, 2=Inherited from mothers side (wife), 3=Inherited from fathers side(wife), 4=Inherited from mothers side (husband), 5=Inherited from fathers side(hasband), 6=Rented, 7=purchased, 8=farming as

G. List crops and inputs on each plot in the last cropping season (2009/10)

Plot ID	Crop Code	SEEDS				PESTICIDES					FERTILISER			
		Source	Type / Variety	Amount G4	Cost	Source	Type / Variety	Amount G4	Cost	Source	Type / Variety	Amount G4	Cost	Did you use subsidized fertilizer on this plot? <i>1-yes</i> <i>0-no</i> <i>G14</i>
1														

2

G2, G6 G10; 1=own 2=bought(own money) 3 bought (credit) 4= bought(coupon) 5 =gift 6=others

G9;1=*CAN* 2= *Urea*, 3=23:21:0, 4=20:20:0, 5=*D* compound, 6= super *D*, 7= SA, 8= others (specify

Input use cont ------

Plot	Plot name	Did you apply any	Manure									
ID		manure on this plot? 1-yes 0-no G15	What was the type of manure? <i>G16</i>	Amount G.	of manure 17	Source of manure G18	If bought how much did it cost (MK) <i>G19</i>	How many days did it take you to apply the manure?				
				Quantity	Unit			G20				
1 2												
3												
4 5												
6												
7 8 9 10 11			1=Compost 2=wastes 3=livestock 4=green manure 5= tobacco stems 6=others		1= basket 2= oxcart 3=pail 4=wheeelbarrow 5=bags (50kg) 6=bags (90kg) 7= bales 8=Nkhokwe 9= lichelo (basin)	 self made (compost) own animal manure given by friend relative bought other 						

Plot ID	What crops w	ere grown or (2009/10	n this plot last))?	Identify type of Cropping System	What factors are taken into account in making decision on what crops to grow on each plot or leaving the plot fallow? (in order of priority starting with the most important)				What major reasons did the household have for monocropping or mixed cropping? (in order of priority starting with the most important)				
	1 st	2 nd	3 rd	4 th									
	Fc1	Fc2	Fc3	Fc4	Fc5		F	c7			i	Fc8	
1													
2													
3													
4													
5													
6													
7													
8													
9													
10													
11													
12													
	Crop codes 0 fallow 1 Maize Hybrid 2 Compost Maize (OPV) 3 Maize Local 4 Beans Dry 5 Beans Green (Zitheba) 6 Peas 7 Ground nuts 8 Tobacco 9 Cassava 10 Pigeon peas 11 Irish potato 12 Sweet Potato 13 Cabbage	14 Tomatoes 15 Onions 16 Lettuce 17 Rape 18 Mpiru 19 Pumpkins 20 Garlic 21 Cucumber 22 rice 23 Millet 24 sorgum 25 sugarcane 26 soyabeans 27 other (specify)		1= Mixed cropping 2= Monocropping 3= Intercropping	1= Land availab 2= Labour avail 3= Prevailing m 4= Seeds, fertili 5= Meeting hou 6= Past crop per 7= Expected raii 8= Crop rotatior 9= Other (specif	ility ability arket prices ser, availability sehold basic consu formance (in previ afall patterns.	imption needs Cred	it	1= Maximise revenue from land 2= Allow positive complementarity efects among crops (e.g. N-fixing,) 3= Save time and labour in crop management 4= To produce quality standards for exclusive for marketing 5= other				

Fc: Crops grown on each plot

Labor use on plots

For each of the plots and crops cultivated by the household indicate how many man-days did household member work in the following activities within the last season (2009/10)

Plot ID	Land pro G	eparation 21	Planting G23		Fertilizer application G25		Weeding G26		Harvesting G27	
	No of members	No of days	No of members	No of days	No of members	No of days	No of members	No of days	No of members	No of days
1										
2										
3										
4										
5										
6										
7										
8										

Hired Labor

Plot ID	Plot Name	Did you hire any labor to work on this plot?(1=Yes, 2=No) G28	Why did you hire in labor on this plot? G29	For how ma days did you labor G.	ny man 1 hire the 30	How much did y	ou pay for the labor? G31
1				No of workers	No of days	Cash	In kind
2							

3				
4				
5				
6				
7				
8				

H. Harvest How much did you harvest last season (2009/10)

	Crop code						Harv	Indicate the major reasons for the change				
					1 st		2 nd		3 rd		4 th Others	
					Quantity	Unit Code	Qunatity	Unit code	Quantity	Unit code	Estimat ed value	
	H1				H2	H3	H4	H5	H6	<i>H7</i>	H8	H11
1												
2												
3												
4												
5												
6												
7												
8												
9												
10												
11												
12												
Use Crop codes												
----------------	--											

Code H3, H5, H7: 1= basket 2=oxcart 3=pail 4=wheelbarrow 5=bags (50kg) 6=bags (90kg) 7= bales 8=Nkhokwe 9= lichelo (basin) 10=others

Ys:Forest Products, Sales and markets

				If answer to Ys2 is YES					If Ys2 i	If Ys2 is NO Transport costs			State of yield in 5 years	Reasons for the change
Product code	Estimate the amount of forest products	Did you sell your forest products last year (2008/2009)?	Harves	est was ld?	Average price of the product	Dist anc e to mar	Type of market	Total value sales	Why wasn't some products sold?	Estimate the value not sold	Means of transport to the market	Average total cost of transpor		
		1-yes 0-no	Qty	Units		ket						t		
Ys1	Ys1.1	Ys2	Ys3	Ys4	¥5	Ys6	Ys7	Ys8	Ys9	Ys11	Ys12	Ys13	Ys 14	Ys 15
-														
				1								1		

Ys1 1= firewood, 2= charcoal, 3. honey, 4=food from the forest, 5= medicine, 6= forage 7. Cactus 8. Gums and resins 7= other specify

Ys4 1= basket, 2= oxcart 3=pail 4=wheeelbarrow 5=bags (50kg) 6=bags (90kg) 7=headload 8=othersYs6 Walking 5 Km/hrsOxcart 4km/hrBicycle 15km/hrYs71= Forest-gate, 2= Local (primary) markets, 3= District Assembly markets, 4= Urban markets, 5 = Other (specify)Ys91=home consumption 2=lack of market3=other (specify)Ys12 1=Head load, 2 Ox cart, 3 Bicycle, 4 Vehicle, 5 Wheel barrow, 6 othersYs14 1=increasing 2=decreasing 3=constant

H. Marketing. (For each crop that was sold, please ask)

	What was the means of transport to the market? <i>Hs12</i>	What was the cost of transport to the market? <i>Hs13</i>	When did the household sell the crop? <i>Hs14</i>	Why did the household opt to sell/store at that period? <i>Hs15</i>		If they stored, what kind of storage mechanisms did the household use? <i>Hs16</i>		How long was the produce stored in months? <i>Hs17</i>	Did yo any p when y your c sa H	ou incur roblems ou stored crops for ale? Is18	Did you grade your produce before selling? <i>Hs19</i>
				Sell Hs15a	Store Hs15b						
Use crop code	1 Head load 2 Ox cart 3 Bicycle 4 Vehicle 5 Wheel barrow 6 others		1 – Immediately after harvest 2 – They stored and sold at later date 3 – Sold some after harvest but stocked some for sale at later period 4 – Other	 arrousenoid needed an immediate source of income 2=To take advantage of prevailing high prices at the time 3=Lacked storage place/ mechanism 4= Wanted to wait for better prices after harvest season 5=Others (specify) 		1=Granary (Nkhokwe) 2=In the home kept in Bags, sacks, baskets 3=Late harvest 4=pit storage 5=Others (specify)			0=No 1=Theft 2= Loss of q 3=Destructio 4= Prices ne 5= Other (sp	uality on by pests ver went up becify)	0= No 1= Yes

FO: Farmer Organizations

Fo1: Do you belong to a farmer farming organisation? 1=Yes 0=No

If yes to Fo1

What kind of	How long ha	ve you been a	Why did you join the organisation?	What 3 im	portant functions o	loes the FO carry out?	
organisation is it?	member?		Fo4		Fo5		
Fo2	F	'o3			100		
	Years	Months					
1= Farmer cooperative 2=Farmer club 3=Association 4= Others (specify)				0- Nothing 1-Helps farmers access inputs on loan 2=Markets produce for farmers 3=Provides extension advice 4=Others (sneecify)			

Fo4: 1 = reqired to join 2 = Everybody belongs 3 = invited to join 4 = volunteered 5 = family/friends helped me to join.

I. Animal Husbandry

Livestock code	How many do	What is the	Did you bu four Month	y any during 1s?	the last	Did you sell a	ny during the Months?	e last four	How many	How many	How many
п	you have now? I2	estimated price if you were to sell today? I2.1	Number bought (if none, 0) I3.1	Total purchase value I3.2	Source of finance I3.3	Number sold (if none, 0) I3.4	Total sales value I3.5	Reason for Sale I3.6	slaught ered in the last four months I3 7	have died I3.8	were stolen I3.9
bulls/oxen									1017		
Cows											
Sheep											
Goats											
Camels											
Donkey/mul											
es/horse											
Chicken											
Bees											
Others											

Livestock Expenditure and Income

During the last four months , have you had expenditure related to livestock during the last four	Cash value (if in kind estimate	Gross income from the sale of household's animal product during the last	Did you sale any Yes=1	Amount sold	Unit L2c	Total revenue obtained L2d
		four months:	1.2a	L20		
111	110	L2a	Liza			
Type of Expenditure		Туре				
Labor for herding		Meat				
Feed, including salt		Hides and skins				
Veterinary services/Medicine		Butter /cheese				
Transport of animal feed		Milk /Cream				
Commission on the sale of animals		Dung cakes				

Other expenses(specify)	Chicken		
	eggs		

E2. Proportion of Labor allocated to the different activities in a Year (Out of 20 matchsticks, how many would you give in terms of labor allocated to the following Activities)

Member	Household	Collecting	Other	Main	PSNP	Other	Non Agricultural
ID	Activities	Firewood	Forest	Agricultural		Agricultural	activities
			Activities	fields	E25	Activities	E27
	E21	E22				E26	
			E23	E24			

E.28. Indicate distance (in km) from home to forest thinning or forest clearing area:_____

E.29. Indicate distance (in km) from home to firewood collection area:

E.30. Indicate distance (in km) from home to forest based area of wage.....

E.31. Wage per hour of forest based wage work.....

M. Forest Resources and Tenure

M.1. Do you have private woodland or grass land ? 1. Yes 2. No

If yes, Please state your total collection of non – timber forest products from natural forest (Private forest)

Forest product	Qt collected per trip	Number of trips a month	Most frequent used quarter	Total time spent per trip			Average time to go to Natural Forest		
				AM	AF	YM	YF	Adults	Young
Resins and gums									
Honey									
Cactus									
Fuel wood									
leaves									
Crop residues									
Medicinal herbs									
Thatching grass									
Fruit /nuts									
Bee wax									

Other, specify					
A A A A A A A A A A A A A A A A A A A	1 1 1 1 1 1 1	F 1			

Am= Adult male, AF = Adult Female, YM = Young male, YF = Young Female

M.1. How much income did you generate from sale of the non- timber forest products obtained from your private woodlot ? -----

M.2. Assuming that all the benefits you obtain and use it from your private woodland/grass were obtained through purchase, how much would you incur /pay month ? ------

Forest product	Qt collected per trip	Number of trips a month	Most frequent used quarter	Total time spent per trip			Average time to go to Natural Forest		
				AM	AF	YM	YF	Adults	Young
Resins and gums									
Honey									
Cactus									
Fuel wood									
leaves									
Crop residues									
Medicinal herbs									
Thatching grass									
Fruit /nuts									
Timber									
Other, specify									

Please state your total collection of non - timber forest products from natural forest other than private forest

Am= Adult male, AF = Adult Female, YM = Young male, YF = Young Female

M.3. How much income did you generate from sale of the non- timber forest products obtained from other forests? ------

M.4. Assuming that all the benefits you obtain and use it from other forests obtained through purchase, how much would you incur /pay month? ------

N. Climate change Related Issues

N.1. Have you noticed any long -term changes in climate over the last 20 years? 1. Yes 2. No

N.2. If yes what are the manifestations of change in climate change (multiple answers possible) 1. Increase in temperature 2. Decrease in Temperature 3. Increase in Rainfall 4. Decrease in Rainfall 5= other, specify

N.3. Do you have access to information related to climate change 1. Yes 2. No

N.4. How many times did you encounter complete crop failure due to climate change? ------

N.5. Have you made any adjustments in your farming practice due to climate change? 1. Yes 2. No

N.6. If yes , what adjustments related to farming, livestock and forests have you made to the change in climate

No	Farming Practice	Livestock	Forests

RS.Recent shocks to household welfare

Has this household experienced ANY major shock since 2006?

GO THROUGH THE ENTIRE LIST	Did you experience a shock this year? 1-yes 0-no	The year shock occurred	Note down the three most significant shocks you experienced for each year	Degree of coverage	Duration of shocks in weeks	Effect of the shock	Estimated total value of loss (not for 11-14)	Wha respons try t forme	t did you se to this s to regain y er welfare	do in hock to your level?
	K1.1	K1.2	K2	КЭ	K4	К5	КО		K /	1
1- Lower yields due to drought or flood			1							
2-Crop disease or crop pests		2006/7	2							
3-Livestock dies or were stolen			2							
4-Large fall in sale prices for crops 5 Household buisness failure			3							
6-Loss of salaried employment		2007/8	1							
7-Non-payment of salary			2							
8-End of regular assistance, aid, or			3							
remittances from outside HH			1							
9-Large rise in price of food		2008/9	1							
12-Death of working members of the			2							
HH			3							
13-Illness or accident of household member		2009/10	1							
14-Death of other family member			2							
16-Dwelling damaged, destroyed 17-Theft			3							
18-Other (specify)			1							

R3: 1=Own HH only 2=Some other HHs too 3=All HHs in community

R5: 1=*Reduction in income* 2=*Reduction in assets* 3=*Both* 4=*Nothing*

<i>R7:</i> 0=Nothing	8=Removed children from school to work
1=Spent cash savings	9=Sent children to live with relatives
2=Sold assests (tools etc)	<i>10=Went elsewhere to find work for more than one month</i>
3=Sold forest products	11=Borrowed money (relatives, bank, local money lender)
4=Sold animals	12=Received help (governent, NGO, etc)
5=Sold more crops	13=Reduced food consumption (smaller proportions, fewer meals per day)
6=Worked more (incl. other HH members,)	14=Diversify food consumption (Wild foods, meal sharing, no meat or fish)
7=Started a new buisness	

	Questions	Answers	Codes
S1	All things considered, how satisfied are you with your life over the past 12 months?		<i>1=very unsatisfied; 2=unsatisfied; 3=neither unsatisfied or satisfied; 4=satisfied; 5=very satisfied</i>
S2	Has the household's food production and income over the past 12 months been sufficient to cover what you consider to be the needs of the household?		0=no 1=yes 2=reasonable (just about sufficient)
S 3	Compared with other households in the village (or community), how well-off is your household?		1=worse-off 2=about average 3=better-off
S4	How well-off is your household today compared with the situation 5 years ago? If 1 or 3, go to S5. If 2, go to S6.		<i>1=less well-off now 2=about the same 3=better off now</i>
85	If worse- or better-off: what is the main reason for the change? <i>Please rank the most important responses, max 3.</i>		<pre>1=off farm employment 2=land holding (e.g., bought/sold land) 3=forest resources 4=output prices (forest, agric,) 5=outside support (govt., NGO,) 6=remittances 7=cost of living (e.g., high inflation) 8= civil strife, unrest 9=conflicts in village (non-violent) 10=change in family situation (e.g. loss of family member/a major bread-winner) 11= illness 12=good infrustracture (access, e.g. new road) 13=other (specify):</pre>
S6	Do you consider your village (community) to be a good place to live?		0=no 1=yes 2=partly
S7	Do you in general trust people in the village (community) whencooperating on ?		0=no 1=yes 2=partly, trust some and not others

S. Social capital and welfare perceptions

S8	Can you get help from other people in the village (community) if you are in need, for example, if you need extra money because someone in your family is sick?	<i>1=Definitely 2=Probably 3=Probably not</i> <i>4=Definitely not</i>
S9	About how many friends do you (HOUSEHOLD HEAD) have in your community these days? These are people you feel at ease with, can talk to about private matters or call on for help	
S10	About how many friends do you/does your SPOUSE have in your community	
	matters or call on for help.	
S11	In the past 12 months, how many people with personal problems have turned to you for any form of assistance?	

W. Credit, Saving and Extension Services

W.1. Have you taken any financial Credit in 2009/10? 1. Ye s 2. No If yes, give the following details

Outstanding Loans

	Loan so W2	ource	Original reason for taking out loan W3 (Multiple answers possible)		When loan or taken? W4	was the iginally	Amount borrowed W5		Amount need to be repaid W6		What w actuall W7 (Multip	vas the loan money y used for? le answers possible)	What is the amount of interest W8	
	1. Bureau of Agriculture (HH extension package)		1. To buy livestock(2. To buy food((Month year in	n and	(Indicate birr or (I		1. To buy livestock(Indicate birr or2. To buy food		uy livestock uy food			
	2. Moneylender 3. To buy/have		uy/have agricultural inputs	Ethiopi	an	value in birr)		value in birr)		3. To buy agricultural inputs				
	3. Friend/neighbor/ (fertilize		lizer, seed, irrigation)	calenda	ar)					4. Heal	th expenses			
	relati	4. Health expenses		th expenses							5. Educ	ational expenses		
	5 Bank		6 Housing materials expenses								7 To nav for hired labor			
	6 DECSI (DEDEBIT) 7 To nav for hired labor		av for hired labor							7. TO p	art new husiness			
	7 Cooperatives 8 To start new busines		art new business							9. Othe	r (specify)			
	8. Other (specify)		9. Other (specify)								10. Has	not used loan yet		
LOAN 1	30.1		31.1		32.1		33.1		34.1		35.1		36.1	
LOAN 2	30.2		31.2		32.2		33.2		34.2		35.2		36.2	
LOAN 3	30.3		31.3		32.3		33.3		34.3		35.3		36.3	

W9. Did you have any financial savings in the Bank? 1. Yes 2. No

W10. If yes how much did you save in Birr ? -----

W11. Did you have visits from extension staff last year? 1. Yes 2. No

W12. If yes how many times? -----

W13. Is the extension advice you get in the last 5 years teach about climate change ? 1. Yes 2. No